

## SSE Riga Student Research Papers 2023 : 10 (262)

# JUSTIFYING LIMITATIONS OF LIBERTY: EVALUATION OF ARGUMENTS USED IN SAEIMA'S AND THE CABINET OF MINISTERS' COVID-19 VACCINE MANDATE DISCUSSIONS

Authors: Evelīna Daniela Baltiņa Toms Truksnis

ISSN 1691-4643 ISBN 978-9934-623-12-7

> May 2023 Riga

# Justifying Limitations of Liberty: Evaluation of Arguments Used in Saeima's and the Cabinet of Ministers' COVID-19 Vaccine Mandate Discussions

Evelīna Daniela Baltiņa and Toms Truksnis

Supervisor: Xavier Landes

JEL codes: I18

May 2023 Riga

#### **COPYRIGHT DECLARATION AND LICENCE**

Names of the authors in full: Evelīna Daniela Baltiņa, Toms Truksnis

Title of the Thesis: Justifying Liberty: Evaluation of Arguments Used in Saeima's and the Cabinet of Ministers' COVID-19 Vaccine Mandate Discussions

We hereby certify that the above-named thesis is entirely the work of the persons named below, and that all materials, sources and data used in the thesis have been duly referenced. This thesis – in its entirety or in any part thereof – has never been submitted to any other degree commission or published.

In accordance with Section 1 of the Copyright Law of Latvia, the persons named below are the authors of this thesis.

Pursuant to Article 40 of the Copyright Law the authors hereby agree and give an explicit licence to SSE Riga to deposit one digital copy of this thesis in the digital catalogue and data base at SSE Riga Library for an unlimited time and without royalty. The licence permits SSE Riga to grant access to the duly deposited thesis to all users of the catalogue and data base without royalty and limitations to downloading, copying and printing of the digital thesis in whole or in part provided we are indicated as the authors of the thesis according to Clause 4 Section 1 Article 14 of Copyright Law. We assert our right to be identified as the authors of this thesis whenever it is reproduced in full or in part.

Signed

F RI

/signed/\_\_\_\_\_

/signed/\_\_\_\_\_

/Evelīna Daniela Baltiņa/

/Toms Truksnis/

Date

03.04.2023.

## **Table of Contents**

Abstract	6
1. Introduction	6
2. Literature review	7
2.1. Vaccines: controversies and hesitancy	8
2.1.1. Global vaccine controversies	
2.1.2. Vaccination opinions in Latvia	10
2.1.2. Vaccination opinions in Earvia     2.1.3. COVID-19 vaccine development	11
2.2. Defining vaccination mandates and other policies	12
2.3. Arguments on vaccination mandates	14
2.3.1. Arguments for vaccination mandates	15
2.3.2. Arguments against vaccination mandates	
3. Methodology	16
3.1. Data collection	
3.1.1. Data sources	17
3.1.2. Analysing transcripts	18
3.1.3. Grouping arguments	19
3.2. Reformulating arguments	20
3.3. Evaluation of arguments	21
3.3.1. Evaluation of soundness	22
<ul><li>3.3.1. Evaluation of soundness</li><li>3.3.2. Evaluation of moral strength</li></ul>	22
3.3.3. Ethical theories	23
3.4. Limitations	23
4. Analysis of Results	24
5. Discussion	29
5.1 Main Dromises Used in Arguments For Vessingtion Mondetes	20
5.1 Main Premises Used in Arguments For Vaccination Mandates	

5.1.2 Motivation	31
5.1.3 Non-Discriminatory/Fair	33
5.1.4 Testing Ineffectiveness	35
5.1.5 Legitimacy of Freedom-Reducing Policies	37
5.2 Arguments for vaccination mandates	
5.2.1 The Right Not to Be Harmed	
5.2.2 Economic Recovery	41
5.2.3 Easing, not Restricting	42
5.2.4 Achieving Herd Immunity Safely	44
5.2.5 High-Risk Professions	46
5.2.6 Critical Professions	
5.2.7 Vaccine Safety	49
5.2.8 Necessity for Reciprocal Policies	51
5.3 Arguments against vaccination mandates	52
5.3.1 Wrongful Discrimination	52
5.3.2 Social Capital	54
5.3.3 Free Choice/Personal Autonomy	55
5.3.4 Vaccine Ineffectiveness	57
5.3.5 Anti-Paternalistic View	
5.3.6 Less Restrictive Options	
5.3.7 Vaccine Safety	62
5.3.8 Government Failure	
5.3.9 Slippery Slope	66
5.3.10 Unfair Targeting of Protected Population	68
5.3.11 Unintended Consequences	70
5.3.12 Unemployment Effects	72
5.3.13 Employer Ethical Dilemma	74

5.4 Results	76
6. Conclusion	77
Glossary	81
7. References	83



# **SSE RIGA**

#### Abstract

The COVID-19 pandemic took the world by storm, and the common belief was that policymakers were not sufficiently prepared to deal with this global crisis. We attempt to explore this in our work – did Latvian politicians do a sufficient job of arguing for or against mandatory vaccination within the state? Such a policy is extremely intrusive onto personal rights; thus, politicians should express just reasoning for doing so. We approach the problem through document analysis of the Latvian Parliament's verbatim meetings and the Cabinet of Ministers' video meetings to collect arguments, following that with moral analysis to evaluate the arguments. We find that both sides of the debate used both strong and weak arguments to make their point. We outline the main arguments that were strong and explain how they could be used in other types of policy debates to ensure that the citizens' interests are well-represented.

#### **1. Introduction**

Often political decisions can impact us to a great extent. We would like to have a say in how our lives are dictated, but rarely can we – at least not directly. In a democratic state, such as Latvia, people get to have a say in the country's political decisions by casting a vote during the elections. It's not a perfect solution as there is no guarantee that the politicians that we elect will stay true to their word or be as competent as they claim to be. However, it is the situation we find ourselves in – given the incapacity of the public to vote on issues directly, politicians are expected to (and they promise to) act in the best interests of their voters, for example, by representing them with their voice and vote in the Latvian Parliament, Saeima.

Some decisions may dramatically impact the population. For example, a vaccination mandate, which requires a person to choose between giving up something meaningful for their lives and well-being like studies or work or getting vaccinated. Such freedom-altering policies should be passed only when the benefits of such a policy outweigh the disadvantages. Therefore, we believe that when debating on decisions to be made in Saeima, politicians ought to use sound and morally convincing arguments that correctly highlight the perceived benefits or costs. Sadly, we expect that many arguments made on important decisions will not be formulated properly, will be lacking soundness and moral strength, and may be emotional and fallacious.

This paper is interested in COVID-19 vaccine mandates. It is a recent phenomenon that has impacted many Latvian residents in one way or another, through restrictions on work, studies, leisure activities, and so on. These policies were hardly uncontroversial -

according to Pētījumu Centrs SKDS (2021a), at the end of 2021, only 55% of respondents stated that they support vaccination against COVID-19, meaning that nearly half of the population either do not support vaccination outright or are not completely sure. Given that these numbers represent support only for vaccination, one could expect the support for mandates to be even lower. Additionally, according to the same source, 35% of fully vaccinated respondents got vaccinated due to the vaccine mandates, rather than due to genuine support for this decision, and more than 80% of the people who are only planning to vaccinate will do so as a result of mandatory vaccination policies. It is visible that the vaccine mandates have had a large impact. What remains unclear is how exactly this decision was made. We want to find out whether the vaccine mandates that imposed restrictions on the freedom to work were based on strong arguments (as well as the arguments against them). In this study, we have defined two main research questions:

- 1. What were the arguments used by Saeima and Cabinet politicians in debates about COVID-19 vaccination mandates?
- 2. How strong were the arguments used (based on the premise, deductive/non-deductive reasoning, logic, validity, and persuasiveness)?

This type of coordinated global response has not been seen in previous pandemics and may not appear for the last time, which is what makes this topic even more necessary to study. COVID-19 is likely not the last pandemic we will face, in which case this debate will restart, and by that time we need to be ready to answer the most difficult questions about how ethical it is to mandate vaccination. Or, at the very least, we should know which arguments to avoid. We might not even need to wait for a new disease, as newly updated vaccines are in production now (Balch, 2022). The majority of the Latvian population is not planning to take these vaccines (TV3, 2022), which inescapably points to new debates, and possibly – vaccine mandates.

Although there is vast academic literature on vaccination and infectious disease ethics, we have not been able to find studies that evaluate arguments used by politicians on this topic, especially not in Latvia. However, we have drawn inspiration from similar studies that researched and assessed politicians' arguments used in Saeima and elsewhere about unrelated topics like same-sex marriage and civil unions (Baranova & Mežmale, 2021).

#### 2. Literature review

To better frame our proposed research, we begin the literature review with a look at controversies surrounding vaccination, both historically and in the context of the COVID-19

pandemic. Next, we outline a framework for defining what counts as a vaccine mandate. We finish by providing academics' arguments on vaccine mandates, to get an understanding of what we might find when doing our analysis, as well as to be able to better perform it.

#### 2.1. Vaccines: controversies and hesitancy

Vaccines are a means of protection against diseases. They create immunity against a certain disease before the person even comes in contact with it. A vaccine trains a person's immune system to create antibodies against the given disease much like in the way contracting the disease would, but this method is safer and causes little to no complications, with exceptions in the case of allergies or being immunocompromised (World Health Organization [WHO], 2021b). The contribution that vaccination has made to society as a whole is colossal – smallpox has been completely eradicated and many other diseases (e.g. polio, measles) are now extremely rare thanks to vaccines against them (Greenwood, 2014).

Despite the success that vaccines have shown so far, they remain controversial among the general public as they have been since their first appearance. It makes sense that in highincome countries, where vaccine-preventable diseases are no longer prominent, the concerns about vaccine risks will tend to be stronger than concerns about risks associated with the specific disease, simply because people have had no experience with those types of illnesses and cannot evaluate the possible severity of the outcome of infection (Dubé et al., 2015).

Before looking at specific vaccination policies and determining what the arguments are in the debate of those, we will first look at the reasons behind vaccine hesitancy and past controversies surrounding vaccination, as these may have set the tone for how people view them. Although the studies we look at research the general public's reasoning for vaccine hesitancy, we believe this translates into politicians' statements as well, since they are representatives of the public and may use the same or similar arguments. We draw information on vaccination controversies from global studies, as we were unable to find local ones, but we trust that these controversies may have appeared in Latvia as well. Additionally, we give a brief overview of the general attitude of Latvians on vaccination, which can give some insights into the issues that may arise when trying to convince people to vaccinate. It is important to note here that some of the controversies and/or reasons for vaccination refusal or hesitancy will be discussed in terms of parental decisions for child vaccination since that is the most prevalent debate as of late, and we believe that similar arguments against COVID-

19 vaccination may be used. Lastly, we give an overview of the development of the COVID-19 vaccines and some public opinions on these particular vaccines.

#### 2.1.1. Global vaccine controversies

Despite smallpox being a potentially lethal disease that had haunted humanity for multiple centuries, the first ever vaccine created against this disease nonetheless faced resistance, even from the co-discoverer of natural selection, Alfred Russel Wallace. Promoting vaccination in the US during smallpox outbreaks in the late 19<sup>th</sup> century and making smallpox vaccination compulsory in the UK during approximately the same time both were faced with resistance. The public responded by creating anti-vaccination leagues in both countries, as well as releasing books and journals promoting anti-vaccination ideology (Dubé et al., 2015). We are mentioning this just to show that anti-vaccine opinions have existed for as long as two centuries, if not even longer, thus these types of movements are not something unique to our time.

It is alleged that controversies on the pertussis vaccine that began in the 1970s were the start of the vaccination controversies of our times. This specific controversy began in the UK when the Great Ormond Street Hospital for Sick Children in London published a report that supposedly found that thirty-six children had developed serious neurological disorders after having received the combined diphtheria, pertussis, and tetanus vaccine (DPT). In response, The Association of Parents of Vaccine Damaged Children was founded in 1974. The report and attention to the association caused concerns among many parents, which lead to actual decreases in child immunization rates. This in turn caused new major pertussis epidemics, and despite the efforts of the UK Joint Commission on Vaccination and Immunization to convince the public that, indeed, the vaccine had not caused any neurological issues by giving the results to a study that had looked at all of the hospitalised children with neurological diseases, the resistance against the vaccine persisted. These concerns and controversies traveled across borders to Europe, the Soviet Union, the USA, Japan, Australia, and perhaps others. In the 2000s came another controversy in the UK, where it was claimed that the MMR vaccine has a connection with autism in children, more specifically, that the vaccine causes it. Unsurprisingly, such public unease yet again led to decreased immunization rates, this time for measles, which caused new outbreaks of the disease and multiple deaths (Dubé et al., 2015).

The tipping point for the USA from a period of general vaccination acceptance to anti-vaccination movements was with the release of the documentary 'DTP: Vaccination

Roulette' in 1982. The film stated that the pertussis portion of the combined DTP vaccine was causing brain damage, seizures, and mental deficiency. This began a wave of lawsuits against vaccine manufacturers, which resulted in fewer companies entering the market and more exiting it, as well as increased vaccine prices (Dubé et al., 2015; Insel, 2012).

The age of the internet has made it much simpler to share your ideas with an extremely large amount of people, and most of the time, there is little to no policing on what you say, especially on user-generated sites. What this means is your statements will not be fact-checked, and the reader cannot know whether it is true and must use their judgment to decide whether to trust the statement or not. Some studies have shown that parents will typically look for information about vaccination on the internet when trying to decide on vaccinating their children (Ekos Research Associates Inc., 2011; Stefanoff et al., 2010). One study on Danish parents' decisions to not vaccinate (partially or fully) has shown that refusal or hesitancy reasons are: the belief that providing the child with a healthy lifestyle will reduce the risk of getting ill; the immune system of a child is not sufficiently developed to handle vaccination; the risk of actually contracting or coming in contact with the disease is extremely low; vaccines are unsafe; it is healthier to gain immunity from disease by contracting it instead of vaccinating (Harmsen et al., 2013). Another study on parents in the US showed similar reasons for hesitancy, in addition to those mentioned, other arguments were about vaccines being ineffective in protecting against disease, vaccines being unnecessary for the health of a child, children receiving too many vaccines, and that getting multiple vaccines within short time intervals could overwhelm the child's immune system (Smith et al., 2011).

#### 2.1.2. Vaccination opinions in Latvia

In a study conducted before the pandemic, Larson et al. (2018) found that Latvia was the second to last country from all 28 EU member states in terms of belief in vaccine safety with 68.2% of respondents agreeing to the statement that vaccines are safe, while the EU average stood at 82.1%, as well as 55.2% Latvians responding that they believe the seasonal influenza vaccine is safe and again ranking as the second to last under this question with an EU average of 67.8%, and Latvians had the lowest percentage of respondents indicating that vaccines are effective, 70.9%, while the EU average was 86.5%. This points to potential issues with vaccine hesitancy or refusal, which is what was observed during the COVID-19 pandemic.

Additionally, some research has shown that a person's level of trust in the government influences vaccine acceptance, namely, people who trust their government tend to be more likely to accept vaccines than their not so trusting counterparts (Lazarus et al., 2020; Lee et al., 2016; Trent et al., 2022). Data provided by OECD (n.d.) shows that in 2019, 23.9% of Latvians said they trust the government, in 2020 it was 30.7%, and in 2021 - 29.5%. Though the level of trust did rise during the pandemic (and then dropped off a bit), the level of trust nonetheless is low. A survey done by SKDS showed that the trust in official government information was 63% among 1,000 respondents in the spring of 2020 but had dropped down to 53% by the fall of the same year, with the proportion of respondents saying they don't trust government information rising from 28% to 37% from spring to fall (Zvirbulis, 2020). These statistics are another indicator pointing to Latvia having possible difficulties with vaccination uptake once the COVID-19 vaccines become available.

The COVID-19 vaccination uptake (in terms of the percent of the population having finished the primary course, according to each manufacturer) in Latvia lagged behind the EU/EEA country average by about 17 percentage points starting from the beginning of August 2021 up until about the beginning of October 2021, when the rates began to converge, and as of January 1, 2023, the portion of Latvian citizens who have finished the primary vaccination course against COVID-19 is 69% compared to a 73% average between EU/EEA countries (European Centre for Disease Prevention and Control [ECDC], n.d.).

A study on the Latvian anti-vaccination movement on Facebook during the COVID-19 pandemic by Eiduks & Ozola (2022) found that the main reasoning behind vaccine hesitancy or refusal was most often due to a lack of trust in government officials and healthcare practitioners, as well as perceived risks of vaccination outweighing the benefits due to the abundance of misinformation available. In our opinion, the reasoning for not trusting the government may not show up in politicians' arguments, but arguments based on misinformation likely will.

#### 2.1.3. COVID-19 vaccine development

The COVID-19 vaccines were developed at an incredible speed – the first vaccines began rolling out about a year after the first case of the virus had been recorded. Given the sudden onset of the pandemic, the need for a solution was dire, which lead to increased cooperation efforts between vaccine developers to create an effective immunization tool against COVID-19 much quicker. Moreover, while typically vaccine testing goes through three phases, most

of the time one phase at once which can take 10 to 15 years on average, the need to stop the spread of this new disease meant that some phases were merged (typically phases 2 and 3, which are the phases for testing vaccine safety on people with health issues and for testing the efficacy of the vaccine in protecting against the disease, respectively) and some vaccines were even tested together. Additionally, for the development of COVID-19 vaccines, scientists had to collaborate and share research data among themselves because time was of the essence in this situation (Solis-Moreira, 2021; WHO, 2021a). Another determinant for the quick development tempo was that scientists had been studying other coronaviruses many years before the COVID-19 pandemic hit, meaning that there was already a considerable amount of data gathered on the specificities of these types of viruses. Of the first vaccines to become available, two were from the companies Moderna and Pfizer, both of which used a new technology called mRNA, which had also been researched in depth many years beforehand ("COVID-19 Vaccine Development: Behind the Scenes," n.d.; Solis-Moreira, 2021).

A survey by Byron (2020) across 27 countries with nearly 20,000 respondents found that the main reason for vaccine hesitancy towards COVID-19 vaccines was due to fear of side effects (56% of respondents), followed by disbelief in the vaccines' effectiveness (29%) and not being at risk of serious complications in case of infection with COVID-19 (19%).

### 2.2. Defining vaccination mandates and other policies

To better understand the policies imposed during the COVID-19 pandemic in Latvia and to define what we will understand as a vaccination mandate in this paper, we need to have some sort of framework to go by.

Giubilini (2019) provides an intervention ladder of vaccination policies with the assumption that the goal of such policies is achieving herd immunity. The ladder is built on the principle of least restrictive alternative (PLRA), developed by Childress et al. (2002), and defined by Saghai (2014) as follows: "if two interventions can both efficaciously and effectively address a public health or health policy issue and are equal in all other morally relevant respects, the intervention least restrictive of personal liberties ought to be preferred" (p. 350). We will be using this framework to determine what level of restriction was used in each of the policies introduced in Latvia during the COVID-19 pandemic. Giubilini (2019) argues that a combination of two criteria should be adopted for ethical guidance for public policies: "the maximin criterion for the distribution of the burdens of a certain policy,

constrained by a utilitarian calculus based on the consideration of the number of people who are burdened by a certain policy." (p. 77) Additionally, he mentions that according to these two criteria, the ranking of policies may differ across different contexts (Giubilini, 2019). The ladder refers to child vaccination policies mostly since their effectiveness has been researched more, but since it is the parents (adults) making the decision, we decide to infer that research on these policies can be applied to policies implemented on adults during the COVID-19 pandemic also. Additionally, since we are not examining whether the Latvian government took the right steps in the right order, we will present the ladder as it is presented by Giubilini (2019) since our objective is to simply identify what is to be considered a mandate and what is not.

The ladder goes as follows:

- 1. **Persuasion**: lowest level of restrictiveness/coercion. It is non-manipulative and preserves individual autonomy by using factual information to educate the public and give them reasons for participating in vaccination. An example would be health education campaigns.
- 2. **Nudging**: non-coercive and minimally restrictive. It takes advantage of natural decision biases and automatic cognitive processes humans have to encourage vaccination, i.e. it is manipulative. An example would be opt-out policies.
- 3. **Incentives**: more coercive and restrictive than the former. This type of policy comes in monetary form (known as conditional cash transfers). For the more financially vulnerable population, such incentives can appear irresistible, which is why this type of policy can be considered more coercive.
- 4. **Disincentives**: coercion through (usually) exploiting loss aversion and the endowment effect. The three types of disincentives mentioned (in growing order of restrictiveness) are:
  - a. Withholding of financial benefits: examples examined here are withholding childcare benefits that would otherwise be paid but are not if the child is not vaccinated against certain diseases.
  - **Tax**: a financial penalty imposed on people refusing vaccination that is proportionate to the infection risk of infection a person, who is not vaccinated, poses to others.
  - c. **Mandatory vaccination**: an example here is denying enrolment in school and daycare if the child has not received certain vaccines (such as in the US and Italy). Here, parents continue being legally free in the

choice of whether to vaccinate their children, but it comes with the cost of having to home-school the children or send them to a private school, which may not be accessible to all parents. In some cases, non-medical exemptions are allowed on the basis of religious or philosophical beliefs against vaccines.

5. **Compulsion**: a policy that makes it illegal to refuse vaccination. This would mean legal penalties (anything from a fine to imprisonment) in the case of non-vaccination (Giubilini, 2019).

The mandatory vaccination that Giubilini (2019) discusses fits the criteria we logically and intuitively want to look at for the policies imposed in Latvia over the COVID-19 pandemic. It was also the case that some of the policies implemented by the Latvian government restricted certain activities for the unvaccinated population (work, visiting shops, meeting with friends and family, etc.). This is the definition of mandatory vaccination that we will stick to in this paper.

Importantly, although many policies that infringe on individual freedoms can be classified as vaccine mandates, for this paper we reduce our scope to only include vaccine mandates regarding work-related restrictions. This is done for two reasons. Firstly, a reduction in scope makes also reduces the vast amount of data to be covered and makes it more plausible for this type of study to be performed. Secondly, if we were to reduce scope, we find it more impactful to focus on the most severe policies above all. To our minds, restrictions to work have been the most impactful COVID-19 vaccine mandates as those are the social activities necessary to make a living now and/or build a better future for oneself.

#### 2.3. Arguments on vaccination mandates

A vaccination policy needs to strike a balance between caring for public health and maintaining individual liberties. Vaccination mandates have possibly the most ethically controversial nature, and there is no lack of arguments on either side. We will look at the arguments provided by academics so that we can later compare them with the arguments posed by Latvian politicians in the discussions of vaccination mandates meant for stopping the spread of the COVID-19 disease. We would additionally like to mention that we believe arguments for or against vaccination may have been used in the discussion of vaccination mandates as well, thus a few of such arguments may be mentioned here as well.

#### 2.3.1. Arguments for vaccination mandates

Vaccination benefits both the individual who is vaccinated and his community – one through direct immunity, and one through herd immunity. It is often argued that an unvaccinated person imposes risk or harm on others, as well as that each person has the right not to be harmed by others (Flanigan, 2014). This touches on John Stuart Mill's harm principle, that one's liberties are not unlimited if they cause harm to another, and restriction of liberty is allowed in such a case (Brink, 2022). The (no) harm principle is not sufficient as a standalone argument for restricting a person's liberties and it should be supplemented by other arguments.

When looking at mandates from an ethical perspective, one can argue that they are supported by utilitarianism. The expected outcome of mandatory vaccination is reaching herd immunity, which results in a positive net benefit after reducing the cost of getting the vaccine, which is small since the vaccine is free of charge and the chance of serious side effects is minimal. Herd immunity, however, results in more lives saved, less stress on healthcare systems (which, in turn, leads to fewer elective procedures being canceled), better well-being, and higher economic growth because of fewer restrictions (Gandjour, 2022). Additionally, an argument can be made that not imposing vaccine mandates can lead to deaths and uncertainty, which in turn – to radical ideas and conspiracies. In other words, not playing into conspiracy theorists by seemingly appearing less tyrannical (not mandating vaccination) can in turn result in more conspiracy theories (Lewandowsky et al., 2022).

It can also be said that vaccination mandates are justified by deontology. If a person does not vaccinate, they are imposing harm on another person, and in deontology, this means that they are restricting the autonomy of the other person, which goes against the ethical theory (Gandjour, 2022). There are separate arguments to be made for the vaccination of healthcare professionals, as they have the duty to act in ways benefitting public health, i.e. leading by example. Non-vaccination of healthcare professionals could undermine the public belief in vaccines, which in turn could cause harm to individuals that decide not to vaccinate, ultimately going against the aforementioned duty of these professionals (Fumadó et al., 2021).

#### 2.3.2. Arguments against vaccination mandates

When talking in particular about the mandates introduced during the COVID-19 pandemic, some argue that they also cause social polarization, and perhaps the implementation of such a

policy is even driven by the desire to reinforce discrimination and stigmatization. Additionally, it can be argued from a behavioral psychology perspective that such a policy can create resistance instead of submission if distrust in authorities is already present. Moreover, it can create not only resistance to COVID-19 vaccination but may cause reluctance in getting vaccinations against other diseases in the future and non-compliance with other public health initiatives (Bardosh et al., 2022).

There are consequentialist arguments for vaccination mandates since they are expected to save lives. However, this is an equation with two sides. A potential negative to take into account is how the overall well-being and utility of unvaccinated people are diminished due to "reduced individual autonomy and infringement of bodily integrity" (Gandjour, 2022, p. 11). The harm done by forcing people to vaccinate could potentially outweigh the benefits of a vaccinated populace.

Additionally, from a deontological perspective, while COVID-19 vaccines are touted as safe with small chances of side effects, they are not zero – although a rarity, potential side effects include anaphylactic and other allergic reactions, thrombosis, Guillain-Barré Syndrome, myocarditis and pericarditis, and others (Centers for Disease Control and Prevention [CDC], 2022a). Some individuals might prefer their choice to not vaccinate. According to Hughes et al. (2021), mandating vaccination could build distrust between the public and the healthcare system that has relied on paternalistic policies before.

According to United Nations International Children's Emergency Fund (2021), an undeniably large share of people believe that vaccines are a global conspiracy to boost pharmaceutical profits. At least a part of this theory is not untrue, as there has been evidence of undeniable profit to be reaped for pharmaceutical companies. For example, Pfizer doubled its profitability in 2021 (Allen, 2022). According to Deruelle (2022), there is evidence of ill profit-and-power-driven intentions from government officials in mandating COVID-19 vaccination.

#### 3. Methodology

This section describes our chosen methods for answering our research questions. The next subsection will contain information about how we collected data using document analysis. Next, we describe our shift from document analysis to moral analysis when we begin reformulating the arguments we've gathered. Next, we explain our approach to argument evaluation. This section ends with an overview of the limitations of our research.

#### 3.1. Data collection

#### 3.1.1. Data sources

The methodology for this paper can be split into two distinct parts: document analysis for finding, identifying, and compiling arguments and the logical and normative evaluation of such compiled arguments. Bowen (2009) defines document analysis as a "systematic procedure for reviewing or evaluating documents" (p.27). Documents used in different analyses may vary – in our case, these are expected to be mainly comprised of debate transcripts from the Latvian Parliament and our own transcriptions from video recordings of meetings of the Cabinet of Ministers. Written documents are one of the key methods of qualitative research. As it is qualitative research, the data ought to be compiled and structured according to themes or patterns, in our case – political argument types (Labuschagne, 2003).

The politicians' arguments will be drawn from publicly available documents. As the main data sources, the Parliament of the Republic of Latvia was selected (further, "Saeima") and the Cabinet of Ministers (further, "Cabinet"). Creating legislation stands as one of the main functions of Saeima (Saeima, n.d.). Therefore, given its legislative power, Saeima's politicians have an impact on the public, making it an appropriate target for this type of analysis. Additionally, Latvian law allowed the Cabinet to fulfill legislative duties during state emergencies, but during the COVID-19 pandemic, a new law was passed that they reserve this right regarding COVID-19 restrictions even outside of emergencies (Saeima, 2020), which required us to look at the arguments made during the Cabinet's meetings.

Since we are analyzing the argumentative force of arguments mobilized by politicians, we will focus on the debate section of Saeima politicians' meetings. The debate transcripts of Saeima's meetings are publicly available at saeima.lv. We looked at Saeima meetings starting from January 2021 (Latvia got their first vaccine delivery on December 26<sup>th</sup> of 2020 (Kinca et al., 2020)) to December 2022, which was a total of 247 meetings. Of these 247 meetings, we ended up with a total of 150 meetings to look through for arguments by looking at the contents of the meeting and seeing whether COVID-19-related issues are discussed. This is likely due in part to the fact that vaccines were being slowly rolled out at the start of 2021 and at first, they were only available to medical staff (Kinca, 2020). We ended up compiling arguments only from 41 meetings, as the rest did not contain relevant arguments. Out of those 41 meetings, we gathered 566 unique arguments.

The approach to the Cabinet's meetings is a bit different since they are in video format publicly available on the platform YouTube and not all of the discussion is relevant to

our research. Meeting agendas are publicly available on mk.gov.lv, where one can see the questions discussed in each meeting. We looked through all 150 meeting agendas starting January 2021 to December 2022 to determine which ones include discussion of COVID-19-related issues, from which we found a total of 71 meetings potentially relevant to us, totaling up to approximately 249 hours of streaming time. Of those 71 meetings, only 23 ended up containing arguments relevant to our study. To create a set of data we could use for grouping and reformulating the arguments, we transcribed the arguments ourselves, and ended up with a total of 132 unique arguments from the Cabinet's meetings.

# 3.1.2. Analysing transcripts

There are advanced methods for debate transcript analysis that usually include creating models that capture sentiment from speech (Abercrombie & Batista-Navarro, 2018; Bhavan et al., 2019; Bhavan et al., 2020). However, our technical abilities are limited, and in our case, it is important not only to understand whether an argument is for or against a mandate but also to understand the argument itself on a deep enough level to be able to deconstruct it and reformulate it – something that could not be done with a simple model. Therefore, we resort to manually reading through the 150 meeting transcripts one by one. The workload is reduced by the transcript structure, as the Saeima debate transcripts include an introduction to topics to be covered, which will ease our use, as we will be able to skip unnecessary parts. Similarly, we also manually watched all 249 hours of the Cabinet's streamed meetings but were able to skip over certain parts where we were able to discern that COVID-19-related issues are not being discussed thanks to the meeting agendas, which have the questions ordered in the same way as they are discussed in the meeting and also note which Ministers or advisors have spoken in the discussion of a specific question.

When reading the transcripts and watching the videos, we will be using document analysis. According to Bowen (2009), document analysis is the combination of elements from content analysis and thematic analysis. Content analysis means organizing and eliciting "meaning from the data collected ... to draw realistic conclusions from it" (Bengtsson, 2016, p. 10), whereas thematic analysis is "a method for analyzing qualitative data that entails searching across a data set to identify, analyze, and report repeated patterns" (Kiger & Varpio, 2020, p. 2). Bowen (2009) describes document analysis as having three parts: "skimming (superficial examination), reading (thorough examination), and interpretation" (p. 32).

#### 3.1.3. Grouping arguments

The arguments used by both Saeima and Cabinet politicians were often produced in a manner that made it unclear what kind of vaccine mandate the argument is about – travel, education, work, finances, or leisure. When possible, we excluded arguments that clearly targeted non-work-related mandates, as well as arguments that tackled specific policy aspects that fell out of our scope, such as the question of whether vaccine mandates should also be placed on Ukrainian refugees and/or people with sufficient levels of antibodies. These questions were excluded, as they do not wrestle with the ethics of vaccine mandates themselves, but rather the specifics of their application.

The collected arguments are initially grouped by whether they are for or against the mandates. Later, we join them in rather narrow groups, which we chose to do jointly for both Saeima's and the Cabinet's arguments due to both bodies having a similar influence on the population, as well as the existing overlap between the arguments used. If arguments have essentially the same meaning with different wording, they are taken as one. For example, these artificial arguments "Since herd immunity protects a population from disease, and it can be achieved through mass vaccination, and a vaccine mandate will enhance mass vaccination, there should be a vaccine mandate" and "Since a population that has the vast majority of vaccinated people is more likely to be the solution for a pandemic, and a vaccine mandate would enhance the percentage of the population that is vaccinated, there should be a vaccine mandate" would be classified as one, since they both play on the same idea of herd immunity, but with different wordings. We initially distinguished 23 groups of arguments in favor of mandatory vaccination and 51 groups against mandatory vaccination. Those were further narrowed down, some were joined together as one group, and some were excluded entirely.

The combining of different argument groups was performed according to:

- Argument groups having similar origins/meaning (e.g. "COVID-19 is not dangerous enough to warrant strict mandates" and "Healthy people that have good immune systems should not be restricted, as they are likely not going to face severe COVID-19 symptoms").
- Some argument groups being more specific subsets of other argument groups (e.g. "There should be a vaccine mandate for school teachers, as it is crucial for the youth to study on-site and receive high-quality education come Autumn" is a

subset of "Some professions warrant a vaccine mandate, as these professions are crucial for the on-going of key processes in our society").

- 3. Some argument groups work only as premises that build stronger arguments when combined with other argument groups. (For instance, "Vaccine mandates are needed to motivate the unvaccinated to vaccinate" does not work separately, as it lacks reasoning behind why the unvaccinated should be vaccinated in the first place. However, when combined with, for example, the argument "Increasing the share of the vaccinated populace would reduce the number of deaths and critical health statuses due to COVID-19, as the vast majority of people in hospitals and lethal cases are unvaccinated", a stronger, more coherent argument is created.)
- 4. The final argument groups used in the analysis can be seen in Section 4 of this paper. In total, 8 (and 5 implicit premises) pro-vaccine mandate groups and 13 anti-vaccine mandate groups were created.

#### 3.2. Reformulating arguments

There are two key issues in collecting data that we battle with our methodology. Firstly, multiple arguments can have the same intrinsic meaning but can be worded differently. Collecting them as separate arguments isn't worthwhile. Secondly, and more importantly, due to the setting of the debates, the arguments may not be "perfect". In a political debate, politicians have little time to make a coherent argument in front of their peers in a highly stressful setting. Politicians may use calls-to-emotion and casual oversimplification (or other various fallacies) to make their point, either because of the lack of time or because fallacious arguments might do a better job at convincing people. However, we consider that when arguments are incomplete, it is because of a lack of time – we assume the best intentions. We want to avoid judging arguments that are flawed because of wording, and we want to avoid examining arguments without addressing their main points in their strongest form – which would constitute a straw man fallacy – misrepresentation, deliberate manipulation of an argument in order to make it weaker (Damer, 2009; Swatridge, 2014).

Taking the previous example of herd immunity, under a great deal of stress and shortened timeframe to make an argument, a politician might have worded the argument like this: "Since herd immunity is good, there should be a vaccine mandate". This example shows hasty argument construction, as "good" isn't the clearest way to define herd immunity, and some argumentation steps are missing (the link between the vaccine mandate and herd immunity). However, judging the argumentation skills of politicians wouldn't be as useful as the arguments themselves. Additionally, there might appear moral arguments in which implicit moral premises are missing – and they need to be made explicit to be reasoned with (Damer, 2009). However, the missing parts of arguments cannot be made explicit if they are not carefully reformulated.

We intend to fix these two issues by reformulating the arguments in the strongest possible manner (opposite to straw-manning), based on the available literature on healthcare ethics. By reformulating arguments, instead of multiple similarly worded arguments, we would obtain one argument that encompasses the intrinsic meanings of the other arguments. And in recreating these arguments in the strongest possible manner, we would be able to further analyze the arguments themselves, not just the argumentation errors made by politicians.

#### 3.3. Evaluation of arguments

Arguments typically are based on some premises, which can be either true or untrue, and in some cases, plausible or implausible if the truth cannot be known, and the premises lead to a conclusion, which can be either valid or invalid. The conclusion is valid if it logically follows from the premises, and vice versa. Combining the trueness of premises and the validity of conclusions, we can then determine the soundness of the argument. An argument is sound if both conditions stand: the premises are true, and the conclusion is valid. Logically, if either condition is not present, then the argument is unsound (Gula, 2002). Arguments can also include fallacies. Fallacies can be formal or informal. If a fallacy is formal, then there is an issue with the way the argument is structured. If it is informal, conclusions have been drawn due to incorrect inferences or there is a premise present that is untrue or weak, and this aspect has gone unnoticed (Gula, 2002).

The next sections focus on the argument evaluation process. Important to note – as there are many different types of vaccine mandates, all arguments will be evaluated by assuming a work-related mandate context, meaning, we will exclude arguments that aren't relevant to vaccination mandates for work (e.g. arguments on travel mandates). To our minds, this is also prudent, as it creates a higher burden of proof for pro-vaccine mandate arguments because this type of mandate diminishes the liberty of the unvaccinated populace the most. Similarly, weak anti-vaccine mandate arguments will likely remain weak in other contexts. On the contrary, perhaps some further discussion may be had if there are weak pro-vaccine mandate and strong anti-vaccine mandate arguments, as the strength dynamics might be

affected by less strict contexts (e.g. passing vaccination mandates as a way of rewarding the obedience of the vaccinated populace might be a stronger argument when talking about less strict mandates like access to leisure activities rather than restricting the income of the unvaccinated).

#### 3.3.1. Evaluation of soundness

The evaluation of the newly reformulated arguments happens in two steps. Firstly, the evaluation of argument soundness. A sound argument is one in which the conclusion can be derived logically from the premises, and the premises themselves are factually true. The first part refers to the validity of an argument. For example, an invalid argument would be as follows, "If it is raining, the streets are wet. The streets are wet. Therefore, it is raining". In this case, the premises are factually plausible, but the conclusion cannot be drawn from the premises, for example, it could have already stopped raining, but the streets can still be wet, they may have recently been washed by municipal services, or there may have been snow that has melted. The second part is intuitive as well, as it refers to the use of false premises, e.g. "COVID-19 is not an infectious disease". As for factuality, we intend to base our research on the latest available literature and information at the time of research. We wouldn't consider a premise factual if it is disproven now but was backed by the literature or believed at the point in time when the argument was made. There may be arguments that were sound at the time of making them but do not hold up now.

# 3.3.2. Evaluation of moral strength

Secondly, the evaluation of moral strength/convincingness. This part only follows for arguments that are both valid and factual, as it doesn't make sense to further evaluate arguments that can be refuted outright. Knowledge of basic ethical theories like consequentialism, deontology, and virtue ethics (see next section) will be useful in orienting in the "grey area" of arguments. We choose consequentialism, deontology, and virtue ethics mostly due to our limited knowledge in the ethics field, and due to these three being the main ethical theories. This evaluation is a necessary step, as there could be multiple sound arguments, but they still need to be compared to one another. We have identified grounds for discussion that may appear in arguments – e.g. the utility of herd immunity, individual rights (both the right to choose not to vaccinate and the right to a safe environment, which are at odds with one another), the riskiness of hastily developed vaccines, and so on. This part will

appear in our Discussion section. While the first part of (soundness) evaluation may be useful in showing outright flawed arguments, the second part will potentially decipher between weak, strong, and strongest arguments, and, together with part one, may provide insight into how successfully each political aisle made strong arguments for their propositions.

#### *3.3.3. Ethical theories*

Two main ethical theories will help in the classification and evaluation of arguments – consequentialism and deontology. It is important to distinguish between the theories, as the theories have significant differences (LaFollette, 1997).

To generalize, consequentialist arguments only care about the consequences – the end result – no matter the way of obtaining it. While there may be arguments about whether and to what extent immoral actions impact the utility of the result, the society we live in, trust, etc., that sort of analysis still is consequentialist, as it tries to calculate the end result and from it derives the rightness of the action itself. Deontological arguments are more focused on the action itself, the rules and principles that are followed or violated; it is not as concerned with the outcome (LaFollette, 1997). For example, deontologists will acknowledge the duties of special relationships we have with our loved ones – the fact that we are often required to do more for our loved ones than for people we do not know. Some constraints do not allow us to harm people for the outcome we desire – a key difference between consequentialism, which only looks at the end result. And lastly, virtue ethics is involved in assessing what kinds of virtues are associated with actions, e.g. courage, charitableness, and kindness (LaFollette, 1997). This short section is by no means an exhaustive explanation of ethical theories, but it helps to show how different ethical theories may be, and where the differences in arguments may lie. These theories will be revisited and broken down further under the inspection of specific arguments. SSE RIGA

#### 3.4. Limitations

We have noted some limitations to our approach. There may be some issues with data availability. We are not concerned with obtaining the main parts of our analysis, e.g. the Saeima debate transcripts because they are publicly available. However, we are unable to obtain data from Saeima commissions' meetings, since not all commissions have websites with publicly available information, and some have audio recordings of meetings, and some do not. There is no uniform process for meeting documentation, which would complicate our research even further. With our abilities and time available we concluded that it is not feasible to include the Saeima commissions' meeting data, which may result in a gap in our research. However, we believe that looking at the two main government bodies – Saeima and the Cabinet – provides great insights, nonetheless. Another gap in our research remains since the Cabinet's meetings have closed sections – parts of meetings that are held "behind closed doors" and not streamed to the public. We can see what questions they've discussed in these sections, but there is no information or data on what is being said. Unfortunately, this is a gap that cannot be improved upon.

We expect that some arguments may be worded so poorly that they cannot be properly reformulated and have to be discarded.

Once we have compiled and reformulated the available arguments, we may run into the issue of empirical disagreements between health expert opinions. For soundness evaluation, the factuality of the premises is checked. However, in the case of contradicting reports by health experts and scientists, it is difficult to comment on the soundness of the argument. This limitation will be minimized by utilizing the most recently available information.

Our research may not be as thorough as it possibly can be due to our limited knowledge in the ethics field. With the time available to us (and the amount of data we have to work with) it is not feasible to study niche ethical theories and concepts unknown to us, as we recognize their complexity and depth. This can be improved upon in further studies on the Latvian government's arguments on vaccination mandates with a stronger focus on and a more detailed analysis of the morality aspect.

#### 4. Analysis of Results

Following our methodology, larger argument groups were formed for both pro- and anti-vaccine mandates. It is important to note that some anti-vaccine mandate argument groups were outright discarded due to them being too weak or speculative to be used in further analysis (see Table 1).

Similarly, we discarded legal arguments both for and against vaccination mandates. There were two key reasons for this. Firstly, the legal arguments had many different variants which could not be grouped under one legal argument (e.g. separate sections for anticonstitutional argument and employment laws argument, and all the other legal arguments). Secondly, and more importantly, we, as students of business and economics, reasoned that evaluating legal arguments fell outside of our zone of competency. We have not studied law, and we feel the obligation to only include arguments that we are capable of successfully evaluating. Thus, all legal arguments were excluded from our analysis.

The pro-vaccine mandate argument groups that were created in the research process (these arguments will be further reformulated and evaluated in the following sections) can be found in Table 2.

The anti-vaccine mandate argument groups that were created in the research process (these arguments will be further reformulated and evaluated in the following sections) can be found in Table 3.



# SSE RIGA

Argument	Description	Reason for exclusion
Collectivization	Vaccine mandates are like collectivization tactics used in the USSR. The	This is a weak argument due to it being a fallacy of association, i.e. since
	comparison is made between people having to potentially change their	forced collectivization could be considered unethical, so can vaccine
	workplace or be left without a job in case of vaccine hesitancy and people	mandates. However, further proof is not provided for this generalization.
	being exiled to Siberia in case they did not want to work in collective	Moreover, this argument is immoral, as it fails to respect the historical
	farms.	meaning of Collectivization.
Totalitarianism	Vaccine mandates are like totalitarianism. The comparison is made due to	Weak argument due to a fallacy of association, i.e. since totalitarianism
	vaccine mandates causing fear in the unvaccinated society, similar to how	could be considered unethical, so can vaccine mandates. However, no
	totalitarian governments would strike fear in their populace to rule with	proof is provided for this generalization. Moreover, this argument is
	power over them.	immoral, as it fails to respect the historical meaning of Collectivization.
Government	Vaccine mandates are part of a larger political, social, and economic	The soundness of this argument cannot be determined, as it is purely
Conspiracy	conspiracy that is based on vaccine profiteering and forceful segregation,	speculative.
	and the destruction of the liberty of citizens.	18
Genocide	Vaccine mandates are like genocide (often a more specific comparison is	This is a weak argument due to it being a fallacy of association, i.e. since
	made with the Holocaust).	genocide could be considered unethical, so can vaccine mandates.
	Econo	However, further proof is not provided for this generalization. Moreover,
	I CONO	this argument is immoral, as it fails to respect the historical meaning of
		Genocide.
Foreign	Latvia should follow other countries' examples. Other countries (our Baltic	"Appeal to common opinion" fallacy where it is assumed that a given
Examples	neighbours – Estonia and Lithuania) have not passed such strict	claim is more truthful due to its popularity. However, the truthfulness of a
	regulations, and it is unclear why Latvia has.	claim does not necessarily follow from its popularity (Damer, 2009). The
		merits of vaccination mandates would have to be challenged, rather than
		the popularity of their implementation.

Table 1. Excluded arguments. Created by the authors.

Argument	Description
Vaccine effectiveness*	COVID-19 vaccines are effective at reducing the risk of getting vaccinated, the severity of the symptoms, and the probability of death. Being vaccinated significantly increases protection from infection and
	severe symptoms, compared to being unvaccinated. Thus, vaccines are an effective tool to use in reducing infection rates, hospitalisations, and fatalities from COVID-19.
Motivation*	Vaccination mandates for work will leave the people (at least those targeted by the mandate) with the decision either to vaccinate or leave work. Since work is an essential aspect of people's lives, these
	mandates will be effective at motivating people that have not yet gotten vaccinated and/or do not wish to get vaccinated, to get vaccinated.
Non-	Since unvaccinated people expose themselves and others to significantly more harm than vaccinated people, as well as the fact that the COVID-19 pandemic is dangerous to society and its functions, there is
discriminatory/fair*	a legitimate reason to discriminate against unvaccinated people in creating vaccination mandates.
Testing	Given that tests are more expensive than vaccines, but, more importantly, tests are not effective as a determinant of health status and do not protect from infection, they are an ineffective measure to use
ineffectiveness*	instead of vaccines.
Legitimacy of	The government has a duty to protect its people. Given that a pandemic like COVID-19 poses a significant threat to society, the government can pursue freedom-reducing policies if they are more effective
freedom-reducing	at mitigating the risks of such a health crisis.
policies*	
The right not to be	When living in a society, every individual has the right not to be harmed. This is especially true when talking about people/professions that interact with vulnerable groups (e.g. ill people, the elderly)
harmed	regularly. Not vaccinating causes harm to others, as non-vaccinated are more likely to transmit the disease further, as well as take up space in hospitals that could be used by other people in need.
Economic recovery	COVID-19 has significantly harmed Latvia's economy. One vehicle for this harm is the halting of the work of some industries. Vaccination mandates are crucial for the economic recovery of the country, as
	well as protecting the economy against further economic downturns (due to lockdowns), as some industries would be able to start working, and as further regulations given a vaccinated populace would not
	be needed.
Easing, not restricting	Instead of restrictions, vaccine mandates should be viewed as easing the restrictions currently in place. There are industries in which nobody (vaccinated and unvaccinated both) can work. Due to vaccination
	mandates, these industries can experience a lift in restrictions. Thus, vaccination mandates should be seen as an easing of restrictions, rather than as an additional restriction.
Achieving herd	Herd immunity from vaccination is the only way out of the COVID-19 pandemic and its restrictions. Reaching herd immunity from mass infections is dangerous, and should be avoided. To reach herd
immunity safely	immunity safely (relative to natural herd immunity), unvaccinated people must be motivated to get vaccinated by using various sticks and carrots.
High-risk professions	Some specific individuals working in certain industries need to be protected by vaccine mandates due to the extra risk of interacting with many people in closed spaces regularly (e.g. healthcare,
	hairdressers).
Critical professions	Some specific individuals working in certain industries need to be protected by vaccine mandates due to critical service to society provided by said industry (e.g. healthcare, education).
Vaccine safety	A popular argument used by people opposing vaccine mandates is that vaccines are unsafe. However, no cases of adverse side effects have been recorded in Latvia. Many such claims are purely works of
	disinformation, not science. Getting vaccinated is relatively less dangerous than contracting COVID-19.
Reciprocity towards	There should be an easing of restrictions for vaccinated people because they have followed the government's suggestions/regulations.
the vaccinated	

Table 2. Descriptions of arguments for vaccination mandates. Created by the authors.

\*this is an implicit premise, see Section 5.1.

discrimination live Social Capital Pas vac Free choice/personal Vac	accine mandates wrongfully discriminate against people that do not want to get vaccinated. Such vaccination mandates use cruel, unethical, and unjustified measures (e.g. restricting income, destroying velihood) in achieving their goal. Therefore, vaccination mandates are ethically wrong. assing vaccine mandates will cause tension among the population, as well as general displeasure and distrust towards the government, which in turn may cause increased resistance towards getting accinated, making this policy potentially unsuccessful at increasing vaccination uptake. accine mandates effectively force people to choose between having a job (a crucial aspect of surviving and life itself) and getting vaccinated. Vaccination mandates disrespect each person's right to choose of each person's right to choose between the each person's right to choose of each person's right to choose between the each person's right to choose between the each person's right to choose of each person's right to choose between the ea
Social Capital Pas vac Free choice/personal Vac	assing vaccine mandates will cause tension among the population, as well as general displeasure and distrust towards the government, which in turn may cause increased resistance towards getting accinated, making this policy potentially unsuccessful at increasing vaccination uptake.
Free choice/personal Vac	accinated, making this policy potentially unsuccessful at increasing vaccination uptake.
Free choice/personal Vac	accine mandates effectively force people to choose between having a job (a crucial aspect of surviving and life itself) and getting vaccinated. Vaccination mandates disrespect each person's right to choose
-	
utonomy to g	o get vaccinated or not. Moreover, free choice is a key aspect of a democratic state; such mandates would be anti-democratic. Vaccinating should be a free choice, thus, vaccination mandates are unethical.
Vaccine ineffectiveness Vac	accines are ineffective as a policy tool, as either they outright do not limit transmission and/or do not limit symptoms and hospitalisations from COVID-19, or they are not effective enough to stop the
spr	pread. Additionally, such mandates might cause an unjustified sense of security to rise among the vaccinated population which will put them at more undue risk.
Anti-paternalistic view Un	Invaccinated people only expose themselves and other unvaccinated people (who have also made the decision not to vaccinate) to harm. The government should allow people to make free choices (even if
the	ney are harmful to themselves).
Less restrictive options The	here are other, less intrusive methods of fighting the spread of COVID-19. For example, allowing people to be tested and improving the healthcare system, as well as infection-limiting infrastructure in
bui	uildings like better ventilation systems. Other examples are hybrid regulations that would allow unvaccinated people to work, however, they would have to undergo stricter epidemiological regulations (e.g.
wea	rearing face masks).
Vaccine safety The	hese mandates are based on vaccines that have not completed their testing phase, increasing the uncertainty of potential side effects. In case of negative side effects, none of the involved parties (doctors,
pha	harmaceutical companies, the government) are willing to take responsibility in case of adverse effects after receiving the vaccine. Additionally, there may be some people with contraindications to
vac	accination. It is unethical to force people to vaccinate given the increased risks from these vaccines.
Government failure Sor	ome time ago, the Cabinet politicians that are now in favour of vaccine mandates, explicitly stated that vaccination will not be mandatory. Politicians should not go against their word in passing policies
the	ney said they would not. In fact, the government has failed at raising the vaccination uptake due to undue spending, low-quality campaigns, botched vaccination logistics, and other reasons. The government
is t	trying to pass its own failures as responsibility onto the people with mandatory vaccination. The government should take accountability for its failures.
Slippery slope The	hese mandates are only the start of a slippery slope that will lead to stricter mandates and further segregation of the unvaccinated populace.
Unfair targeting of Suc	uch mandates are not justified, as COVID-19 does not pose large health risks, especially to young and healthy people, who are also targeted by these policies.
protected population	
Unintended Vac	accination mandates will lead to unintended consequences, such as purposeful infection.
consequences	SSE RIGA
Unemployment effects Lin	imiting working opportunities for the unvaccinated will raise unemployment, which will harm the economy and the ongoing of specific institutions (education, healthcare), as fewer people will work and
cor	onsumption will be lower. Further, this could worsen emigration, which has been a constant issue for Latvia.
Employer ethical Suc	uch vaccination mandates put employers into inappropriate ethical dilemmas by forcing them to decide whether to take care of their employees and their livelihoods or save money.
dilemma	

Table 3. Descriptions of arguments against vaccination mandates. Created by the authors.

#### 5. Discussion

Vaccination mandates may or may not be justified, the discussion on vaccine mandates involves ethical deliberation on the interplay between the necessity to vaccinate, the proportionality in harm generated, as well as the extent to which individual freedoms are infringed upon (O'Sullivan, 2022). These next Sections involve our reformulations and evaluations of the arguments expressed by Saeima and Minister Cabinet politicians. Out of a total of 698 unique arguments, we ended up with 21 specific arguments. There were far more arguments in opposition to vaccine mandates, which also explains why the specific groups we ended up with are 8 in favor and 13 against.

#### 5.1 Main Premises Used in Arguments For Vaccination Mandates

For more effective argument evaluation, five main arguments will resurface in various combinations as implicit premises in all the arguments in favour of mandatory vaccination. In the arguments we reviewed, they were often not stated, but it was clear that they are implied as premises for the conclusion. For example, when a politician claims that vaccine mandates should be introduced to reduce critical cases and hospitalisations, there is an implicit premise pointing towards the effectiveness of vaccines in reducing critical cases and hospitalisations. Since these specific premises would reappear so often, we believe it would be less redundant to evaluate these premises at the very beginning, since if these were to be found untrue or unfounded, some argument groups could be outright determined to be false or unfounded as well. These arguments are as follows:

- Vaccines against Covid-19 are effective at reducing infection spread and, in case of infection, the risk of severe symptoms.
- Vaccination mandates are effective at increasing vaccination uptake.
- Vaccination mandates are not discriminatory.
- Testing mandates are not an effective tool to use instead of vaccination mandates.
- The government can pursue the use of freedom-reducing policies in crisis situations.

We will evaluate each argument separately by setting the context, then providing our reformulation, followed by challenging views and facts to test the factual and moral strength of the argument.

#### 5.1.1 Vaccine Effectiveness

Initial studies in the first half of 2021 in Israel on the effectiveness of COVID-19 vaccines found that they significantly prevented infections and hospitalisations, as well as lethal outcomes, and overall decreased rates of infection in the country upon increased vaccination uptake, all suggesting that the vaccines could be a successful tool in the fight against the COVID-19 pandemic (Haas et al., 2021; Leshem & Wilder-Smith, 2021). In the same period, there was also a Harvard professor, Galit Alter, who stated that despite new variants appearing, the vaccines will still protect against the disease due to vaccines activating our immune systems in multiple ways (Powell, 2021). Additionally, there were studies conducted also in 2021 on the effect vaccination has had on the COVID-19 pandemic in the U.S. and found that it significantly reduced the number of infection cases and prevented many hospitalisations (Chen et al., 2022; Christie et al., 2021; Scobie et al., 2021). Equal outcomes from increased vaccination uptake were found in a study conducted in EEA countries also in 2021 (Sikora & Rzymski, 2022). An April 2022 study that modelled infection outcomes in different unvaccinated-vaccinated people mixing scenarios showed that vaccinated people are more protected from infection, however, the level of unvaccinated people mixing with vaccinated people impacts the safety of vaccinated people at a level "disproportionate to the fraction of unvaccinated people in the population" – a key takeaway for further sections (Fisman, Amoako, & Tuite, 2022). A newer study from January 2023 also shows that vaccines are effective at protecting against symptomatic infection with COVID-19, though the study is conducted on individuals who have also received a booster dose and it is in particular about the Omicron variant of the virus (Link-Gelles et al., 2023). Besides that, there were many more studies essentially proving the same thing since the vaccines were first developed up to this day (e.g. Eyre et al., 2022; Hause et al., 2022; Link-Gelles et al., 2022; Mohammed et al., 2022; and others), but to mention them all and each of their findings would 12 be repetitive.

Overall evidence pointed to vaccines being an effective tool for preventing infection with COVID-19 and extreme cases of the disease in the case of infection despite vaccination. This argument was implicitly stated in all arguments in favour of vaccination mandates, as it wouldn't make sense to impose such a policy unless the vaccine itself is effective at doing what it promises to. Thus, we arrive at our reformulated argument: **Premise 1:** COVID-19 vaccines are effective at reducing the risk of infection for the vaccinated (Mohammed et al., 2022).

**Premise 2:** COVID-19 vaccines are effective at reducing the risk of transmission from the vaccinated (Eyre et al., 2022).

**Premise 3:** COVID-19 vaccines are effective at reducing the severity of infection (Mohammed et al., 2022).

Conclusion: Vaccines are an effective tool for combating the COVID-19 pandemic.

Despite all the questioning evidence about the longevity of the vaccines' effectiveness and resistance against newer variants (expanded on in Section 5.3.4.), it does not disprove the fact that vaccines were, in fact, effective against infection, transmission, and severity of disease, thus not disproving any of the premises.

The conclusion does logically follow from the premises. All of the premises are independent from one another, yet each one supports the conclusion. Overall, this argument is sound. While the effectiveness of vaccines depletes against newer variants and after an increased time period since vaccination, this alone does not disprove the relative effectiveness of vaccination versus being unvaccinated. Thus, we find this argument to be strong. However, this is only an implicit premise for pro-vaccine-mandate arguments, not an argument for vaccination mandates in and by itself.

#### 5.1.2 Motivation

During the COVID-19 pandemic, the Latvian government introduced vaccination mandates for certain professions (more on the specifics of this in sections 5.2.5. and 5.2.6.) with a prohibition to work unless being fully vaccinated from COVID-19 or infected and recovered. For other professions, it was up to the employer to decide whether and which employees needed to be protected from COVID-19 to fulfil their job duties, giving them the right to fire an employee in case he did not comply. The government wished to use this type of policy to motivate more people to get vaccinated due to previously discussed slow vaccination rates and low uptake when compared to other EU/EEA countries and the EU/EEA average (see Section 2.1.2.).

Yet, a survey of 6838 Latvian educational institution employees in July 2021 revealed that more than a third are unwilling to vaccinate against COVID-19 despite the newer policies at the time (becoming effective as of September 2021) stating that you must be vaccinated to work in this field (Kuzmina, 2021). However, surveys may not be entirely reliable, since it is possible to simply state that you are willing to leave your job, but following through is far more difficult than just saying it. According to LETA (2021a), in December 2021, there were more than 600 pedagogues who had left their jobs due to the requirement to vaccinate, which, according to the newest available data from 2019 on the total number of pedagogues in Latvia (Valsts Izglītības Informācijas Sistēma, 2019), is around 14%, not more than a third, as it was found in the survey.

According to data from Nodarbinātības Valsts Aģentūra (2021), the number of unemployed persons decreased from the beginning of 2021 to the end, while the number of COVID-19 vaccine doses administered grew from around 2,260 in January to 2.55 million in December (Trading Economics, 2022). According to Oficiālās statistikas portāls (2022), the unemployment rate in Latvia in the final quarter of 2021 (during mandates) was 0.8pp smaller than in the first quarter of 2021 (before mandates). Though it is not concrete proof of people's lack of willingness to leave their jobs instead of vaccinating, this is the closest inference we can get on people's inclination to remain employed, even if it implies getting vaccinated. With this, we arrive at our reformulated argument:

**Premise 1:** Vaccination mandates for work in Latvia do not allow some professions to work unless vaccinated (Saeima, 2020).

**Premise 2:** Vaccination mandates for work in Latvia give employers (in professions where a vaccination certificate is not outright demanded) the possibility to determine the necessity for a COVID-19 certificate (Saeima, 2020).

**Premise 3:** A significant number of people in Latvia would choose to get vaccinated instead of leaving work.

Conclusion: Vaccination mandates are effective at increasing vaccination uptake.

Challenging the truthfulness of premise 3 is difficult, as it is hard to prove and disprove due to a lack of available data. However, some arguments are provided in Section 5.3.2 – the impact on vaccination uptake might be limited due to generated distrust. Some opposition is provided under section 5.3.12. Unemployment Effects, where Premise 1 is the exact opposite – "a substantial amount of people would rather lose their job or move country than get vaccinated", although it does not provide strong empirical evidence or statistics to back up this claim. While these arguments oppose each other, they aren't necessarily

mutually exclusive, as neither of them claims that the amount of vaccinated or amount unemployed will be larger than the other, but simply that it will be a substantial amount.

Premises 1 and 2 are factual claims, and we simply know them to be true due to the regulations that were in place. We cannot provide evidence to challenge Premise 3, and we do see some indication of it being true due to the lack of increase in unemployment, as well as an increase in vaccinations over the relevant time period. We find that the conclusion follows logically from the premises, as well as that the overall argument is strong. This argument is also just an implicit premise for arguments in favor of vaccination mandates and not a standalone argument.

#### 5.1.3 Non-Discriminatory/Fair

Discrimination means treating some people differently than others. Vaccination mandates are discriminatory (different levels of freedom for different people based on their vaccination status). However, not all discrimination is wrongful discrimination. For example, the necessity for two-year-olds to sit in a special car seat can be looked at as discriminatory. Smokers having to smoke in designated places is also discriminatory. What differs here is that the reason to discriminate is legitimate.

According to King & Ferraz (2021), "any interferences with basic human rights by state and non-state actors must at a minimum be in accordance with the legal principle of proportionality." (para. 32) While the principles described by the authors are primarily focused on limiting human rights, we argue that by extension they can be used to also justify differential treatment of people or discrimination.

The first test is whether the discrimination is prescribed by law. In other words, whether "it is based on clear legal authority under primary or secondary legislation, and the guidance given under the law is reasonably clear, foreseeable, and accessible to the subjects of that law." (King & Ferraz, 2021, para. 33). Vaccination mandates in Latvia were to be passed by the Minister Cabinet (secondary legislation), and to our understanding, the law was publicly available to all in the manner as described by King and Ferraz (2021).

The second test is whether the discrimination has a legitimate aim. King & Ferraz (2021) describe this in two ways – the harm unvaccinated people expose others to, and the harm unvaccinated people expose themselves to. The first option is the public health argument. A higher vaccination uptake would reduce transmissions (and possibly lead to herd immunity – Section 5.2.4)), and reduce hospitalisations (Section 5.2.1), which frees up more

resources for everybody to use, as well as protects services that are critical for the ongoing of society (Section 5.2.6). An additional argument can be made for the easing of restrictions elsewhere (a similar point is made in Section 5.2.3) due to higher vaccination uptake (King & Ferraz, 2021).

The second option described by King & Ferraz (2021) is the "paternalistic" argument. We generally use Dworkin's (1972) definition of paternalism: "the interference with a person's liberty of action justified by reasons referring exclusively to the welfare, good, happiness, needs, interests or values of the person being coerced." (p. 65) Such a policy protects unvaccinated people that put themselves in harm's way by not vaccinating and being protected by vaccines (see Section 5.1.1, as well as the similar point made in 5.2.5).

The third and fourth tests are 1) necessity – whether a measure is applicable in achieving the aim of the policy (test two), and 2) fair balance – whether the penalty that unvaccinated people bear for not vaccinating is proportional to the necessity and aim of the policy. In other words, firstly, is there scientific backing justifying the mandate, and, secondly, is the scope of the mandate justified by the current situation (King & Ferraz, 2021). The points outlined in test two and its relevant sections outline the justification for scientific backing and the proportionality of the measure used (see relevant Sections).

King & Ferraz (2021) also outline that legal risks of wrongful discrimination can arise if vaccination itself isn't accessible and affordable. These risks are not present in Latvia, as people have had the ability to vaccinate (and still do) for free. Moreover, various services were provided for people who could not get to a vaccination centre for legitimate reasons, for example, vaccination buses that drove out to more remote regions with no vaccination centres nearby and limited opportunities for transport (Āboliņš, 2021).

Additionally, people that had a legitimate reason – a medical contraindication – not to get vaccinated, were exempt from this and all other vaccination mandates (Ministru Kabinets, 2021b). Thus, people who either couldn't afford to vaccinate, transport themselves to a vaccination centre, or vaccinate at all, weren't wrongfully discriminated against. We arrive at our reformulated argument:

**Premise 1:** COVID-19 is a dangerous disease (CDC, 2022b; Miller, 2023; O'Mahoney et al., 2023).

**Premise 2:** Unvaccinated people pose substantially more risk to themselves and others by not vaccinating (Fisman et al., 2022).

**Premise 3:** Vaccines have been available and free long enough for everybody to get vaccinated.

**Premise 4:** People that have a medical condition that provides a reasonable concern about the safety of vaccination are exempt from vaccination mandates (Ministru Kabinets, 2021b).

Conclusion 1: The level of risk of harm to others and themselves by unvaccinated people justifies differential treatment of people based on their vaccination status.Conclusion 2: Vaccination mandates do not wrongfully discriminate against people

that have a legitimate reason not to be vaccinated.

**Conclusion 3:** Given the risk danger posed by COVID-19, and vaccine availability, vaccination mandates are not illegitimately discriminatory.

While a direct challenge to this is provided in Sections 5.3.1 Wrongful Discrimination and 5.3.5 Anti-paternalistic View, a general challenge as to whether vaccination mandates for work pass the four tests by King & Ferraz (2021) are outlined in all of the Sections against vaccination mandates. Given that this argument incorporates the main arguments for vaccination mandates (some being independently strong arguments), we have to conclude that this is a strong argument (although, by the same logic, with a strong counterargument as well). To remind the reader, this is not a standalone argument, but rather an implicit premise for arguments used in favor of vaccination mandates.

### 5.1.4 Testing Ineffectiveness

Initially, the Latvian government allowed people to obtain the Digital Green Certificate (for certain activities) by being fully vaccinated against COVID-19, being infected with COVID-19 and receiving, or getting tested for COVID-19 and receiving a negative result, but the test had to be done through a laboratory, and at-home tests were not accepted for this purpose. At one point, arguments appeared in the Cabinet of Ministers about how tests are not an effective determinant of a person's health status, since the negative test provides a valid certificate for 48-72 hours, and the person may get tested and receive a negative result, while being already infected but not yet having any symptoms, and then become infectious to others during that time frame when the certificate is active (Nacionālais Veselības Dienests, 2021). Meaning, the test only gives insight into the person's health at one point in time. Additionally, the test only tells you whether a person is infected, but the person remains in no

way protected against infection due to a lack of immunity against the virus. Apart from this, politicians also argued about the costs between vaccines and tests, indicating that vaccines are less costly than tests for the government, thus making it more reasonable to continue providing free vaccination against COVID-19 as a means of trying to achieve herd immunity while discontinuing the offering of free COVID-19 laboratory tests due to vaccine availability to all. According to news articles, indeed on average one COVID-19 vaccine cost the Latvian government 22 euros (Žilde-Krēvica, 2021), whereas one COVID-19 laboratory test costs anywhere from 43 to 53 euros (LETA, 2021b). Through all of this, we arrive at our reformulated argument:

**Premise 1:** Tests are more expensive than vaccines (LETA, 2021b; Žilde-Krēvica, 2021).

Premise 2: Testing is not an effective determinant of health status.Premise 3: Testing does not protect a person from infection.Conclusion: Testing mandates are not an effective tool to use instead of vaccination mandates.

However, a common argument against this is that despite tests being more expensive than vaccines, both expenses come from the government's budget, which gets its money from taxpayers. And, as we know, every working citizen is a taxpayer (or at least should be), regardless of vaccination status. Thus, there is a level of unfairness against the unvaccinated – if a person is strongly set on not vaccinating, then they have to pay for laboratory testing out of their own pocket<sup>1</sup>, while persons who are willing to vaccinate incur no direct costs for the vaccine, yet the unvaccinated persons have in part paid for it through taxes.

Another refutation to this argument is that vaccination status is also not an effective determinant of health status, since the vaccine is not 100% effective against infection, and arguably, it is even more dangerous to lift restrictions for the vaccinated and not require them to test at all, because they can more often get infected and transmit the virus without experiencing symptoms.

We were unable to find studies that test the link between mandatory vaccination and infection rates or rates of severe disease cases, however, one study by Mills and Rüttenauer

<sup>&</sup>lt;sup>1</sup> This is in reference to a period during the COVID-19 pandemic in Latvia when it was still possible to obtain a Digital Green Certificate through testing, but those who did no longer had the costs of the test paid for by the government, instead they had to pay out of their own pocket.

(2022) found that introducing mandatory vaccination increased vaccination uptake in countries that had a below-average uptake before the introduction of such a policy (which is the case of Latvia, see Section 2.1.2.). A different study by Huang et al. (2022) found that higher vaccination coverage had an inverse relationship with infection and hospitalisation rates. Accordingly, we can infer that in the case of Latvia, vaccination mandates were a more appropriate policy than less restrictive measures. Thus, this argument is strong. However, this is only an implicit premise for arguments pro-vaccine mandates, not an argument for vaccination mandates in and by itself.

# 5.1.5 Legitimacy of Freedom-Reducing Policies

According to a WHO/Europe report, "The concept of health as a public good is widely accepted, as is the fundamental duty of government to promote and protect the health of the population." (Chichevalieva, 2011, p. 11). Moreover, the government's duty to protect the health of its population is baked into the Latvian Constitution since 1998 (Satversmes Sapulce, 1998). Thus, we argue that since a disease outbreak like COVID-19 threatens public and individual health, the government has a duty to protect it.

The degree and use of individual freedom-reducing policies have been a debated topic in healthcare ethics for some time now. For one, there is a framework for evaluating human rights restricting policies laid out in Section 5.1.3 by King & Ferraz (2021). There are both consequentialist and deontological arguments for it (as will be discussed in further sections). J.S. Mill in his in his classic work argues that the prevention of harm to others is the sole reason that power (in our case, such policies) can be used (Mill, 1859) – exactly this line of reasoning is provided in Section 5.2.1. In his 2021 work, Vaccination Ethics, Giubilini stated: "Arguably, it is an individual's, a parent's, and a state's responsibility to take reasonable steps to prevent individuals from posing serious risks of harm to others." (Giubilini, 2021, p. 9), furthering the case for restriction of harm. Moreover, such measures are preferred over less restrictive ones, like testing, due to their inability to sufficiently protect public health (see Section 5.1.4). Giubilini (2021) also argues that the use of policies that are slightly more restrictive than absolutely necessary can be beneficial, as it provides more predictable and better results in terms of vaccination uptake. Thus, we arrive at our reformulated argument:

37

**Premise 1:** COVID-19 is a dangerous disease that can cause high rates of infection, hospitalisation, and death (CDC, 2023a; CDC, 2022b; Fisman et al., 2022; Miller, 2023; O'Mahoney et al., 2023; SPKC, 2022).

**Premise 2:** The government should protect its people from disease outbreaks (Chichevalieva, 2011; Satversmes Sapulce, 1998).

**Premise 3:** Given increased risk levels (health crisis), policies that reduce individual freedoms are a legitimate tool in battling pandemics.

**Conclusion 1:** The government has a duty in protecting people from the COVID-19 pandemic.

**Conclusion 2:** In trying to mitigate the increased risk levels of pandemics, the government can opt to pursue freedom-reducing policies.

The main question here is to what extent can freedom-limiting policies be used so that they are indeed preferable over other, less-restrictive options (an argument is made for such options in 5.3.6). While testing does not provide the same benefits as a vaccine, it is also less restrictive. Verweij and Dawson (2004) argue that when evaluating such a policy, the burdenbenefit ratio needs to be evaluated for all options. However, we cannot prepare a cost-benefit analysis, as the authors argue for.

Moreover, there are both deontological and consequential arguments to be made against such restrictive policies. There is a Kantian argument to be made about not using people as a means to an end (see Section 5.3.3), as well as an argument against the effectiveness of such policies due to the negative impact on trust in society (section 5.3.2).

There isn't one distinctive answer as to what extent government can pursue freedomreducing policies, that is an ethical topic of the debate itself. Thus, we conclude that this argument is strong (but with a strong counterargument as well).

#### 5.2 Arguments for vaccination mandates

Arguments in favor of vaccination mandates mainly came from the Cabinet of Ministers. There were a total of 131 arguments in favor of mandatory vaccination, which we narrowed down to eight specific argument groups. We shall outline each one by providing context, and our reformulation, followed by a challenge to the argument to test its strength (both factually and morally). One thing to note, these arguments also incorporate additional premises from 5.1., which will not be explicitly stated and remain implicit to not make the arguments even bulkier. For example, 5.2.1. could incorporate all five of the 5.1. premises.

#### 5.2.1 The Right Not to Be Harmed

Like what will be done in Section 5.3.3, but at the very same time to achieve opposite ends, deontological constraints can be used in arguments regarding vaccination mandates. Deontological constraints exist as a barrier to what a person can do, and a generally accepted concept in deontological thought is the duty to do no harm or at least minimize the harm done to others (LaFollette, 1997). In this case, an argument is made that being unvaccinated is significantly harmful to other people by virtue of not vaccinating.

To grasp the harm done to others, the extent of the potential damage caused by COVID-19 needs to be understood. While the context and importance of COVID-19 have been laid out throughout this paper, we can further structure this analysis by evaluating the key points that Verweij & Dawson (2004) lay out for the evaluation of the seriousness of the disease (both for the individual and society). Firstly, can COVID-19 result in serious implications for the individual? And secondly, does COVID-19 pose a public health problem? Most evidence shows that the answer to the former question is yes. Although certain factors make it more likely to experience severe outcomes from infection with COVID-19, it can happen to anyone, and everyone is at risk of getting infected (CDC, 2022d). Additionally, many are affected by symptoms that continue long after recovering (CDC, 2022b). As for the latter question, the answer is also yes, because COVID-19 has been declared a worldwide pandemic (WHO, n.d.) Thus, we conclude that COVID-19 is indeed a serious disease that poses a risk to society at large.

Unvaccinated people are at an increased risk of infection and serious disease. While vaccines are not perfect, they are an effective tool for mitigating the impacts caused by COVID-19 (see Section 5.1.1).

An April 2022 study, already described in Section 5.1.1, showed that when mathematically modelling various mixing between vaccinated and unvaccinated populations, vaccinated people were exposed to disproportionately higher levels of risk (Fisman et al., 2022). This outlines the direct harm unvaccinated people inflict on others. An analogy can be drawn to a hypothetical situation in which a bottle of bleach ought to be removed from a toddler so that unintentional harm to others cannot take place (Bambery et al., 2013). Insights from Section 5.1.1 also outline the indirect risk caused by unvaccinated people, as they are exposed to a higher risk of severe illness and hospitalisations (see Section 5.1.1). Given their lack of vaccination status, they put undue stress on the healthcare sector. Skelton & Forsberg (2020) argue that since, one, unvaccinated people expose others to harm, and, two, the cost of vaccinating is negligible, vaccination mandates are justified. A similar point is expressed by Giubilini (2019) – "If non-vaccination harms or risks harming others, then failing to vaccinate is as bad as positively doing something that harms or risks harming others" (p. 36).

In addition to increasing vaccination uptake (see Section 5.1.2), such vaccination mandates would also limit interactions between vaccinated and unvaccinated people at work, both effects protecting the right not to be harmed. Thus, we arrive at our reformulated argument:

**Premise 1:** Each person has the right to live without being unnecessarily harmed by others.

Premise 2: Being unvaccinated is significantly harmful to other people due to infection risk relative to that of being vaccinated due to the direct effect of infection and the indirect effect of undue stress inflicted on the healthcare sector.
Premise 3: The government should protect the right not to be harmed of its people.
Conclusion: The government should protect people's right to life by imposing vaccination mandates.

Premise 2 is empirical and cannot easily be challenged. The main challenge in this argument originates from the question of to what extent can people's right not to be harmed infringe on other people's right to individual autonomy. Giubilini (2021) agrees that this argument is weaker for autonomous individuals who can vaccinate to protect themselves. Argumentation for this is provided in Section 5.3.3. Literature in healthcare ethics tends to agree that both rights are not absolute, and some trade-off needs to be evaluated based on the specific situation and its variables like the infection rates, the dangerousness of the variant, and so on. The argument for the right not to be harmed would be weaker with an Omicron-like variant that may be more infectious, but also less dangerous, and the infection rates tend to become more similar for the vaccinated and unvaccinated people (Curley, 2022). An additional argument is made by Saeima politicians that the lack of vaccine safety also infringes on people's right not to be harmed (expressed in Section 5.3.7). For example, J.

Stepaņenko – "If we talk about human rights, here, of course, Article 116 of the Constitution is very widely applied, which limits a certain set of human rights, which, unfortunately, honorable colleagues, honorable authors of the bill, are not the human right to life and the human right to health. Because in this case, we're talking about vaccines that have been granted provisional approval, and we're talking about vaccines for which every patient signs an informed consent and a waiver of any claims in bad cases." (Stepaṇenko, 2021)

Overall, we conclude that this is a strong argument, but so is the counterargument. Continuous evaluation of the pandemic – infection rates, mortality rates, stress on the healthcare system, the safety of the vaccine, etc. – needs to happen to ultimately arrive at a clearer picture of which of these two conflicting principles takes precedence.

#### 5.2.2 Economic Recovery

During COVID-19 spikes in Latvia, the work of some industries like beauty care, hairdressers, and shops, had to be halted or severely affected (Ministru Kabinets, 2020a). According to the International Monetary Fund [IMF] (2020), lockdowns, like were introduced during COVID-19, are "powerful instruments to reduce infections" (p. 76), but also produce significant negative effects on the economy (the study shows a negative effect on pre-pandemic forecasts of GDP relative to the measures imposed). Moreover, it is found that voluntary social distancing that comes with high infection rates also produces a negative effect on the economy.

Bauer et al. (2021) modelled restriction relaxing scenarios under different vaccination uptakes and found that the rate at which it is optimal to lift restrictions is constrained to the level of vaccination uptake. Moreover, the authors claim that for evading a wave entirely, a high vaccination uptake is crucial. Arbel and Pliskin (2022) find that vaccination campaigns are much more cost-effective in preventing deaths than lockdowns.

These studies show that, firstly, vaccination campaigns are much cheaper than lockdowns in terms of lives saved, and, secondly, given that these campaigns work (see Section 5.1.2), they may evade costly lockdowns that have significant negative impacts on the economy altogether. Thus, we arrive at our reformulated argument:

**Premise 1:** Given a spike in infections, some industries would have to be halted (e.g. shops, hairdressers).

**Premise 2:** Halting the ongoing of whole industries has a negative effect on the economy (IMF, 2020).

**Premise 3:** High vaccination uptake could prevent an infection spike from occurring (Chen et al., 2022; Christie et al., 2021; Scobie et al., 2021).

**Conclusion:** Vaccination mandates should be introduced as a means to prevent unnecessary economic contractions.

Note that all the premises of this argument need to be true in order to make this argument sound. Premise 1 reflects what actually was observed in Latvia (Ministru Kabinets, 2020a), but the question of whether the freezing of industries is significantly beneficial to fighting the pandemic remains to be answered. For one, there is some evidence that lockdowns aren't effective at fighting the COVID-19 outbreak (Herby, Jonung, Hanke, 2022; Spiliopoulos, 2022) and the cost of lockdowns is higher than the benefits incurred from them (Yanovskiy & Socol, 2022). Thus, while Premise 2 is true, if Premise 1 is untrue, the whole argument falls apart.

The trueness of Premise 3 can also come into question, as it isn't clear how effective high vaccination uptakes are at preventing infection spikes. Some studies in countries with high vaccination uptake show that breakthrough infections at quite high rates are possible (e.g. Feng et al., 2022). Once again, if Premise 3 is untrue, the argument is not sound.

Moreover, since this is a consequential argument, some consequential opposition is provided by Section 5.3.12, which argues that the vaccination mandates will increase unemployment levels that, in turn, would hurt the economy (however, we found little evidence for the unemployment increase).

We conclude that this argument is unsound due to limited evidence for the necessity of lockdowns and the effectiveness of high vaccination uptake in containing COVID-19 outbreaks.

# 5.2.3 Easing, not Restricting

To restate what was touched upon in the previous argument, during COVID-19 spikes in Latvia, the work of some industries like beauty care, hairdressers, and shops, had to be halted or severely affected (Ministru Kabinets, 2020a). The proposed vaccination mandates would allow these people to work, given that they are vaccinated (Saeima, 2020).

This constitutes something resembling a Pareto improvement – "if each individual prefers x to y, then x is socially better than y" (Kornhauser, 2022). Of course, this is not a perfect Pareto improvement due to unvaccinated individuals being unhappy due to the unequal distribution of benefits (see Section 5.3.11). However, from a purely technical aspect, this is a Pareto improvement, as the total number of working people would increase with minimal risks due to the specifics of vaccine effectiveness (Section 5.1.1).

Therefore, we can look at vaccination mandates as an easing mechanism, instead of further restrictions, as these restrictions had already taken place. Thus, we arrive at our reformulated argument:

**Premise 1:** For many professions (both vaccinated and unvaccinated), work has been halted due to COVID-19.

**Premise 2:** By introducing vaccination mandates for work, the vaccinated populace will be able to work.

**Conclusion 1:** Vaccination mandates aren't a restriction, but rather an easing mechanism.

**Conclusion 2:** Where it is possible to do so without generating disproportional risk, easing some restrictions are better than keeping all restrictions.

**Conclusion 3:** Vaccination mandates should be introduced.

There are some key issues with this line of logic. Firstly, as is stated in the argument itself, this is not a Pareto improvement, as the damage to the social capital from the use of discriminatory policies needs to be taken into account (see Sections 5.3.1 and 5.3.2). Many unvaccinated people were completely opposed to mandatory vaccination (Pētījumu Centrs SKDS, 2021b).

By the logic of this argument, a case can be made for a reduction in restrictions across the table, as some studies have found that lockdowns aren't an effective way of managing the COVID-19 pandemic (Yanovskiy & Socol, 2022; Spiliopoulos, 2022). Bavli, Sutton, & Galea (2020) state that such lockdowns can produce significant harm to society – even fatalities from such an economic contraction, as well as negative effects on mental health, health issues due to delayed treatments, etc. Thus, by lifting restrictions on work, everybody would at least be better off due to the ineffectiveness and harmfulness of lockdowns.

But most importantly, Conclusion 1 is not supported by its Premises due to a couple of facts. Firstly, even though work was halted for many industries, they still had a working

contract and received pay (Ministru Kabinets, 2020b). Now, due to this policy change, people will, in fact, lose their jobs and their salaries. This argument only takes into account the self-actualising aspect of working, but completely disregards the fact that unvaccinated people will not be able to receive a working wage. Furthermore, the new policy also allows employers the right to fire employees who are not vaccinated (Saeima, 2020) – this point stands for industries that could work during lockdowns as well. Thus, the idea of the policy working as an easing mechanism rather than a restriction is factually flawed, and the whole argument, therefore, is unsound.

)LM

# 5.2.4 Achieving Herd Immunity Safely

When herd immunity is achieved within a society, it remains protected against the disease because the possible spread of it is limited due to most high immunization, thus making it a public good. Moreover, it protects people that cannot get vaccinated due to contraindications (Verweij, 2022). Each of those persons has gained their immunity from the disease in one of two ways – through vaccination or infection (also known as natural immunity). As mentioned previously, vaccination carries few risks with it, however, infection can lead to unforeseen outcomes (which is discussed in more detail in Section 5.3.11) such as severe disease at the time of infection and prolonged health issues after recovery. Additionally, a study by Cavanaugh et al. (2021) found that persons with natural immunity from COVID-19 were over two times more likely to get reinfected when compared to fully vaccinated persons' chances of infection.

Vaccination refusal can be seen as freeriding, as the benefits of herd immunity would be shared among all members of society (Verweij, 2022; van den Hoven, 2012), hence one could argue that in order to arrive at a just distribution of costs and benefits, as described by Verweij & Dawson (2004), vaccination mandates that cover a substantial amount of population (if not all) are justifiable. Giubilini (2019; 2021) and Giubilini, Douglas, and Savulescu (2018) argue that collective vaccination programs are justifiable on the grounds of fairness, as the costs of achieving said herd immunity should be distributed fairly. Giubilini et al. (2018) also argue that even if the marginal effect of each vaccination may be imperceivable, unvaccinated people are still morally obligated to add to the social benefit of herd immunity. In other words, imposing collective action may be justifiable due to collective benefit.

44

This argument was used rather often in the meetings of the Cabinet of Ministers. The reformulated argument is as follows:

**Premise 1:** Herd immunity protects a society from high infection and death rates (Desai & Majumder, 2020).

**Premise 2:** High vaccination uptake is one way of achieving herd immunity from COVID-19 (Mayo Clinic Staff, n.d.-a).

**Premise 3:** The other way - through natural immunity (getting sick) – is dangerous due to unpredictable outcomes (Mayo Clinic Staff, n.d.-a).

**Conclusion:** The government should impose vaccination mandates, as it is the only safe way to achieve herd immunity.

There is empirical evidence of waning vaccine effectiveness (Feikin et al., 2022; Ferdinands et al., 2022), thus Premise 2 can be questioned. Additionally, there is also empirical evidence from countries with high vaccination uptake (that should, in theory, have achieved herd immunity), like Israel, still experiencing high infection rates and outbreaks (Feng et al., 2022), although this can be interpreted as questioning Premise 1, this is more so an additional argument against Premise 2. According to Morens, Flokers, and Fauci (2022), the traditional idea of herd immunity might not apply to COVID-19 due to mutations and new variants, the lack of long-term immunity, and the occurrence of transmission without symptoms. If the premises were true, there is still the argument from Section 5.3.2, which questions whether such policies do increase vaccination uptake (although we do not find evidence for this), and deontological arguments, e.g. Section 5.3.3 that would insist that people should not be sacrificed for the greater good of society.

Additionally, Bradley and Navin (2021) insist that, as people who opt not to vaccinate generally don't do so with the intention of benefiting from herd immunity without contributing to it, but rather due to their lack of belief in the vaccines themselves; therefore going unvaccinated should not be treated as freeriding from herd immunity. While Verweij (2022) agrees that not vaccinating is a form of freeriding, the author also claims that it does not constitute unfairness as described by, for example, Giubilini (2019), because the people who experience the costs of vaccinating also enjoy the individual benefits of vaccinating. In other words, while herd immunity is a public good, it is not strictly a private good due to individual benefits.

While there may be ethical arguments for herd immunity, we conclude that there isn't sufficient evidence for the viability of achieving long-lasting herd immunity from vaccination in the case of COVID-19, thus, this argument is weak.

#### 5.2.5 High-Risk Professions

COVID-19, as an infectious disease, is highly contagious (Slimību Profilakses un Kontroles Centrs [SPKC], 2022). People, who work in jobs that necessitate increased interactions with people in general (especially infected people, like in hospitals; especially indoors), are more exposed to the risk of infection (Hengel et al., 2022). A study done by Lu (2020) found that the most at-risk professions are typically in the healthcare, education, retail, social work, beauty care, and restaurant industries. This is reflected in the ensuing vaccination policies developed by the Latvian government – professions that reflect the aforementioned characteristics could only be filled by people with a vaccination certificate.

This argument has two sides to it – the paternalistic argument (people in these professions should vaccinate to protect themselves, as they are put under increased risk), and the right not to be harmed (people in these professions should vaccinate to protect the people they engage with). Regarding the latter, Section 5.2.1 provides an argument for the right not to be harmed, so additional repetition is not necessary, however, an additional point needs to be mentioned. This argument is especially strong for healthcare workers, and slightly weaker for other professions due to the fact that people, as clients of these services, can decide not to receive these services, thus, protecting themselves, with the clear exception of healthcare services. Latvian Prime Minister Krišjānis Kariņš made this point during a Cabinet of Ministers' meeting on April 8th, 2021 – "a person can choose to do or not do a lot of things - he can go outside the house, or not, but if a person happens to get sick, there [isn't] actually [a] choice to go to a medical institution. And that is perhaps one quite important difference here, which speaks of those contacts, forced contacts, contacts that the patient cannot avoid." (Ministru Kabinets, 2021a).

There are arguments to be made for the paternalistic aspect as well. People may not know what is best for them, so providing them with a nudge may be beneficial – this is the idea of libertarian paternalism (Dworkin, 2020). In our case, vaccination is not compulsory, as people still have a choice not to vaccinate (however disincentivised it may be). Another case can be made on consequentialist grounds, that if more good than harm is produced by coercing people, then that is morally good, as consequentialism only concerns itself with the results of

decisions, not how these decisions are made (LaFollette, 1997; Dworkin, 2020). Given the vaccine effectiveness discussed in 5.1.1, there is an empirical claim that vaccinating produces better outcomes for the individual (as well as for society). Thus, we arrive at our reformulated argument:

**Premise 1:** There are professions (e.g. healthcare workers) that are more prone to become infected due to higher frequency of interactions with infected people (e.g. in hospitals), working in close proximity to other people indoors (e.g. hairdresser), and/or interacting with a high number of people for whom their health status is unknown (e.g. retail) (Hengel et al., 2022; Lu, 2020).

**Premise 2:** Being unvaccinated increases the vulnerability of the people working in these professions (Johnson et al., 2023).

**Premise 3:** Being unvaccinated exposes other people to more danger (Fisman et al., 2022).

**Premise 4:** People working in high-risk professions should be protected.

**Premise 5:** Clients of people working in high-risk professions should be protected. **Conclusion:** Vaccination mandates should be introduced for specific professions at a higher risk of infection.

We find that Premises 1, 2, and 3 are factual premises and all are true. Additionally, all people have the right to not be harmed (as we explore in Section 5.2.1), therefore we accept Premise 4 is weak due to it having strictly paternalistic grounds. The main challenge to paternalistic vaccination mandate policies is provided in Section 5.3.5. Although the Anti-paternalistic View argument itself isn't strong against all vaccination mandates, it provides very strong argumentation against paternalistic policies, as reflected in Premise 4. However, this argument still remains strong due to Premise 5 – the argument for why vaccination mandates should be implemented for high-risk professions is strong if justified by the harm done to clients/colleagues, but not the person himself. We conclude that the argument is strong with a clarification that Premise 4 is weak.

# 5.2.6 Critical Professions

This argument plays on a similar, but slightly different notion than the one in the previous section. Some professions provide services that are essential for society at large. These are

47

professions that cannot afford not to work, as the services they provide are either an irreplaceable part of our day-to-day lives, and/or the removal of these services would produce significant harm to society even in the short run. An example of the former aspect would be supermarket workers, as it would not be prudent to assume that people keep a large stock of food and other consumables at home to last a quarantine. As for the latter, think of critical services, such as firemen, policemen, healthcare workers, and so on. Examples of some essential professions/industries are provided by European Institute for Gender Equality (n.d.) and McNicholas & Poydock (2020).

As discussed previously (e.g. Section 5.2.1), being unvaccinated poses an additional risk for the people working in these professions (Hengel et al., 2022; Johnson et al., 2023; Lu, 2020). This can be treated as an extension of "The Right Not to Be Harmed" argument, but rather specified for essential workers, due to the harm to society generated from underprovision of essential services that would arise in case of an outbreak in these industries. Thus, these professions must be protected, but not necessarily for their own good (as was discussed in the previous section). Thus, we arrive at our reformulated argument:

**Premise 1:** Some professions provide services that are essential for society at large, e.g. healthcare workers, supermarket cashiers, teachers, and others (European Institute for Gender Equality, n.d.; McNicholas & Poydock, 2020).

**Premise 2:** Being unvaccinated poses an additional risk for the people working in these professions (Johnson et al., 2023).

**Premise 3:** A high number of infections within these professions could cause severe consequences, e.g. a high number of infected healthcare workers could lead to many critical patients being untreated simply due to a staff shortage.

**Conclusion:** Vaccination mandates should be introduced for professions providing essential services for society at large.

There may be some probability that introducing vaccination mandates in these industries would cause additional unemployment, which the industries are already struggling with. For example, the education system has been struggling with a shortage of teachers for multiple years already, with some school directors stating that some classes simply won't be taught to the students due to a shortage of teachers (Dēvica, 2021; Zvērs, 2022). The picture is similar in the healthcare industry, with all regional hospitals feeling the effects of a lack of a skilled workforce to some extent (Feldmanis, 2021c). Increased unemployment in these industries would also affect their ability to continue providing the services needed by society. As we discuss in Section 5.3.6, it is possible to use less restrictive measures to protect these professions as well, but logically, if all less-restrictive options are exhausted and the infection trend does not improve, then mandatory vaccination does become the only feasible way of protecting these critical professions from shutting down. Additionally, it seems unlikely to us that a significant amount of people would truly be willing to lose their jobs instead of getting vaccinated (which we've explored further in Sections 5.1.2, 5.3.11, and 5.3.12), therefore the stress caused by a handful of people leaving their job in these critical industries would be incomparable to that of high infection rates within one given critical industry.

We accept all premises to be true, and we cannot refute them. The conclusion logically follows from the premises. Therefore, the argument is sound and strong.

#### 5.2.7 Vaccine Safety

This argument was mainly used to refute other politicians' arguments about the lack of safety in COVID-19 vaccines (which we discuss in Section 5.3.7). However, most data shows that COVID-19 vaccines are safe, and adverse reactions are uncommon. More specifically, adverse reactions include thrombosis with thrombocytopenia syndrome (TTS for short), which causes large blood clots and a low platelet count in the blood, platelets being the blood cells that help form clots, Guillain-Barré Syndrome, which makes the "body's immune system damage nerve cells, causing muscle weakness and sometimes paralysis." (CDC, 2023b, para. 11), and myocarditis and pericarditis, which are both different types of inflammation in the heart. However, incidences of all of these complications are 5 or fewer cases per one million doses administered (CDC, 2023b). It is worth mentioning that many individuals experience some form of reaction soon after receiving the vaccine, yet this is a normal immune response and these symptoms (fever, headache, pain or swelling at the injection site, fatigue) relieve after a few days (CDC, 2022c). Given the rarity of serious health complications caused by vaccination, and the various risks associated with contracting COVID-19 (described in more detail in Section 5.3.11), as well as the high rate of hospitalisations and death in Latvia related to COVID-19 infection, which, if we transform it to be more comparable to the data on vaccination, would be 2,300 hospitalisations per one million inhabitants and 281 deaths per one million inhabitants (SPKC, 2022), one can say that on average it is safer for an individual to vaccinate than get infected. Thus, we arrive at our reformulated argument:

**Premise 1:** There is a lack of evidence about the prevalence of health risks and severe side effects that vaccines may produce (CDC, 2023b).

**Premise 2:** On the other hand, science shows that contracting COVID-19 may produce severe health risks (CDC, 2022b; Miller, 2023; O'Mahoney et al., 2023). **Premise 3:** Contracting COVID-19 produces more fatalities and severe symptoms than vaccination (CDC, 2023b; SPKC, 2022).

**Conclusion 1:** The probability of contracting COVID-19 and experiencing severe symptoms/death is significantly larger than the probability of experiencing severe symptoms/death after vaccination.

**Conclusion 2:** Vaccination mandates should be introduced.

We are unable to refute Premise 1 due to the findings described above. It is true that severe health risks associated with COVID-19 vaccination are extremely rare, thus we accept this premise to be true. The same goes for Premises 2 and 3, we also accept them as being true. Thus, we can accept Conclusion 1, since it follows logically from the three premises.

However, Conclusion 2 is more complicated. Verweij and Dawson (2004) outline seven principles of collective vaccination programs, the second one describing how vaccines should be held to high standards of safety because they are being administered to "healthy individuals as a measure to prevent possible future harm" (p. 3123). The basis for this is that it would be unlikely for a healthy individual to become seriously ill from the disease that the government is trying to immunize its population from. Therefore, the burden of justifying vaccine mandates is higher. However, as we describe under Section 5.3.10, even healthy individuals are at risk of severe complications after infection with COVID-19 (Cunningham et al., 2020; Office for National Statistics, 2023). The question is where does the right to individual autonomy take precedence (Section 5.3.3) over the expected utility (comparable safety) of vaccinating versus not vaccinating. What is more, since this argument argues purely from a paternalistic standpoint – that vaccination mandates ought to be introduced in order to protect the person to be vaccinated, a strong counterargument is provided in Section 5.3.5 Anti-paternalistic View, which does not admit that the government can mandate vaccination for autonomous adults with only a paternalistic justification. Thus, we conclude that this argument is weak, although sound.

50

#### 5.2.8 Necessity for Reciprocal Policies

The Latvian government, more specifically, the Ministry of Health had recommended getting vaccinated quite some time before the mandatory vaccination policy took place (e.g. Veselības ministrija, 2021a; Veselības ministrija, 2021b). A portion of society did indeed get vaccinated already in the summer of 2021 when vaccines had become available for everybody that wanted to get one (Anstrate, 2021a). Whether or not they did it because of the government's specific recommendation or out of their own desire, it is impossible to know, however, we can say that they at least acted per what the government had recommended them to do – they were compliant.

Rewarding good behaviour – being reciprocal – may be beneficial in generating necessary outcomes, as "a sizeable proportion of economic actors act on considerations of reciprocity" (Fehr & Gächter, 2000, p. 178). Given this, one could argue that compliance ought to be rewarded in order to generate more compliance – it could prove valuable in having a society that follows government regulations, especially during a pandemic. In our case, the reward – the reciprocal mechanism – is allowing people to go to work, as many people were not able to go to work due to restricted industries (see Section 5.2.3). Thus, we arrive at our reformulated argument:

**Premise 1:** Some part of the population obediently follows the government's regulations and suggestions for stopping the pandemic by getting vaccinated. **Premise 2:** Reciprocal policies can be useful in promoting compliance with government regulations.

**Premise 3:** Vaccination mandates will benefit vaccinated people, as they will be able to return to work.

**Conclusion:** Vaccination mandates should be introduced as a reciprocal mechanism towards the people that follow the government's recommendations.

From the consequentialist perspective, this could in fact lead to decreased compliance from unvaccinated people (see Section 5.3.2).

From the deontological perspective, the right to individual freedoms and autonomy would have to be prioritised over reciprocal relationships. It is expressed in Kantian theory, which expresses the necessity to not treat people as a means to an end (LaFollette, 1997). In this government-person relationship, the government is taking away some people's freedom in order to be reciprocal towards a different group of people. This is unjustifiable and goes against the principle of not doing harm (LaFollette, 1997).

Moreover, expressing this viewpoint diminishes and damages the reason why people should vaccinate in the first place – to protect themselves and others. Arguably, this builds a perverse notion that the reason people should vaccinate is to benefit from governmental policy. This claim damages the way people look at vaccination, as was expressed by some Saeima members, e.g. R. Petraviča – "Vaccination must be purely voluntary. And it must not be turned into a bribery campaign by offering gifts, or money, or by implementing any threat campaigns. First, it is unethical. It's like setting a rich table and expecting a vegetarian to grab a piece of meat just because it looks tempting or because it's free." (Petraviča, 2021) We conclude that this argument is weak due to strong deontological counterarguments.

## 5.3 Arguments against vaccination mandates

In the following sections, we introduce our reformulated arguments against vaccine mandates, which are evaluated exactly the same way as in Sections 5.1 and 5.2. We evaluate 13 specific argument groups against mandatory vaccination, which came from 567 unique arguments that we had collected.

#### 5.3.1 Wrongful Discrimination

As discussed throughout this paper, vaccination mandates prohibit unvaccinated people from participating in activities that were available to them in a non-pandemic context, while allowing those same activities to vaccinated people. In our case, we are specifically interested in mandates that may prohibit an individual from working due to a lack of vaccination.

For this argument, the same framework by King & Ferraz (2021) can be used, as described in Section 5.1.3, to argue that the discriminatory freedom-reducing policies developed for unvaccinated people did not possess a legitimate reason for such discrimination. Similarly to Section 5.1.3, this argument will encompass different arguments used throughout this paper to build a case for the lack of legitimate reason, and so the strength of this argument will also depend on the evaluation of the individual arguments supporting it.

One could argue against the notion that there is a legitimate reason on three grounds – the second, third, and fourth tests described by King & Ferraz (2021), as described in Section 5.1.3. First, the aim – stop unvaccinated people from harming others and themselves. One

could argue that unvaccinated people predominantly harm themselves and other unvaccinated people, making it a paternalistic policy instead. Section 5.3.5 expands on this logic and argues against such policies.

Second, arguments that are described in Sections 5.3.2, 5.3.4, and 5.3.12 question whether vaccination mandates are the tool to use in limiting harm, as vaccination mandates generate distrust that will lead to limited vaccination uptake increase and lack of compliance, and even if vaccination uptake were to increase, vaccines are ineffective at protecting people from COVID-19.

Finally, the test of proportionality. One could argue that the penalty put on unvaccinated people outweighs the benefit of increased vaccination uptake and limiting social interactions between unvaccinated people. The reasoning for such an argument is presented in numerous sections – Section 5.3.3 argues that the right to choose is both an intrinsic and democratic value; Section 5.3.6 points out that other less restrictive options could be used instead, and potentially grant a more justifiable proportionality between the restrictiveness and the benefit of a policy; Section 5.3.7 shows that there is a higher burden of justification for such policies, as vaccines have a level of unsafety to them; Section 5.3.8 argues that this policy undermines politicians' duties of consistency and responsibility; Section 5.3.10 points out that, arguably, is unjust in its distribution of burdens and benefits; and Section 5.3.12 argues from a consequentialist standpoint that the costs of increasing unemployment would make this policy more harmful than beneficial.

Thus, we finally arrive at our reformulated argument:

**Premise 1:** Vaccination mandates prohibit unvaccinated people from participating in activities that were available to them in a non-pandemic context while allowing those same activities to vaccinated people.

Premise 2: Doing so without a legitimate reason is wrongfully discriminatory.Premise 3: The COVID-19 pandemic does not constitute a legitimate reason to discriminate against unvaccinated people.

**Conclusion:** Vaccination mandates wrongfully discriminate against people who decide not to vaccinate.

Once again, as this argument pivots around the idea of legitimate/illegitimate reason to discriminate (and infringe on freedoms), the counterargument is trying to prove that there, in fact, is a legitimate reason to discriminate. Section 5.1.3 provides a thorough argument for

this. Thus, the main argument hinges on whether Premise 3 is true or not, which itself is up to ethical debate. Thus, we conclude that this is a strong argument (with a strong counterargument as well).

#### 5.3.2 Social Capital

As discussed throughout this paper, regardless of whether this discrimination is just or unjust (Section 5.1.3 versus Section 5.3.1), vaccination mandates discriminate against unvaccinated people by restricting their freedom.

From a consequentialist basis, an argument can be made that such discriminatory vaccination mandates will generate more harm than good, and thus should not be implemented. For this argument, we need to define social capital. We use the definition provided by Encyclopaedia Britannica writer Margarita Poteyeva – "Social capital revolves around three dimensions: interconnected networks of relationships between individuals and groups (social ties or social participation), levels of trust that characterize these ties, and resources or benefits that are both gained and transferred by virtue of social ties and social participation." (Poteyeva, 2018). Bardosh et al. (2022) provide a framework that outlines the harm that these policies might cause, like the generation of distrust in the government, increased inequality, and others. The authors state that "these policies are likely to entrench distrust and provoke reactance—a motivation to counter an unreasonable threat to one's freedom." (p.3). In this specific argument, the damage would be done to the social capital (trust aspect) of Latvia, which is already quite low (OECD, n.d.).

There are many negative externalities produced by low social capital, such as lower compliance with the law and higher economic insecurity (United Nations [UN], 2021). Additionally, this may result in a negative effect on vaccination uptake due to diminished compliance, which also plays out in individuals not partaking in activities that limit the spread of the infection (Sprengholz, Betsch, & Böhm, 2021; Sprengholz et al., 2022). This is a key takeaway, as an increase in vaccination uptake is one of the key results that this policy wishes to achieve. Porat et al. (2021) find that the introduction of vaccination passports (vaccination mandates) could impair the people's willingness to comply with the government's recommendations to vaccinate. Thus, we arrive at our reformulated argument:

**Premise 1:** Vaccination mandates enforce different rules for the unvaccinated and vaccinated populations.

Premise 2: Vaccination mandates may negatively impact the social capital by creating a sense of division among the population that will result in lower trust in the government, and less compliance with regulations (Bardosh et al., 2022; UN, 2021).
Premise 3: The social capital in Latvia is already quite low (OECD, n.d.).
Conclusion 1: Vaccination mandates may fail to increase vaccination uptake (i.e. more people will resist instead of getting vaccinated due to low social capital).
Conclusion 2: This may result in the negative side effects on social capital being larger than the positive gains from the vaccination mandates due to large negative externalities and reduced vaccination uptake.

Conclusion 3: Vaccination mandates should not be introduced.

In this argument, Premises 1, 2 & 3 may be accepted, but Conclusion 1 still might fail to be accepted. In fact, there is some empirical evidence to the contrary of Conclusion 1. As we examined in Sections 5.1.2 and will in Section 5.3.12 when vaccination mandates for work were introduced, one could have expected that unemployment levels will go up, yet instead they increased, which we've used as a proxy for people's willingness to vaccinate under mandatory vaccination. Given the limited effect of avoiding vaccination, the other negative effect that needs to be evaluated is the damage done to the social capital in Latvia. Arguably, healthcare takes precedence over social capital, especially during pandemics like COVID-19. Thus, the potential damage to be done to social capital has arguably a more limited downside than what can be done onto healthcare.

The overall argument is a consequentialist one – the weighing of the benefits from a high vaccination uptake (which is challenged, but not strongly) with the downside of low social capital. With limited evidence for net negative effects on society, we conclude that this argument is weak.

# SSE RIGA

## 5.3.3 Free Choice/Personal Autonomy

Politicians would also use the argument that one of the human rights that is protected by democracy is freedom of thought (UN, n.d.; UN, 1948). Although many arguments for vaccination mandates (see Section 5.2.4, for example) arise from the notion of social responsibility and benefits to society that arise from high vaccination uptake, a conflicting thought is that people have intrinsic rights that must be protected, i.e. people shouldn't be treated simply as parts of society, but also as individuals. Moreover, it is important to respect individual autonomy and the right to choose (Verweij & Dawson, 2004). This is a deontological thought, as defined by LaFollette (1997) (definition of deontology) – "it denies that the rightness of an act depends solely on the amount of good produced" (p. 39). Moreover, vaccination mandates go against Kantian deontological perspectives – "Now I say that the human being and in general every rational being exists as an end in itself, not merely as a means to be used by this or that will at its discretion; instead he must in all his actions, whether directed to himself or also to other rational beings, always be regarded at the same time as an end" (Kant, 1998, p.37), as the act of forcing people to vaccinate would treat individuals as means toward a common end.

Although individuals still have the "right to choose", as our topic is mandatory, not compulsory vaccination, this right is at the very least severely impaired if not completely undermined due to the design of work mandates. Working is a significant aspect of people's lives, not only as a right of self-actualisation but more importantly, a way of covering physiological and safety needs, which are the primary needs that drive human motivation (Maslow, 1954). We argue that a threat to primary needs can be treated as a strong form of coercion.

One can argue that vaccination mandates undermine people's right to choose, and treat people as a means to an end, and, thus, are unethical (at least under Kantian deontological thought) and anti-democratic. With this, we arrive at our reformulated argument:

**Premise 1:** In democratic states, people have the freedom of thought (UN, n.d.; UN, 1948).

**Premise 2:** In addition, they have the right to make free decisions on what medical procedures to undertake (Abraham & Abraham, 2021; Saeima, 2009).

**Premise 3:** Undermining this right to choose (e.g. by using coercion) is immoral and unethical (Kant, 1998).

**Premise 4:** Vaccination mandates that prohibit working unless you are vaccinated can be perceived as coercion affecting the people's right to choose.

Conclusion 1: Vaccination mandates undermine people's right to choose.

Conclusion 2: Vaccination mandates are anti-democratic, immoral, and unethical.

The obvious challenge to this deontological claim is another deontological claim of the right not to be harmed by others (Section 5.2.1). Similarly, rules that prohibit drunk driving

reduce autonomy, but they also protect people from harm (both the drunk driver and other people engaging in traffic). This is a good example because it captures both the right not to be harmed by others and the value of paternalistic policies.

Although not a perfect alternative, Premise 4 can be addressed by saying that at least some people could transfer to other workplaces/professions that would allow them to work while unvaccinated, not affecting people's right to choose. However, we accept that in cases where a person has studied for years to become a doctor, this is a very weak counterargument.

As for the anti-democratic argument, there are numerous other coercive measures that democratic states use, e.g. compulsory education, seatbelt, and helmet laws, the aforementioned drunk driving laws, etc. Therefore, the claim that vaccination mandates are anti-democratic appears weak.

The interplay between the deontological arguments in this Section and Section 5.2.3 are subject to ethical debates. We conclude that this argument is strong (with some weak elements – anti-democratic argument), with a strong counterargument as well.

#### 5.3.4 Vaccine Ineffectiveness

It was argued by politicians that introducing vaccination mandates that allow vaccinated people to live life close to the way they did before the pandemic will instil a false sense of security in the sense that they will believe they are safe and cannot become infected when that is not necessarily the case. A study published in March 2022 found significant decreases in the vaccine's protection against infection with COVID-19 after 6 months of vaccination, with a slight decline in protection against severe disease, also after 6 months (Feikin et al., 2022). A more recent study from October 2022 has found that COVID-19 vaccine effectiveness against critical disease begins to wane after approximately four months (both after receiving two and three doses), indicating that the effectiveness is rather short-lived (Ferdinands et al., 2022). Additionally, it was found that the vaccines were less effective against the newer COVID-19 variant – Omicron (Andrews et al., 2022). An even more recent study from March 2023 presents similar findings – vaccine effectiveness (after receiving two doses and a booster) against critical cases of infection significantly declines four months after receiving the booster vaccine (ECDC, 2023). Additionally, even COVID-19 vaccination is unable to protect every single individual from COVID-19-related death, with data from

February 2022 showing that 40% of COVID-19-related adult deaths in the U.S. were of vaccinated persons (Amin et al., 2022).

This was a particularly popular argument among the politicians who were opposed to mandatory vaccination. Our reformulation of their arguments is as follows:

**Premise 1:** Vaccination mandates will give vaccinated people a sense of security, as they will interact more with other people.

**Premise 2:** COVID-19 vaccination does not guarantee that the person cannot get infected or experience severe symptoms or fatal consequences (Amin et al., 2022; ECDC, 2023; Feikin et al., 2022; Ferdinands et al., 2022).

**Premise 3:** COVID-19 vaccines are less effective against the newer COVID-19 variants (Andrews et al., 2022).

**Premise 4:** Countries with high COVID-19 vaccination uptake still experience breakthrough COVID-19 infections (Feng et al., 2022).

**Conclusion 1:** COVID-19 vaccines are not effective at protecting people from infection and severe symptoms.

**Conclusion 2:** Vaccination mandates will give vaccinated people an unjustified sense of security.

**Conclusion 3:** Vaccination mandates do not serve the purpose of protecting people.

We accept Premises 2, 3, and 4 as being true due to the findings mentioned above. However, we cannot accept Conclusion 1. Despite COVID-19 vaccination showing decreased effectiveness against newer variants, this is in comparison with their effectiveness against the original variant, so it is not accurate to believe that they are completely ineffective against new mutations. Studies show that the vaccines do protect against infection with the newer variants when taken in comparison with unvaccinated individuals (Link-Gelles et al., 2023). Conclusion 2 seems sound to us since it is reasonable to assume that people will feel unfoundedly safe when restarting their social lives upon the introduction of vaccination mandates, while the risk of becoming infected still exists. Conclusion 3, however, we cannot accept. As we mentioned in Section 5.1.1, what actually poses a higher infection risk to vaccinated people is not increased interaction with other vaccinated people, but rather interaction with unvaccinated persons, who themselves have a higher risk of becoming infected and also transmitting the disease (Fisman et al., 2022). Therefore, mandatory vaccination would indeed serve the purpose of protecting people. We find this argument to be unsound due to Conclusion 1 not arriving logically from the premises. The premises are focused on the flaws of current vaccines, but those flaws do not render the vaccines ineffective (as was stated in Section 5.1.1).

#### 5.3.5 Anti-Paternalistic View

Vaccines are generally effective at protecting people from infection, severe symptoms, hospitalisations, and death (Section 5.1.1). Given this knowledge, it is fair to assume that unvaccinated people expose mostly themselves and other unvaccinated people (who have made the same decision not to vaccinate) to harm by not vaccinating. If the effects on the vaccinated people were substantial, this would call into question the vaccine effectiveness, and, by extension, the rationale for vaccination mandates. Therefore, this is (at least to some extent) a paternalistic policy and the question of whether a paternalistic policy is justified in this scenario arises.

Some people hold autonomy, in and by itself, as a value (Dworkin, 1976). Take Mill's (1859) statement as an example of this thought – "If a person possesses any tolerable amount of common sense and experience, his own mode of laying out his existence is the best, not because it is the best in itself, but because it is his own mode." (p. 121) Since autonomy can be held as a separate value, this potentially compromises the consequentialist argument for paternalistic vaccination mandates, especially when paired with questions about potential vaccine ineffectiveness (Section 5.3.4), limited benefit to healthy people (Section 5.3.10), and the damage done to the social capital (Section 5.3.2). From the deontological perspective, Kantian views also oppose paternalism, as it treats them as "means to their own good, rather than as ends in itself" (Dworkin, 2020). Skelton & Forsberg (2020) point out that although paternalistic policies could be acceptable for children, as they lack the capacity to make decisions, for independent adults such policies are generally unacceptable. Considering this, one could argue that a paternalistic policy is not justified in this scenario, and so we arrive at our reformulated argument:

Premise 1: COVID-19 vaccines protect vaccinated people from infection and severe symptoms (Chen et al., 2022; Christie et al., 2021; Scobie et al., 2021; and others).Premise 2: Unvaccinated people only expose themselves and other unvaccinated people (who have also made the decision not to vaccinate) to harm.

**Premise 3:** People should have the right to make choices, even if those choices put them in harm's way (anti-paternalistic view).

**Conclusion 1:** Since vaccinated people are protected, unvaccinated people do not pose material harm to them.

Conclusion 2: Unvaccinated people should have the right to not vaccinate.Conclusion 3: Vaccination mandates should not be imposed.

The main issue with this argument comes from the assumption that is outlined in Premise 2, which would essentially make this policy into a purely paternalistic one. However, this assumption is weak, as unvaccinated people do cause harm to vaccinated people as well. Firstly, a study has shown that vaccinated people have a higher chance of getting infected if the population has an increased mixing of unvaccinated and vaccinated people. This effect is said to be "disproportionate to the fraction of unvaccinated people in the population" (Fisman, Amoako, & Tuite, 2022, p. 579). Another indirect effect comes from the fact that unvaccinated people are more exposed to more serious symptoms from COVID-19 (Section 5.1.1), and so they would take up more resources in hospitals due to hospitalisations (Apollo.lv, 2021), and if they provide essential services to society, these services (and society) are exposed to harm (Section 5.2.6). Moreover, some people are unable to vaccinate due to medical contraindications (SPKC et al., 2022), and must remain unvaccinated, but not by choice. Unvaccinated people who do not vaccinate by choice impose harm on these people as well.

Although this argument provides logic for why vaccination mandates should not be accepted on paternalistic grounds, the policy itself is not just paternalistic – there is harm imposed on vaccinated people as well.

# 5.3.6 Less Restrictive Options SSE RIGA

Politicians whose stance was against mandatory vaccination would often argue that less restrictive measures can be used for decreasing the spread of COVID-19 and overcoming the pandemic. Namely, they believed that testing more people more often would increase the transparency of the disease spread which would allow to keep things under control and then put the infected persons into self-isolation until they recover to prevent further spread or quarantine those who may have been exposed (CDC, n.d.-b; Philippe et al., 2023). Additionally, they thought that epidemiological safety measures, such as wearing masks in

public and keeping a two-meter distance in public places would help prevent the spread in case someone was infected but didn't know it yet (Mayo Clinic Staff, n.d.-b; Talic et al., 2021). These measures are considerably less restrictive in nature than mandatory vaccination. As we discussed beforehand, the PLRA should be applied by policymakers when thinking about vaccination, and if there exists an opportunity to use less intrusive measures for incentivizing vaccination (such as nudging or incentives) and still the pandemic can be maintained with the help of less intrusive measures like quarantine, isolation, masks, etc., then that should be the first option (Giubilini, 2019). Additionally, Verweij and Dawson (2004) argue that one significant pillar of collective vaccination programmes should be that the burden-to-benefits ratio for vaccination should be more favorable than the same ratio for other options. It may well be reasonable to imagine that in the case of the COVID-19 pandemic, the burdens of vaccination could outweigh the benefits, as we discuss in other sections, since the vaccines are new, and their 'benefits' are not entirely known at the beginning at least, and as it does turn out, their effectiveness is waning and does not provide long-lasting immunity (ECDC, 2023; Feikin et al., 2022; Ferdinands et al., 2022). Whereas other options described here are less intrusive and effective to some extent. Overall, it seems that there are measures other than mandatory vaccination that can be taken to prevent the spread of COVID-19, which are also less restrictive. Less restrictive options may not lead to such negative effects on political trust as described in Section 5.3.2. Dubov & Phung (2015) point out that potentially a better solution is using nudges rather than more coercive measures in achieving higher vaccination uptake. With this, we arrive at our reformulated argument:

**Premise 1:** Increased testing can be used as an effective safety measure (Philippe et al., 2023).

**Premise 2:** Isolating infected people can be used as an effective safety measure (CDC, n.d.-b).

**Premise 3:** Epidemiological safety measures (wearing masks, keeping distance, ventilating rooms, disinfection, screening) can be used as effective safety measures (Mayo Clinic Staff, n.d.-b; Talic et al., 2021).

**Premise 4:** The aforementioned tools aren't as strict in nature and quality-of-life diminishing as vaccination mandates.

**Premise 5:** In general, policymakers should apply the Principle of Least Restrictive Alternative (Giubilini, 2019).

**Conclusion 1:** There are other less strict, but effective alternatives to mandatory vaccination.

**Conclusion 2:** Such measures need to be implemented instead of vaccination mandates.

However, the Latvian government had already implemented less restrictive measures before starting to consider mandatory vaccination, thus they had applied the PLRA and were not violating it. For context, face masks, distancing, isolation, and quarantine were all introduced in 2020 (Veselības ministrija, 2022; SPKC, 2023). Unfortunately, all of the measures were not sufficient for controlling the spread of COVID-19, so, if we imagine that the government was following something similar to Giubilini's (2019) intervention ladder, the obvious next step was to introduce mandatory vaccination. Applying the PLRA in the context of herd immunity means pursuing the least restrictive policy with the constraint that it has to be effective in achieving herd immunity (Giubilini, 2021). Therefore, while Premises 1, 2, 3, 4, and 5 are true, and so is Conclusion 1, we cannot accept Conclusion 2 as being strong since the less restrictive measures had already been implemented. Thus, we conclude that this argument is weak.

## 5.3.7 Vaccine Safety

At the time, there wasn't as much evidence on the possible long-term effects of COVID-19 vaccination, yet this argument was still used. Now it is known that the J&J/Janssen vaccine can sometimes cause thrombosis with thrombocytopenia syndrome (TTS), as well as Guillain-Barré Syndrome; mRNA COVID-19 vaccines, however, can sometimes cause myocarditis and pericarditis (the specific nature of these health issues is described in more detail in Section 5.2.7).

However, at the time a different twist was used to persuade others about the lack of vaccine safety – the fact that they had been developed quickly and hadn't finished all phases of testing. Typically, a vaccine requires 10-15 years of laboratory research, followed by testing in animals, and only then it proceeds to clinical trials, which have four phases. The first phase is testing the safety of the vaccine in people, i.e. side effects and whether it produces immunity, within a small group of up to 100 people, the second tests on more people, up to 300, and they are typically with a certain set of characteristics (such as age and physical health), the third phase includes up to 3,000 people to confirm the efficacy of the

vaccine and determine any additional side effects, as well as support the basis that the vaccine is safe. After phase 3, the vaccine typically receives approval (from FDA in the U.S. and EMA in Europe), which then allows phase 4 to begin, where the vaccine's efficacy and safety are monitored over a prolonged period on larger populations (CDC, n.d.-a). Thus, multiple people (including politicians) felt stressed due to seeing a COVID-19 vaccine rolled out to the market only a year after the first case of infection was discovered while holding the knowledge of the typical time it takes to create a vaccine. This argument was then extended to a lack of knowledge by the researchers creating this vaccine in terms of what are the longer-term side effects. Additionally, they argued that despite all vaccines (besides the COVID-19 vaccine) available to the public being safe for use, they can still sometimes result in unforeseen and serious side effects (The College of Physicians of Philadelphia, n.d.).

In Europe, COVID-19 vaccines received what is known as conditional marketing authorisation. Essentially, this means that the European Medicines Agency [EMA] approved the vaccines for usage while they were able to provide less data than is typically required (as well as non-clinical data, which typically wouldn't be accepted). This authorisation was valid for one year for all COVID-19 vaccines (EMA, n.d.-a; EMA, n.d.-b). On this account, arguments from politicians at times included mention of the Nuremberg Code, especially the first point, which is about a human subject in a medical experiment absolutely having to voluntarily consent to what is being done (Nuremberg Code, n.d.). The COVID-19 vaccines with conditional marketing authorisation were thus compared to medical experiments, and it was said that mandatory vaccination violates the Nuremberg code's first point. With this, we arrive at our reformulated argument:

**Premise 1:** There seems to still be a lot of uncertainty about potential COVID-19 vaccine side effects, especially in the long term.

**Premise 2:** However rare it may be, vaccination can result in potentially severe side effects (CDC, 2023b).

**Premise 3:** Medical manipulations with increased risk or uncertainty should be treated with more caution.

**Conclusion 1:** Forcing people to risk their health through mandatory vaccination is immoral and unethical.

Conclusion 2: Vaccination mandates are immoral and unethical.

The common belief behind this argument was that COVID-19 vaccines took a year to develop when that in fact was not the case. A detailed description of the COVID-19 vaccine development process can be found in section 2.1.3, but in short, the quick speed of development can be attributed to information sharing and cooperation between researchers and manufacturers, as well as due to the fact that COVID-19 belongs to a group of viruses which have been continuously researched for more than 50 years now, meaning that scientists had a decent set of information on these types of viruses (Solis-Moreira, 2021). Moreover, most would agree that COVID-19 vaccines and mandatory policies for them do not violate the Nuremberg code, as the vaccines are not experimental and have been since authorised in the EU, more specifically, the Moderna vaccine was authorised on October 3, 2022; Pfizer on October 10, 2022; Janssen on January 10, 2023, thus refuting this belief (EMA, n.d.-c; Najera, 2021; Sween, Ekeoduru, Mann, 2022).

Additionally, though the COVID-19 vaccines may indeed cause some longer-term health issues, the chance of that happening is extremely rare. More specifically, the incidence of TTS is 4 cases per one million administered doses, for Guillain-Barré Syndrome it is less than 2 cases per one million doses, and for myocarditis and pericarditis, it is approximately 5 cases per one million doses (though the rate is higher for young men, among the age group 12-24 the incidence was on average 71 cases per one million doses administered, but this is still an extremely low rate). (Abara et al., 2023; CDC, 2023b; Oster et al., 2022)

Premise 2 cannot be refuted, since COVID-19 vaccines just like almost any other vaccine carry with them some risk of both short- and long-term side effects. Similarly, Premise 3 cannot be refuted, as it is true that higher-risk and higher-uncertainty medical procedures need to be treated with caution, especially by medical practitioners.

Other people's right not to be harmed (Section 5.2.1) compromises this argument as well. Overall, we find this argument to be strong, but with strong counterarguments as well.

DE KIGA

#### 5.3.8 Government Failure

The Cabinet of Ministers made it known in their regulations in December 2020 that mandatory vaccination will not take place, and unvaccinated people will not undergo any exclusive restrictions (Ministru Kabinets, 2021c). This is not in line with the topic of this paper – vaccination mandates that place additional restrictions on unvaccinated people.

By introducing vaccination mandates, Cabinet of Ministers' politicians would be breaking their promise. Breaking promises by politicians is one vector that will lead to political distrust (Bertsou, 2019). As we discussed in Section 2.1.2, the level of trust in the government is incredibly low in Latvia, even pre-pandemic (OECD, n.d.), and low political trust is associated with lower vaccine acceptance (Lazarus et al., 2020; Lee et al., 2016; Trent et al., 2022). The study by Bertsou (2019) also finds that low levels of trust in the government are related to lower compliance with the law, which is problematic in a pandemic situation.

Additionally, the Minister of Health at the time, Daniels Pavluts, created a Vaccination Office in January of 2021, which was a separate structure under the Ministry of Health intended for the management and planning of vaccination against COVID-19 in Latvia (Sondare, 2021). The office had little success in achieving its goals (which were increasing vaccination uptake and reaching herd immunity) but rather ran into a plethora of issues - the ability to sign up for vaccination via a phone call was nearly impossible, questionable spending fell into the eyes of the public, as well as problems with the vaccine supply chain and incapability to get vaccines delivered on time (Anstrate, 2021b; Feldmanis, 2021a; Zvirbulis, 2021). Much of Latvian society perceived the office to be a cover-up for the government's past and possible future mistakes, as it would give the people an easy target to identify and attack when something (such as mandatory vaccination) happening in the country seems inappropriate to them. Though the specifics of the Vaccination Office's operational choices are not known, many were angered by the large sums of money spent on different things that were meant to increase vaccination uptake but subsequently failed to achieve that, as well as the sky-high salaries intended for the people working in the structure. One example of such spending was for the creation of an informative newspaper, called the "Vaccination Newspaper" meant to increase society's knowledgeability about COVID-19 vaccination so that more people would get vaccinated, but many saw the sums spent for this purpose as inexplicably high, accusing the Minister of Health of misspending government funds (Feldmanis, 2021a). The negative opinions of Latvian society resulted in a restructuring of the office and becoming instead a structure under the National Health Service (Feldmanis, 2021b).

One can imagine that the actions taken by the Vaccination Office damaged society's trust in the government even further, which may have potentially been one of the factors influencing low vaccination uptake in Latvia. Since the government was responsible for decreasing public trust, and, in turn, reducing their willingness to vaccinate, they should not be placing the burden to fix the consequences of their mistakes onto the people by forcing them to vaccinate so that they can be able to continue working and making a living, or

otherwise punishing them by not allowing them to work if they don't vaccinate. With this we've come to our reformulated argument:

**Premise 1:** The Cabinet politicians made promises that COVID-19 vaccination will not be mandatory (Ministru Kabinets, 2021c).

**Premise 2:** Politicians should not break their promises, as breaking promises leads to political distrust (Bertsou, E., 2019).

**Premise 3:** The Cabinet failed its original COVID-19 vaccination campaign (Feldmanis, 2021b).

Premise 4: Politicians should take responsibility for their failures.

**Conclusion 1:** The Cabinet politicians responsible for COVID-19 vaccination mandates are not consistent with their promises.

**Conclusion 2:** The Cabinet politicians are not taking responsibility for their failed COVID-19 vaccination campaign.

Conclusion 3: COVID-19 vaccination mandates should not be implemented.

Premise 1 is a factual premise, and it is true as we described above. As for Premise 2, it can be argued from a consequentialist perspective that since trust in the government was already low beforehand, breaking the promise of not introducing mandatory vaccination for COVID-19 could be acceptable since the benefits of such a policy (increased vaccination uptake, see Section 5.1.2) would outweigh the further damage done to political trust (similar counterargument in for Section 5.3.2). Moreover, it can also easily be argued that politicians are taking on responsibility for their failures by implementing policies that, to their mind, would benefit society the most – in this case, vaccination mandates (at least from their perspective). Not doing anything could instead be interpreted as an admission of failure and giving up (rejection of responsibility to fix the situation). In this case, since the failures of the Vaccination Office are already sunk costs, it would not make sense to integrate these failures into further analysis. We conclude that this argument is weak (with potentially unfounded Conclusion 2).

#### 5.3.9 Slippery Slope

Wilkenfield & Johnson (2022) present an idea that COVID-19 vaccination mandates might give grounds for a slippery slope argument. The authors state that passing such a mandate –

even if it is justified – could lead to further policies that would not be justified. While the authors point out that the right response is "a mechanism in place that pulls the brakes right at the juncture between the justified and the unjustified. The way to prevent unjustified behaviour is not to ban justified behaviour, but rather to be vigilant regarding when one might cross the relevant boundary" (Wilkenfield & Johnson, 2022, p. 32), it is not clear whether such a mechanism is in place. Some Western countries like Germany and Austria had implemented mandatory vaccination for all (Mumcuoglu et al., 2021), so it is not so clear, whether Latvian politicians would stop at this measure. The same logic can be derived from the opposition used for the argument in Section 5.3.6 – in short, less restrictive measures were tried and were subsequently traded for more restrictive measures. Logically, if this policy does not generate the needed results, the next step would be to implement an even more stringent vaccination mandate.

The main issue with this is that the arguments used for why vaccination mandates for work are unjustifiable (see all sections in 5.3) would be even stronger for a more restrictive policy like mandatory vaccination for all. This new policy would have to pass an even more demanding burden of proof to be morally right. As the topic of discussion in this thesis is vaccination mandates for work, we cannot say whether such a burden of proof would be passed – but what we can assume is that, overall, the arguments in Section 5.2 would be weaker than it is now, and, overall, the arguments in Section 5.3 – stronger than it is now. Since the arguments in Section 5.3 argue that vaccination mandates for work are unethical, for this argument we can also argue that, logically, vaccination mandates for work are also unethical. Thus, we arrive at our reformulated argument:

Premise 1: Vaccination mandates in a few sectors present the risk of initiating a slippery slope that will lead to vaccination mandates in all employment sectors.Premise 2: In turn, vaccination mandates in all employment sectors will ultimately lead to mandatory vaccination for all parts of society (i.e. outside of the employed population).

**Premise 3:** People have the right to choose and decline the medical manipulations they receive.

Conclusion 1: Vaccination mandates will lead to mandatory vaccination for all.Conclusion 2: Mandatory vaccination for all is immoral and unethical.Conclusion 3: Vaccination mandates should not be passed.

The slippery slope argument is a common fallacy that may be true at times but needs to be proven (see Glossary for definition). The proof entails showing why the movement from vaccination mandates for work as they were implemented in Latvia would necessarily lead to vaccination mandates for all employed people, why the movement to such a policy necessitates a movement to mandatory vaccination for all, and, ultimately, why mandatory vaccination for all is unethical. Starting with the latter, the argument itself points out the changes in the perceived strengths of each side of the argument – however, it is also still unclear whether this can be accepted. This is because, logically, the ethicality of vaccination mandates is very much affected by the level of danger imposed by the disease itself. So, one could argue that just because vaccination mandates have a higher burden of proof or necessitate a higher level of danger, that does not make them unethical. Since our paper revolves around the evaluation of the strengths/weaknesses of separate arguments used by Latvian politicians, and not to answer the question posed here – are vaccine mandates ethical – we cannot evaluate the truthfulness of Conclusion 2.

Regarding the slippery slope itself, although the argument mentions examples from some Western countries taking these extra measures, they were not imposed by the Latvian government, implying that the slippery slope argument did not fulfil its promise (Mumcuoglu et al., 2021).

We conclude that this argument is weak due to a lack of evidence that vaccination mandates for all are unethical, as well as a lack of concrete evidence for the slippery slope itself.

#### 5.3.10 Unfair Targeting of Protected Population

One argument also often used by politicians was that it is unreasonable to ask those, who are not at risk from COVID-19, to vaccinate. The rationale behind this firstly comes from the belief that COVID-19 is not dangerous to people who are young and maintain a healthy and active lifestyle. COVID-19 most often targets people who are aged 65 and up, namely, more than 80% of deaths related to COVID-19 happen in this age group. Additionally, multiple medical conditions can increase a person's chances of becoming severely ill after becoming infected, these infections include (but are not limited to) cancer, chronic illnesses, diabetes, neurological conditions, disabilities, heart conditions, weakened immune system, or immunodeficiency. What is more, being overweight or obese and living a rather sedentary life also increases your risk for severe disease (CDC, 2023a).

It then logically follows, that since the goal of a mandatory vaccination policy is to achieve herd immunity, this will work to protect the aforementioned vulnerable groups. Comparatively, the vulnerable groups will experience (not perhaps directly, but more in a utility sense) a larger reduction in the risks than young and active people will. While it is true that both groups receive some benefit from herd immunity, this benefit is not equal for all groups, yet the burden (getting vaccinated) to both groups is the same. Verweij & Dawson (2004) argue that in collective immunization programmes, the distribution of benefits and burdens should be as equal as possible since the outcome is a public good, but not everyone receives the same benefits and, in a sense, not everyone experiences the same burden, even if the same activity is being performed. With that in mind, we come to our reformulated argument:

**Premise 1:** COVID-19 is not as dangerous to young, healthy, and active people as it is to older and less healthy people (CDC, 2023a).

**Premise 2:** The benefits of a mandatory vaccination programme are mostly captured by the elderly, immunocompromised, and chronically ill.

**Premise 3:** There should be a just distribution of costs and benefits in a mandatory vaccination policy (Verweij & Dawson, 2004).

**Conclusion 1:** Vaccination mandates against COVID-19 target people that aren't affected by COVID-19.

**Conclusion 2:** Vaccination mandates are useless for these people.

Conclusion 3: Vaccination mandates should not be implemented.

As discussed in Section 5.2.1, unvaccinated persons pose harm both to other individuals, as well as to the healthcare system. Also in Section 5.2.1, we've outlined the right of each individual to not be harmed by others. This clashes with Conclusion 3, not exactly disproving it, but providing an argument against it.

Additionally, Premise 1 does hold true, yet it is also true that young people nonetheless face risks upon infection with COVID-19, even if it is less common than for older individuals. One study on young adults (aged 18 to 34) who had been hospitalized found worrying results – 21% needed intensive care, 10% had to be put on ventilators, and nearly 3% died (Cunningham et al., 2020). Recent data from the Office for National Statistics (2023) shows that in the U.K. from the age group 17-24, about 1.5% reported experiencing long COVID symptoms, in the age group 25-34 this increased to 2.6%, and of people aged 35-49 about 4.2% were affected (as percentages of people living in private households). This shows that there can be unfavourable outcomes for young people as well, and infection with COVID-19 should not be taken lightly. Additionally, this disproves Conclusion 1.

While Conclusions 1 and 2 are flawed, there is some backing for Premise 3. This is a question of whether it is right for the government to impose a policy with imperfect distributions of costs and benefits. Although this may seem like an extreme case, such policies aren't rare. For example, any progressive tax rate policy (which are widely implemented across the world (Hinders, 2023)) would have to be considered unethical on these grounds as well. Section 5.2.1 and 5.2.6 provides strong reasoning for why it may be acceptable – due to the fact that by not vaccinating, the costs are passed onto other people and society at large. Thus, we conclude that this argument is weak.

# 5.3.11 Unintended Consequences

As mentioned in previous sections, Latvian citizens were able to acquire a Digital Green Certificate for use in various public activities either through being fully vaccinated against COVID-19 (at one point, this meant two doses, later it was three) or infected and recovered from COVID-19 (similarly, the rules for infected and recovered persons changed over time, with some form of vaccination on top of natural immunity being required). (Nacionalais Veselības Dienests, 2021) Though there is a lack of concrete evidence on this, there was certainly some portion of the unvaccinated Latvian population that viewed infection with COVID-19 as less dangerous than vaccination and were fully willing to purposefully become infected in hopes of receiving the Certificate and not losing their jobs. There were some news articles documenting this, such as a 53-year-old teacher from Bauska working in multiple educational institutions in Riga had written a public advertisement in Bauska's local newspaper stating that she is looking for a person who could infect her with COVID-19. Upon posting this ad, she received multiple calls from people willing to assist her in her quest, with some asking for 100 euros for such a deal, yet she tells the news media that she would be willing to pay much more than that if needed. What is more, she also stated in an interview that she is not alone and is in fact in a group of other teachers also looking for the same kind of 'service' and that all have agreed upon sharing their contacts if they find some (TVNET/LETA, 2021; Zvirbulis, 2021). There have been other anecdotal examples, such as a Latvian University student, Raimonds Upmalis, posting on Facebook stating that he is looking for infected people in order to continue his studies. Comments were mostly trying to

70

discourage him from such a plan, thus he answered also in the comments with a statement about multiple people messaging him that they are also looking for the same thing (Raimonds Upmalis, 2021). Though none of this is concrete evidence, we can be almost certain that there were people looking to get purposefully infected in order to comply with the government's regulations.

A study that analysed research for a 20-year period, found that people generally did not accept the idea of mandatory vaccination for work and that it may cause purposeful infection (Drury et al., 2021). Though there is no mention of this in any type of Latvian media and no direct proof of it occurring, it may be reasonable to believe that another form of purposeful infection with COVID-19 took place in Latvia, which is known as a "COVID party". A "COVID party" is a gathering where people meet up with the purpose of getting infected with COVID-19 to gain immunity. There is some speculation that this idea may have come from chicken pox parties, where parents would throw a party for their children when one got infected with chicken pox as a way of getting them infected early for them to have lifetime immunity at a time when a vaccine for it was not yet widely available (Healy, 2022; Johnson, 2021). In Italy, where there were regulations similar to those in Latvia (a Certificate requirement for the ability to work amongst other activities), it was widely known that "COVID parties" took place as a means of receiving the certificate of recovery. For multiple people, this ended with hospitalisation and the need for respirators, and one 55-year-old man, who was an attendee at one of the parties, unfortunately, died after becoming infected (van Brugen, 2021; Nadeau, 2021). This showcases the unintended consequences that can accompany a mandatory vaccination policy.

Besides the risks associated with infection itself (in Latvia, a hospitalisation rate of 230 per 100,000 inhabitants and a death rate of 28,1 per 100,000 inhabitants as per the most recent data by SPKC (2022)), there recently have been more people reporting long COVID (see Glossary for definition). A systematic review of multiple studies conducted mostly in Europe and Asia found that around 45% of people who recovered from COVID-19 infection experienced some long COVID symptoms after 4 months after recovery (O'Mahoney et al., 2023). The most common symptoms are fatigue, fever, various respiratory and heart symptoms, neurological symptoms (such as brain fog), as well as digestive symptoms, among others (CDC, 2022b; Miller, 2023).

These implications paint the picture of why it may be undesirable to implement mandatory vaccination, as some people may turn to purposeful infection which has unpredictable outcomes. With this, we arrive at our reformulated argument:

71

**Premise 1:** Certificates can be acquired through vaccination or infection with COVID-19 (Nacionālais Veselības Dienests, 2021).

**Premise 2:** Some people view COVID-19 as less dangerous than vaccination (Nadeau, 2021; Raimonds Upmalis, 2021; TVNET/LETA, 2021; van Brugen, 2021; Zvirbulis, 2021).

Premise 3: People will pick the option they perceive as less dangerous.
Premise 4: Although not perceived by some people this way, contracting COVID-19 is dangerous (CDC, 2022b; Miller, 2023; O'Mahoney et al., 2023; SPKC, 2022).
Conclusion 1: Some people will try to contract COVID-19 instead of vaccinating.
Conclusion 2: Vaccination mandates produce dangerous unintended side effects.
Conclusion 3: Vaccination mandates should not be implemented.

Despite the existence of such a risk that some people will purposefully put themselves in harm's way in case of a mandatory vaccination policy (i.e. we cannot refute Conclusions 1 and 2), there is no evidence that this portion of people is substantial. Namely, we assume that the majority of people are rational enough to reasonably weigh the benefits and disadvantages of both options (vaccination and infection) and come to the less harmful conclusion – to vaccinate. Apart from this, accepting Conclusions 1 and 2 and the remaining premises is not enough to support Conclusion 3. While vaccination mandates may incentivize some persons to purposefully infect themselves with COVID-19, the number of such people is likely to be small, and, oppositely, the number of people that will decide to get the certificate by vaccinating – large. On consequential grounds, we argue that the unintended negative side effects are considerably smaller than the benefits of such a mandate. Thus, we find this argument to be weak.

# SSE RIGA

# 5.3.12 Unemployment Effects

As discussed in Sections 5.1.2 and 5.3.11, there was some level of resistance against mandatory vaccination policies from the Latvian workforce. If we assume that mandatory vaccination truly would lead to higher unemployment rates, then this in turn would lead to a lower GDP in Latvia (Simpson, 2022). Additionally, it may be the case that some individuals would rather choose to work abroad in the case of mandatory vaccination, and it has been

found that higher emigration rates in Latvia may lead to higher unemployment rates (Hazans, 2013), which, as mentioned previously, again leads to lower GDP.

Furthermore, many people leaving their jobs has the potential of causing some industries to collapse due to a lack of workers. There were instances of companies feeling desperate as the due date of receiving a vaccination certificate was approaching, and they could not afford to lose their employees with highly specific skills due to a lack of interchangeability and a qualified workforce (Smagare, 2021). Additionally, it is especially risky when it comes to the healthcare industry, as healthcare workers tend to be viewed as a valuable resource in a pandemic, and Latvia has been experiencing a shortage of healthcare workers year-on-year (Feldmanis, 2021c). Because of a higher risk of infection due to the nature of the job, hospitals tend to already be short-staffed, and the potential risk of losing more workers due to vaccination mandates and some individuals' strong stance against vaccination is simply unimaginable (Sween et al., 2022).

With the risk of some industries shutting down operations (indefinitely or for some time, both instances would leave an impact on the state's economy) and the risk of employees willingly leaving their jobs or homeland, it would create a risk for the Latvian economy in the near- and possibly long-term future. Thus, one can imagine that any sort of renewed economic activity resulting from the introduction of vaccination mandates would be offset by these risks if they come into effect. This was also an argument often used by Latvian politicians in opposition to vaccine mandates, which we have reformulated:

**Premise 1:** A substantial amount of people would rather lose their job or move country than get vaccinated.

**Premise 2:** Higher rates of unemployment and emigration are detrimental to the economy (Hazans, 2013; Simpson, 2022).

**Premise 3:** Understaffed industries affected by vaccination mandates (e.g. healthcare, social care, education) may face potential collapse due to operational risks under vaccination mandates (Dēvica, 2021; Feldmanis, 2021c; Smagare, 2021; Sween et al., 2022; Zvērs, 2022).

**Premise 4:** The boost to the economy given by the vaccination mandates will be limited due to decreased working capacity for industries affected by vaccination mandates.

**Conclusion 1:** Vaccination mandates will be detrimental to the economy.

**Conclusion 2:** Vaccination mandates will cause potential collapse for understaffed industries.

Conclusion 3: Vaccination mandates should not be implemented.

As we discussed in Section 5.1.2, Latvia actually saw a decrease in the number of unemployed people throughout 2021, even when vaccination mandates were in place. According to Centrālās statistikas portāls (2022), the unemployment rate in Latvia decreased during 2021, when there were no mandates at the beginning of the year and the year ended with mandates. Given that the most concerning is the employment level in the healthcare sector, we must note that already in September of 2021, more than 92% of healthcare workers had at least gotten one vaccination dose, indicating that a collapse of the healthcare sector due to unemployment from vaccination mandates as highly implausible (Veselības ministrija, 2021c). Moreover, Sween et al. (2022) argue that any increases in unemployment should also be netted against the reduction in infections in the respective industry.

Although according to The World Bank (2022), the net migration for Latvia in 2021 was indeed negative, that has been the case since 1988. Moreover, we see that net migration in 2021 was closer to zero than in the past thirty years (a net migration increasing trend can be observed since 1998, and 2021 was not an exception). Thus, we can say that Premise 1 lacks evidence. As it is the underlying premise on which the whole argument is based, due to lacking evidence we conclude that this argument is unsound.

Moreover, an opposing consequentialist argument can be made on the grounds that vaccinated people will now have better access to the employment market, mitigating some effects of the proposed increased unemployment. Additionally, while Premise 2 is factual, and even if we assume Premise 1 to be true (even though the substantiality of the number of people leaving work is arguably significant, see Section 5.1.2), we can expect a positive effect on the economy, as the Latvian government would not have to spend as much on welfare packages for people in some industries that could not work at all (similar argument made in Section 5.2.3). Thus, we conclude that this argument is weak.

### 5.3.13 Employer Ethical Dilemma

Many firms underwent financially turbulent times during the COVID-19 pandemic. For example, the restaurant industry was hard hit, with many experiencing losses in sales, and one chain in Latvia even being forced to close 42% of its restaurants due to decreased

74

demand and as a means of saving costs (Zvirbulis, 2020). Firing people would save money, especially for firms that already could not operate to the same degree as in pre-pandemic times or had to decrease their working hours (Eiropas Parlaments, 2020).

Under vaccination mandates (as they were proposed and later accepted), employers had the right to determine the necessity for a vaccination certificate (Saeima, 2020). Whether to vaccinate or not was a controversial decision in Latvia during these debates – one-third of the Latvian population aged 15 years and older did not plan on getting vaccinated, with another third being unsure at the time when vaccination was about to become available for all of society (Ozola-Balode, 2021). While it is safe to assume that a large share of unvaccinated people would end up vaccinating at the request of the employer (see Section 5.1.2), it is also correct to assume that there would be people that would rather lose their job (see Section 5.3.11).

Given that the latter case is acceptable, the employers effectively had a tool at their disposal of firing people and cutting wage costs. Such a tool could be even more important for firms that could not even operate at the time. Presenting employers with a tool that would help their financials during turbulent times at the expense of the employees' wellbeing creates an ethical dilemma for the employer who now has to decide whether to care more about the bottom line or the livelihood of their employees.

The government shouldn't introduce policies that pass down ethical dilemmas onto people and incentivise financial performance over people's wellbeing. Thus, vaccination mandates should not be implemented.

**Premise 1:** Due to vaccination mandates employers can cut costs by firing unvaccinated people.

**Premise 2:** Vaccination mandates place employers in an ethical dilemma of whether to care about their workers or fire them to save costs.

**Conclusion 1:** Vaccination mandates create ethical dilemmas.

Conclusion 2: Vaccination mandates should not be implemented.

This argument rests on the notion that demanding a vaccination certificate from employees points towards valuing financial gain over people's wellbeing. However, this argument does not address the perspective that many vaccinated people share – that vaccines are effective at protecting people from infection and severe symptoms in case of infection. Considering this, employers that demand vaccination certificates may do so because they do care about their employees' wellbeing (and by extension, depending on the level of physical contact in the profession, the wellbeing of their suppliers and customers as well). This point makes Premise 2 unfounded.

Moreover, it is unclear why passing down ethical dilemmas ought to be avoided when creating policies. There are many ethical dilemmas that employers must face in reality, which aren't mitigated by policies, nor they should be. For example, there isn't a set amount that employers must spend on compensation and benefits (apart from minimum wage). Similarly, employers are free to reduce necessary working hours or increase work flexibility. These examples show that employers are faced with countless ethical dilemmas in balancing financial performance and employees' wellbeing – it isn't clear why the government should regulate these situations purely because they produce ethical dilemmas. We argue that Conclusion 2 does not logically follow Conclusion 1.

We find this argument to be unsound.

### 5.4 Results

A summary of the results was compiled in Tables 4 & 5. The arguments were graded either strong, weak, or unsound (as was discussed in Sections 5.1-5.3). The weak arguments (although at times with ethical backing) had deep underlying problems, while stronger arguments didn't reveal scathing issues. Note that even the strong arguments had strong counterarguments – in our analysis, we often found that the two conflicting ideas (e.g. the right not to be harmed and the right to individual autonomy; the existence of a legitimate reason to discriminate) have to coexist, as a balance between the two is found. To reiterate, the purpose of this thesis was not to determine this balance and conclude, whether such mandates had to be introduced, but rather to investigate, whether Latvian politicians did indeed use strong ethical justifications for their stance on vaccine mandates for work.

We find that there were unsound, weak, and strong arguments used on both sides. We note that all the implicit premises used in arguments for vaccination mandates were strong – they had to be strong in order for the underlying arguments to have a chance to be evaluated as strong as well (e.g. if vaccines truly were ineffective, then there would be no point in arguing about vaccine mandates).

We conclude that Latvian politicians used some strong arguments for vaccine mandates like the right not to be harmed, the need to vaccinate high-risk professions (extension of the right not to be harmed), and the need to vaccinate essential professions (also an extension of the right not to be harmed). These arguments can be used in further debates. Additionally, we can say that the herd immunity argument has strong ethical backing, and, under different vaccines/infections, could also be used as a strong argument.

We also conclude that Latvian politicians used some strong arguments against vaccine mandates like the right to autonomy, concerns about vaccine safety (more specific to COVID-19 vaccines), and the fact that there wasn't a legitimate reason to discriminate (extension of both aforementioned arguments). We would also like to point out that although the anti-paternalistic argument isn't strong in and by itself, as vaccination policies aren't necessarily paternalistic, it does provide strong argumentation against potential paternalistic justification (thus, we would also advise not to justify mandating vaccination for autonomous adults based on paternalism). Given better empirical evidence, the argument of net negative effects from the effect on social capital (and, thus, vaccination uptake) and unemployment could become viable to use in debates. Similarly, if other less restrictive options proved to be more efficient, the argument for these options would also potentially be strong.

There were also some very weak (unsound) arguments for both sides – some arguments had to be removed from the sample due to lacking justification. Politicians should also, by all means, avoid using immoral fallacies of association in their argumentation.

# 6. Conclusion

We elect politicians to represent us and our best interests. They make decisions in our stead – often we cannot determine whether they are correct without the benefit of hindsight. But, at the very least, we expect those decisions to be justified. Especially so, when dealing with issues of the utmost significance. This was the case with vaccination mandates in Latvia. During COVID-19, Latvians dealt with verdicts that cost people their job and their livelihood. Oppositely, the months without the strict mandates potentially made some pass away unnecessarily.

In our research, we investigated the justifications made by Latvian politicians from The Saeima and The Minister Cabinet for these decisions made, as well as the opposing side. We concluded that there were many different diverse arguments used in this continuous debate. All in all, we collected 26 argument groups (31, if the implicit premises evaluated separately are counted). Some of the arguments had to be excluded due to them being too weak to justify a further analysis, implying poor argumentation skills. There were unsound, weak, and strong arguments on both sides. The strongest debate found itself clashing the unvaccinated people's right to autonomy with the right not to be harmed by people that are, in fact, harmed by those who do not vaccinate. There are also some implications that can be drawn from the rare, but potentially severe health issues caused by the vaccines. Pitting these points against each other could show us where we stand on whether or not there was a legitimate reason to discriminate against unvaccinated people.

The findings of our thesis can be used by politicians to reevaluate the ethicality of their justifications, by people to see how morally convincing were the justifications for imposing (or not) the vaccine mandates, as well as provide some insights into what the strongest arguments for each side are.



# SSE RIGA

Argument	Evaluation	Grade
Legal	Removed from analysis	Removed from analysis
Vaccine Effectiveness*	Backed by empirical claims that are challenged, but not disproven.	Strong
Motivation*	Backed by empirical claims that are challenged, but not disproven.	Strong
Non-discriminatory/Fair*	Backed by ethical literature (other arguments in this section). Strong counterargument (Wrongful Discrimination).	Strong
Testing Ineffectiveness*	Backed by empirical claims that are challenged, but not disproven.	Strong
Legitimacy of Freedom- reducing Policies*	Backed by ethical literature. Strong counterargument (Free Choice/ Personal Autonomy).	Strong
The Right Not to Be Harmed	Backed by ethical literature. Strong counterargument (Free Choice/ Personal Autonomy).	Strong
Economic Recovery	Limited evidence for the necessity of lockdowns and the effectiveness of high vaccination uptake in containing COVID-19 outbreaks.	Unsound
Easing, Not Restricting	The idea of the policy working as an easing mechanism rather than a restriction is factually flawed.	Unsound
Achieving Herd Immunity Safely	Strong ethical backing, but limited empirical evidence for the viability of herd immunity for COVID-19.	Weak
High-risk Professions	Strong due to ties with The Right Not to Be Harmed. The paternalistic premise is weak.	Strong
Critical Professions	Strong due to ties with The Right Not to Be Harmed.	Strong
Vaccine Safety	Sound but relies on paternalistic justification, which is refuted by Anti-paternalistic view.	Weak
Necessity For Reciprocal Policies	Refuted on deontological grounds. SSE RIGA	Weak

Table 4. Summarised evaluation of arguments for vaccine mandates.

\*implicit premise used in arguments for vaccination mandates.

Argument	Evaluation	Grade
Legal	Removed from analysis due to limitations on authors' expertise.	Removed from analysis
Collectivization	Fallacy of association. Immoral argument.	Removed from analysis (unsound)
Totalitarianism	Fallacy of association. Immoral argument.	Removed from analysis (unsound)
Government Conspiracy	Cannot be evaluated.	Removed from analysis
Genocide	Fallacy of association. Immoral argument.	Removed from analysis (unsound)
Foreign Examples	Fallacy of appeal to common opinion.	Removed from analysis (unsound)
Wrongful Discrimination	Backed by ethical literature (other arguments in this section). Strong counterargument (Non-	Strong
	discriminatory/fair).	
Social Capital	Limited evidence on net negative effects from vaccine mandates.	Weak
Free Choice/ Personal Autonomy	Strong ethical backing. Strong counterargument (Right Not to Be Harmed).	Strong
Vaccine Ineffectiveness	Unsound due to lack of justification for the conclusion.	Unsound
Anti-paternalistic View	Weak overall, but provides a strong argument against paternalistic mandates.	Weak
Less Restrictive Options	Less restrictive options already implemented without success.	Weak
Vaccine Safety	Strong ethical backing. Strong counterargument (Right Not to Be Harmed).	Strong
Government Failure	Weak (potentially unfounded) conclusion.	Weak
Slippery Slope	Lacks justification for unethicality of a slope. Lacks justification for the existence of the	Weak
	slope.	
Unfair Targeting of Protected	Healthy people are affected by COVID-19. Thus, healthy people also benefit from	Weak
Population	vaccination. The Right Not to Be Harmed is a strong counterargument as well.	
Unintended Consequences	Little evidence for net negative effects from unintended consequences.	Weak
Unemployment Effects	Limited empirical evidence.	Weak
Employer Ethical Dilemma	Lacks justification for existence of ethical dilemma and its negative connotations.	Unsound

Table 5. Summarised evaluation of arguments against vaccine mandates.

### Glossary

Autonomy – "the control an individual has over his or her own evaluations and choices." (Hausman & Welch, 2010, p. 128)

**Call-to-emotion fallacy** – "Attempting to persuade others to accept a position by exploiting their emotions instead of presenting evidence for the position." (Damer, 2009, p. 111) **Casual oversimplification** – "Oversimplifying the causal antecedents of an event by specifying causal factors that are insufficient to account for the event in question or by overemphasizing the role of one or more of those factors." (Damer, 2009, p. 178) **COVID-19** – "a mild to severe respiratory illness that is caused by a coronavirus (severe acute respiratory syndrome coronavirus 2 of the genus Betacoronavirus), is transmitted chiefly by contact with infectious material (such as respiratory droplets) or with objects or surfaces contaminated by the causative virus, and is characterized especially by fever, cough, and shortness of breath and may progress to pneumonia and respiratory failure." ("COVID-19," n.d., para. 1)

**COVID-19 pandemic** – a worldwide pandemic of COVID-19 that was declared on March 11<sup>th</sup>, 2020, and is still active as of December 31, 2022 (WHO, n.d.).

**Fallacious argument** – "one that either is deductively invalid or is inductively very weak or contains an unjustified premise or that ignores relevant evidence that is available and that should be known by the arguer." (Dowden, n.d., para. 16)

Formal fallacy – an issue with the structure of the argument. (Gula, 2002, p. 50)

**Herd immunity** – it "works to control the spread of disease within a population when a specific amount of that population (threshold) becomes immune to the disease through vaccination or infection and recovery." (Desai & Majumder, 2020, para. 8)

**Long COVID** - "...signs, symptoms, and conditions that continue or develop after initial COVID-19 or SARS-CoV-2 infection. The signs, symptoms, and conditions are present four

weeks or more after the initial phase of infection; may be multisystemic; and may present with a relapsing– remitting pattern and progression or worsening over time, with the possibility of severe and life-threatening events even months or years after infection. Long COVID is not one condition. It represents many potentially overlapping entities, likely with different biological causes and different sets of risk factors and outcomes." (Department of Health and Human Services, 2022, p. 14)

**Pandemic** – "an outbreak of a disease that occurs over a wide geographic area (such as multiple countries or continents) and typically affects a significant proportion of the population." ("Pandemic," n.d., para. 3)

**Slippery slope fallacy** – "assuming, without appropriate evidence, that a particular action or event is just one, usually the first, in a series of steps that will lead inevitably to a specific, usually undesirable, consequence." (Damer, 2009, p. 185)

Straw man fallacy – "When you take something your opponent has said, exaggerate or distort it, and then attack what you have exaggerated or distorted." (Gula, 2002, p. 86)
Trust in government – "the confidence of citizens and businesses in the actions of government to do what is right and perceived as fair" (OECD, 2013, p. 40)

**Vaccine hesitancy** – "delay in acceptance or refusal of vaccination despite availability of vaccination services." (MacDonald, The SAGE Working Group on Vaccine Hesitancy, 2015, p. 4163)

Vaccine refusal – complete refusal of all types of vaccinations ("Vaccine refusal", n.d.).

## 7. References

- Abara, W. E., Gee, J., Marquez, P., Woo, J., Myers, T. R., DeSantis, A., Baumblatt, J. A. G., Woo, E. J., Thompson, D., Nair, N., Su, J. R., Shimabukuro, T. T., Shay, D. K. (2023). Reports of Guillain-Barré Syndrome After COVID-19 Vaccination in the United States. *JAMA Network Open*, 6(2), Article e2253845. Retrieved from https://jamanetwork.com/journals/jamanetworkopen/fullarticle/2800871
- Abercrombie, G., & Batista-Navarro, R. (2018, October). *Identifying Opinion-Topics and Polarity of Parliamentary Debate Motions*. Presented at the 9<sup>th</sup> Workshop on Computational Approaches to Subjectivity, Sentiment and Social Media Analysis, Brussels, Belgium.
- Abraham, J. A., & Abraham, S. P. (2021). Patient Freedom toChoose orRefuse Treatment: Is Autonomy Upheld? International Journal of Science and Research Methodology, 19(1), 124-134. Retrieved from <u>https://ijsrm.humanjournals.com/wp-content/uploads/2021/08/8.Jennifer-A.-</u> <u>Abraham-Samuel-P.-Abraham.pdf</u>
- Allen, A. (2022, July 5). How Pfizer Won the Pandemic, Reaping Outsize Profit and Influence. Kaiser Health News. Retrieved from <u>https://khn.org/news/article/pfizer-pandemic-vaccine-market-paxlovid-outsize-profit-influence/</u>
- Amin, K., Ortaliza, J., Cox, C., Michaud, J., Kates, J. (2022, April 21). COVID-19 mortality preventable by vaccines. *Peterson-KFF Health System Tracker*. Retrieved from <u>https://www.healthsystemtracker.org/brief/covid19-and-other-leading-causes-ofdeath-in-the-us/</u>
- Andrews, N., Stowe, J., Kirsebom, F., Toffa, S., Rickeard, T., Gallagher, E., Gower, C., Kall, M., Groves, N., O'Connell, A. M., Simons, D., Blomquist, P. B., Zaidi, A., Nash, S., Aziz, N. I. B. A., Thelwall, S., Dabrera, G., Myers, R., Amirthalingam, G., ... Bernal, J. L. (2022). Covid-19 Vaccine Effectiveness against the Omicron (B.1.1.529)
  Variant. *The New England Journal of Medicine, 386*, 1532-1546.
  doi: 10.1056/NEJMoa2119451

- Anstrate, V. (2021a, May 3). Pirmo dienu vakcinācijai var pieteikties visi iedzīvotāji. (It's the first day when all citizens can apply for vaccination). *Latvijas Sabiedriskie Mediji*. Retrieved from <a href="https://www.lsm.lv/raksts/zinas/latvija/pirmo-dienu-vakcinacijai-var-pieteikties-visi-iedzivotaji.a403064/">https://www.lsm.lv/raksts/zinas/latvija/pirmo-dienu-vakcinacijai-var-pieteikties-visi-iedzivotaji.a403064/</a>
- Anstrate, V. (2021b, February 16). Senioriem grūtības pa tālruni pieteikties Covid-19 vakcīnām. (Seniors are having difficulties applying for COVID-19 vaccines via phone). Latvijas Sabiedriskie Mediji. Retrieved from <u>https://www.lsm.lv/raksts/zinas/latvija/senioriem-grutibas-pa-talruni-pieteiktiescovid-19-vakcinam.a393161/</u>
- Apollo.lv. (2021, October 23). "Viņiem te vispār nebūtu jābūt!" Nevakcinētie pacienti strauji piepilda slimnīcas ("They shouldn't even be here!" Unvaccinated patients are quickly filling up hospitals). *Apollo.lv*. Retrieved from <a href="https://www.apollo.lv/7368765/viniem-te-vispar-nebutu-jabut-nevakcinetie-pacienti-strauji-piepilda-slimnicas">https://www.apollo.lv/7368765/viniem-te-vispar-nebutu-jabut-nevakcinetie-pacienti-strauji-piepilda-slimnicas</a>
- Arbel, R., & Pliskin, J. (2022). Vaccinations versus Lockdowns to Prevent COVID-19 Mortality. Vaccines, 10(8), Article 1347. doi: 10.3390/vaccines10081347
- Āboliņš, U. (2021, October 30). Vakcinācija pret Covid-19: uz tālākām lauku nomalēm izbraukuma potēšanā izmantos armijas transportu. (Vaccination against COVID-19: army transport will be used for vaccination on the far outskirts of the countryside). *TV3 Ziņas*. Retrieved from <a href="https://zinas.tv3.lv/latvija/sabiedriba/vakcinacija-pret-covid-19-uz-talakam-lauku-nomalem-izbraukuma-potesana-izmantos-armijas-transportu/">https://zinas.tv3.lv/latvija/sabiedriba/vakcinacija-pret-covid-19-uz-talakam-lauku-nomalem-izbraukuma-potesana-izmantos-armijas-transportu/</a>
- Balch, B. (2022, August 23). What to know about BA.5, Paxlovid, and new vaccines coming out this fall. Association of American Medical Colleges News. Retrieved from <u>https://www.aamc.org/news-insights/what-know-about-ba5-paxlovid-and-new-vaccines-coming-out-fall</u>

- Bambery, B., Selgelid, M., Maslen, H., Pollard, A. J., Savulescu, J. (2013). The case for mandatory flu vaccination of children. *American Journal of Bioethics*, 13(9), 38-40. doi: 10.1080/15265161.2013.813602
- Baranova, D. G., & Mežmale, S. (2021). Same-Sex Relationships: Why Do Many Latvian Politicians Resist Them? Unpublished Bachelor's Thesis, Stockholm School of Economics in Riga, Latvia. Retrieved from <u>https://www.sseriga.edu/student-research?page=0</u>
- Bardosh, K., de Figueiredo, A., Gur-Arie, R., Jamrozik, E., Doidge, J. C., Lemmens, T., Keshavjee, S., Graham, J., Baral, S. (2022). The Unintended Consequences of COVID-19 Vaccine Policy: Why Mandates, Passports, and Segregated Lockdowns May Cause more Harm than Good. *BMJ Global Health*, 7. doi: 10.2139/ssrn.4022798
- Bauer, S., Contreras, S., Dehning, J., Linden, M., Iftekhar, E., Mohr, S. B., Olivera-Nappa,
  A., Priesemann, V. (2021). Relaxing restrictions at the pace of vaccination increases
  freedom and guards against further COVID-19 waves. *PLoS Computational Biology*, *17*(9), Article e1009288. Retrieved from
  https://journals.plos.org/ploscompbiol/article?id=10.1371/journal.pcbi.1009288
- Bavli, I., Sutton, B., & Galea, S. (2020). Harms of public health interventions against covid-19 must not be ignored. *The BMJ*, 371, Article m4074. Retrieved from https://www.bmj.com/content/371/bmj.m4074
- Bengtsson, M. (2016). How to plan and perform a qualitative study using content analysis. *NursingPlus Open, 2*(1), 8-14.
- Bertsou, E. (2019). Political Distrust and its Discontents: Exploring the Meaning, Expression and Significance of Political Distrust. *Societies*, *9*(4). doi: 10.3390/soc9040072
- Bhavan, A., Mishra, R., Sinha, P. P., Sawhney, R., Shah, R. R. (2019). *Investigating Political Herd Mentality: A Community Sentiment Based Approach*. Presented at the 57<sup>th</sup>
   Annual Meeting of the Association for Computational Linguistics: Student Research Workshop, Florence, Italy.

- Bhavan, A., Sharma, M., Sawhney, R., Shah, R. R. (2020). Analysis of Parliamentary Debate Transcripts Using Community-Based Graphical Approaches (Student Abstract).
   Presented at the 34<sup>th</sup> AAAI Conference on Artificial Intelligence, New York, USA.
- Bowen, G. (2009). Document Analysis as a Qualitative Research Method. Qualitative Research Journal, 9, 27-40.
- Bradley, E., & Navin, M. (2021). Vaccine Refusal Is Not Free Riding. *Erasmus Journal for Philosophy and Economics*, 14(1), 167-181.
- Brink, D. (2022). Mill's Moral and Political Philosophy. In E. N. Zalta & U. Nodelman (Eds.), *The Stanford Encyclopedia of Philosophy* (Fall 2022 ed.). Retrieved from <u>https://plato.stanford.edu/entries/mill-moral-political/</u>
- Byron, N. (2020, September 1). *Three in four adults globally say they'd get a vaccine for COVID -19*. Retrieved from <u>https://www.ipsos.com/en/three-four-adults-globally-say-theyd-get-vaccine-covid-19</u>
- Cavanaugh, A. M., Spicer, K. B., Thoroughman, D., Glick, C., Winter, K. (2021). Reduced Risk of Reinfection with SARS-CoV-2 After COVID-19 Vaccination - Kentucky, May-June 2021. Centers for Disease Control and Prevention Morbidity and Mortality Weekly Report, 70(32), 1081-1083. doi: 10.15585/mmwr.mm7032e1
- Centers for Disease Control and Prevention. (2023a). *People with Certain Medical Conditions*. Retrieved March 28, 2023, from <u>https://www.cdc.gov/coronavirus/2019-ncov/need-extra-precautions/people-with-</u> <u>medical-conditions.html</u>
- Centers for Disease Control and Prevention. (2023b). Selected Adverse Events Reported after COVID-19 Vaccination. Retrieved March 30, 2023, from https://www.cdc.gov/coronavirus/2019-ncov/vaccines/safety/adverse-events.html
- Centers for Disease Control and Prevention. (2022a). *Selected Adverse Events Reported after COVID-19 Vaccination*. Retrieved November 27, 2022, from https://www.cdc.gov/coronavirus/2019-ncov/vaccines/safety/adverse-events.html

Centers for Disease Control and Prevention. (2022b). *Long COVID or Post-COVID Conditions*. Retrieved March 27, 2023, from https://www.cdc.gov/coronavirus/2019-ncov/long-term-effects/index.html

Centers for Disease Control and Prevention. (2022c). *Possible Side Effects After Getting a COVID-19 Vaccine*. Retrieved March 30, 2023, from https://www.cdc.gov/coronavirus/2019-ncov/vaccines/expect/after.html

- Centers for Disease Control and Prevention. (2022d). *Understanding Risk*. Retrieved March 30, 2023, from <a href="https://www.cdc.gov/coronavirus/2019-ncov/your-health/understanding-risk.html">https://www.cdc.gov/coronavirus/2019-ncov/your-health/understanding-risk.html</a>
- Centers for Disease Control and Prevention. (n.d.-a). How Vaccines are Developed and Approved for Use. Retrieved March 30, 2023, from <u>https://www.cdc.gov/vaccines/basics/test-approve.html</u>
- Centers for Disease Control and Prevention. (n.d.-b). Quarantine and Isolation. Retrieved March 30, 2023, from <u>https://www.cdc.gov/quarantine/index.html</u>
- Chen, X., Huang, H., Ju, J., Sun, R., Zhang, J. (2022). Impact of vaccination on the COVID-19 pandemic in U.S. states. *Scientific Reports*, 12, Article 1554. doi: 10.1038/s41598-022-05498-z
- Chichevalieva, S. (2011). *Developing a Framework for Public Health Law in Europe*. WHO/Europe. Retrieved from <u>https://www.euro.who.int/\_\_\_data/assets/pdf\_file/0004/151375/e95783.pdf</u>
- Childress, J.F., Faden, R. R., Gaare, R. D., Gostin, L. O., Kahn, J., Bonnie, R. J., Kass, N. E., Mastroianni, A. C., Moreno, J. D., Nieburg, P. (2002). Public health ethics: Mapping the terrain. *Journal of Law, Medicine & Ethics*, 30(2), 170–178.
- Christie, A., Henley, S. J., Mattocks, L., Fernando, R., Lansky, A., Ahmad, F. B., Adjemian, J., Anderson, R. N., Binder, A. M., Carey, K., Dee, D. L., Dias, T., Duck, W. M., Gaughan, D. M., Lyons, B. C., McNaghten, A. D., Park, M. M., Reses, H., Rodgers, L., ... Beach, M. J. (2021). Decreases in COVID-19 Cases, Emergency Department Visits, Hospital Admissions, and Deaths Among Older Adults Following the

Introduction of COVID-19 Vaccine — United States, September 6, 2020–May 1, 2021. *Centers for Disease Control and Prevention Morbidity and Mortality Weekly Report*, 70(23), 858-864. doi: 10.15585/mmwr.mm7023e2

The College of Physicians of Philadelphia. (n.d.) *Vaccine Side Effects and Adverse Events*. Retrieved March 30, 2023, from <u>https://historyofvaccines.org/getting-vaccinated/vaccine-faq/vaccine-side-effects-and-adverse-events</u>

- COVID-19 (n.d.). In *Merriam-Webster.com Dictionary*. Retrieved November 22, 2022, from <u>https://www.merriam-webster.com/dictionary/COVID-19</u>
- COVID-19 Vaccine Development: Behind the Scenes. (n.d.). Retrieved November 27, 2022, from <a href="https://covid19.nih.gov/news-and-stories/vaccine-development">https://covid19.nih.gov/news-and-stories/vaccine-development</a>
- Cunningham, J. W., Vaduganathan, M., Claggett, B. L., Jering, K. S., Bhatt, A. S., Rosenthal, N., Solomon, S. D. (2020). Clinical Outcomes in Young US Adults Hospitalized With COVID-19. JAMA Internal Medicine, 181(3), 379-381. doi: 10.1001/jamainternmed.2020.5313
- Curley, B. (2022, January 14). Omicron Symptoms: How They Compare with Other Coronavirus Variants. *Healthline*. Retrieved from <u>https://www.healthline.com/health-news/omicron-symptoms-how-they-compare-with-other-coronavirus-variants</u>
- Damer, T. E. (2009). Attacking Faulty Reasoning: A Practical Guide to Fallacy-Free Arguments, Sixth Edition. Belmont, CA: Wadsworth Cengage Learning.
- Department of Health and Human Services. (2022, August). *National Research Action Plan on Long COVID*. Retrieved March 27, 2023, from <u>https://www.covid.gov/assets/files/National-Research-Action-Plan-on-Long-COVID-08012022.pdf</u>
- Deruelle, F. (2022). The pharmaceutical industry is dangerous to health. Further proof with COVID-19. *Surgical Neurology International*, *13*(475), 1-18.

- Desai, A. N., & Majumder, M. S. (2020, October 19). What is Herd Immunity? The Journal of the American Medical Association, 324(20), 2113. Retrieved from <u>https://jamanetwork.com/journals/jama/fullarticle/2772168</u>
- Dēvica, P. (2021, September 8). Pedagogu trūkums Latvijas skolās ar katru gadu jūtams arvien vairak. (The lack of pedagogues in Latvian schools is felt more and more every year). *Latvijas Sabiedriskie Mediji*. Retrieved from <u>https://www.lsm.lv/raksts/zinas/latvija/pedagogu-trukums-latvijas-skolas-ar-katrugadu-jutams-arvien-vairak.a420383/</u>
- Dowden, B. (n.d.). Fallacies. In *Internet Encyclopedia of Philosophy* (para. 16). Retrieved February 23, 2023, from <u>https://iep.utm.edu/fallacy/</u>
- Drury, J., Mao, G., John, A., Kamal, A., Rubin, G. J., Stott, C., Vandrevala, T., Marteau, T.
   M. (2021). Behavioural responses to Covid-19 health certification: a rapid review.
   *BMC Public Health*, 21, Article 1205. Retrieved from <a href="https://bmcpublichealth.biomedcentral.com/articles/10.1186/s12889-021-11166-0">https://bmcpublichealth.biomedcentral.com/articles/10.1186/s12889-021-11166-0</a>
- Dubé, E., Vivion, M., MacDonald, N. E. (2015). Vaccine hesitancy, vaccine refusal and the anti-vaccine movement: influence, impact and implications. *Expert Reviews Vaccines*, 14(1), 99-117.
- Dubov, A., & Phung, C. (2015). Nudges or mandates? The ethics of mandatory flu vaccination. Vaccine, 33(22), 2530-2535. Retrieved from <u>https://pubmed.ncbi.nlm.nih.gov/25869886/</u>
- Dworkin, G. (1972). Paternalism. *The Monist*, 56(1), 64-84. Retrieved from https://www.jstor.org/stable/27902250
- Dworkin, G. (1976). Autonomy and Behavior Control. *The Hastings Center Report*, 6(1), 23-28. Retrieved from <a href="https://www.jstor.org/stable/pdf/3560358.pdf?refreqid=excelsior%3A04c801c4c600e">https://www.jstor.org/stable/pdf/3560358.pdf?refreqid=excelsior%3A04c801c4c600e</a> 270f4122cfb0fbac9ac&ab\_segments=&origin=&initiator=&acceptTC=1

Dworkin, G. (2020). Paternalism. In E. N. Zalta (Ed.), *The Stanford Encyclopedia of Philosophy* (Fall 2020 ed.). Retrieved from <u>https://plato.stanford.edu/entries/paternalism/</u>

Eiduks, K., & Ozola, E. (2022). "They will force everyone": main themes, arguments and complaints present in Latvian COVID-19 anti-vaccine movement on Facebook.
Unpublished student research paper, Stockholm School of Economics in Riga, Latvia.
Retrieved from

https://www.sseriga.edu/sites/default/files/2022-11/5Paper\_Eiduks\_Ozola.pdf

Eiropas Parlaments. (2020, April 22). Covid-19 ietekme uz ekonomiku: 100 miljardu eiro atbalsts darba vietu saglabāšanai. (Impact of COVID-19 on the economy: 100 billion euro support to save jobs). Retrieved from https://www.europarl.europa.eu/news/lv/headlines/society/20200416STO77205/covid -19-ietekme-uz-ekonomiku-100-miljardi-eiro-darba-vietu-saglabasanai

Ekos Research Associates Inc. (2011, September). Survey of Parents on Key Issues Related to Immunization. Final Report. Retrieved November 25, 2022, from https://www.ekospolitics.com/articles/0719.pdf

European Centre for Disease Prevention and Control. (n.d.). COVID-19 Vaccine Tracker. Retrieved February 11, 2023, from <u>https://qap.ecdc.europa.eu/public/extensions/COVID-19/vaccine-tracker.html#uptake-tab</u>

European Centre for Disease Prevention and Control. (2023, March). Interim analysis of COVID-19 vaccine effectiveness against Severe Acute Respiratory Infection due to laboratory-confirmed SARS-CoV-2 among individuals aged 20 years and older, ECDC multi-country study – fourth update. Retrieved March 27, 2023, from https://www.ecdc.europa.eu/sites/default/files/documents/COVID-19-vaccineindividuals-20-years-fourth-update-march-2023.pdf

European Institute for Gender Equality. (n.d.). *Essential workers*. Retrieved March 21, 2023, from

https://eige.europa.eu/topics/health/covid-19-and-gender-equality/essential-workers

European Medicines Agency. (n.d.-a). *COVID-19 vaccines: key facts*. Retrieved March 30, 2023, from <a href="https://www.ema.europa.eu/en/human-regulatory/overview/public-health-">https://www.ema.europa.eu/en/human-regulatory/overview/public-health-</a>

threats/coronavirus-disease-covid-19/treatments-vaccines/vaccines-covid-19/covid-19-vaccines-key-facts#vaccine-authorisation-section

European Medicines Agency. (n.d.-b). *Conditional marketing authorization*. Retrieved March 30, 2023, from <u>https://www.ema.europa.eu/en/human-regulatory/marketing-</u> <u>authorisation/conditional-marketing-authorisation</u>

European Medicines Agency. (n.d.-c). *COVID-19 vaccines: authorised*. Retrieved March 30, 2023, from https://www.ema.europa.eu/en/human-regulatory/overview/public-healththreats/coronavirus-disease-covid-19/treatments-vaccines/vaccines-covid-19/covid-19-vaccines-authorised

- Eyre, D. W., Taylor, D., Purver, M., Chapman, D., Fowler, T., Pouwels, K. B., Walker, A. S., Peto, T. E. A. (2022). Effect of Covid-19 Vaccination on Transmission of Alpha and Delta Variants. *The New England Journal of Medicine*, 386(8), 744-756.
- Fehr, E., & Gächter, S. (2000). Fairness and Retaliation: The Economics of Reciprocity. *The Journal of Economic Perspectives*, 14(3), 159-181. Retrieved from <a href="https://www.jstor.org/stable/pdf/2646924.pdf?refreqid=excelsior%3A2d9ff166cd264e">https://www.jstor.org/stable/pdf/2646924.pdf?refreqid=excelsior%3A2d9ff166cd264e</a> <a href="mailto:aebeb410e6147f8c93&ab\_segments=&origin=&initiator=&acceptTC=1">https://www.jstor.org/stable/pdf/2646924.pdf?refreqid=excelsior%3A2d9ff166cd264e</a> <a href="mailto:aebeb410e6147f8c93&ab\_segments=&origin=&initiator=&acceptTC=1">https://www.jstor.org/stable/pdf/2646924.pdf?refreqid=excelsior%3A2d9ff166cd264e</a> <a href="mailto:aebeb410e6147f8c93&ab\_segments=&origin=&initiator=&acceptTC=1">https://www.jstor.org/stable/pdf/2646924.pdf?refreqid=excelsior%3A2d9ff166cd264e</a>
- Feikin, D. R., Higdon, M. M., Abu-Raddad, L. J., Andrews, N., Araos, R., Goldberg, Y. (2022). Duration of effectiveness of vaccines against SARS-CoV-2 infection and COVID-19 disease: results of a systematic review and meta-regression. *The Lancet*, 399(10328), 924-944. doi: 10.1016/S0140-6736(22)00152-0

Feldmanis, K. (2021a, March 12). Iedzīvotāju negācijas par Vakcinācijas biroju skaidro ar tā izmantošanu valdības kļūdu piesegšanai. (Citizens' negativity about the Vaccination Office are explained by its use to cover up the government's mistakes). *Latvijas Sabiedriskie Mediji*. Retrieved from

https://www.lsm.lv/raksts/zinas/latvija/iedzivotaju-negacijas-par-vakcinacijas-birojuskaidro-ar-ta-izmantosanu-valdibas-kludu-

piesegsanai.a396444/?utm\_source=lsm&utm\_medium=theme&utm\_campaign=theme

Feldmanis, K. (2021b, March 15). Reorganizēs Vakcinācijas projekta biroju. (The Vaccination project office will be reorganized). *Latvijas Sabiedriskie Mediji*.
 Retrieved from <a href="https://www.lsm.lv/raksts/zinas/latvija/reorganizes-vakcinacijas-projekta-biroju.a396702/?utm\_source=lsm&utm\_medium=theme&utm\_campaign=theme">https://www.lsm.lv/raksts/zinas/latvija/reorganizes-vakcinacijas-projekta-biroju.a396702/?utm\_source=lsm&utm\_medium=theme&utm\_campaign=theme</a>

Feldmanis, K. (2021c, October 20). Visā Latvijā reģionālajās slimnīcās trūkst strādājošo; mēģina piesaistīt ar reklāmām un elastīgu darbalaiku. (There is a shortage of workers in regional hospitals throughout Latvia; attempts are being made to attract them with advertisements and flexible working hours). *Latvijas Sabiedriskie Mediji*. Retrieved from

https://www.lsm.lv/raksts/zinas/latvija/visa-latvija-regionalajas-slimnicas-trukststradajoso-megina-piesaistit-ar-reklamam-un-elastigu-darbalaiku.a426481/

Feng, A., Obolski, U., Stone, L., He, D. (2022). Modelling COVID-19 vaccine breakthrough infections in highly vaccinated Israel—The effects of waning immunity and third vaccination dose. *PLoS Global Public Health*, 2(11), Article e0001211. Retrieved from

https://journals.plos.org/globalpublichealth/article?id=10.1371/journal.pgph.0001211

Ferdinands, J. M., Rao, S., Dixon, B. E., Mitchell, P. K., DeSilva, M. B., Irving, S. A., Lewis, N., Natarajan, K., Stenehjem, E., Grannis, S. J., Han, J., McEvoy, C., Ong, T. C., Naleway, A. L., Reese, S. E., Embi, P. J., Dascomb, K., Klein, N. P., Griggs, E. P., ...
Fireman, B. (2022). Waning of vaccine effectiveness against moderate and severe covid-19 among adults in the US from the VISION network: test negative, case-control study. *The BMJ*, *379*, Article e072141. doi: 10.1136/bmj-2022-072141

- Fisman, D. N., Amoako, A., Tuite, A. R. (2022). Impact of population mixing between vaccinated and unvaccinated subpopulations on infectious disease dynamics: implications for SARS-CoV-2 transmission. *Canadian Medical Association Journal*, 194(16), 573-580. Retrieved from <u>https://www.cmaj.ca/content/194/16/E573</u>
- Flanigan, J. (2014). A defense of compulsory vaccination. HEC Forum, 26(1), 5-25.
- Fumadó, C. M., Aragonès, L., Areste, M. E., Manso, J. A. (2021). Medico-legal, ethical and deontological considerations of vaccination against COVID-19 in healthcare professionals. *Medicina Clinica*, 157(2), 79-84.
- Gandjour, A. (2022). Vaccination Mandates, Physically Forced Vaccination, and Rationing in the Intensive Care Unit: Searching for Ethical Coherence in the COVID-19 Pandemic. *The American Journal of Bioethics*, 22(11), 11-14.
- Giubilini, A. (2021). Vaccination ethics. *British Medical Bulletin*, *137*(1), 4–12. Retrieved from <a href="https://academic.oup.com/bmb/article/137/1/4/6047735?login=false">https://academic.oup.com/bmb/article/137/1/4/6047735?login=false</a>
- Giubilini, A. (2019). The Ethics of Vaccination. Basingstoke: Springer Nature.
- Giubilini, A., Douglas, T., Savulescu, J. (2018). The moral obligation to be vaccinated: utilitarianism, contractualism, and collective easy rescue. *Medicine, Health Care and Philosophy*, 21(4), 547-560.
- Greenwood, B. (2014). The contribution of vaccination to global health: past, present and future. Philosophical Transactions of The Royal Society B: Biological Sciences, 369(1645), 1-9.
- Gula, R. J. (2002). Nonsense: Red Herrings, Straw Men and Sacred Cows: How We Abuse Logic in Our Everyday Language. Mount Jackson, VA: Axios Press.
- Haas, E. J., Angulo, F. J., McLaughlin, J. M., Anis, E., Singer, S. R., Khan, F. (2021). Impact and effectiveness of mRNA BNT162b2 vaccine against SARS-CoV-2 infections and COVID-19 cases, hospitalisations, and deaths following a nationwide vaccination campaign in Israel: an observational study using national surveillance data. *The Lancet, 397*(10287), 1819-1829. doi: 10.1016/S0140-6736(21)00947-8

- Harmsen, I. A., Mollema, L., Ruiter, R. A. C., Paulussen, T. G. W., de Melker, H. E., Kok, G. (2013). Why parents refuse childhood vaccination: a qualitative study using online focus groups. *BMC Public Health*, 13(1), 1-8.
- Hause, A. M., Marquez, P., Zhang, B., Myers, T. R., Gee, J., Su, J. R., Blanc, P. G., Thomas, A., Thompson, D., Shimabukuro, T. T., Shay, D. K. (2022). Safety Monitoring of Bivalent COVID-19 mRNA Vaccine Booster Doses Among Persons Aged ≥12 Years United States, August 31–October 23, 2022. Centers for Disease Control and Prevention Morbidity and Mortality Weekly Report, 71(44), 1401-1406. doi: 10.15585/mmwr.mm7144a3
- Hausman, D. M., & Welch, B. (2010). Debate: To Nudge or Not to Nudge. Journal of Political Philosophy, 18(1), 123-136.
- Hazans, M. (2013). Emigration from Latvia: Recent trends and economic impact. In OECD (Ed.), *Coping with Emigration in Baltic and East European Countries* (pp. 65 110). OECD Publishing.
- Healy, J. (2022, January 7). Contemplating a COVID party for your kids? It's still a bad idea. Los Angeles Times. Retrieved from <u>https://www.latimes.com/science/story/2022-01-07/covid-party-for-your-kids-still-a-bad-idea</u>

Hengel, K. M. O., Burdorf, A., Pronk, A., Schlünssen, V., Stokholm, Z. A., Kolstad, H. A., van Veldhoven, K., Basinas, I., van Tongeren, M., Peters, S. (2022). Exposure to a SARS-CoV-2 infection at work: development of an international job exposure matrix (COVID-19-JEM). *Scandinavian Journal of Work, Environment & Health, 48*(1), 61-70. Retrieved from

https://www.ncbi.nlm.nih.gov/pmc/articles/PMC8729167/#:~:text=Number%2C%20n ature%20of%20contacts%20and,(eg%2C%20healthcare%20worker).

Hinders, D. (2023, March 3). *Which Countries Use Progressive Taxes?* [Blog post]. Retrieved from

https://www.smartcapitalmind.com/which-countries-use-progressive-taxes.htm

- Huang, C., Yang, L., Pan, J., Xu, X., Peng, R. (2022). Correlation between vaccine coverage and the COVID-19 pandemic throughout the world: Based on real-world data. *Journal of Medical Virology*, 94(5), 2181-2187. doi: 10.1002/jmv.27609
- Hughes, M. T., Auwaerter, P. G., Ehmann, M. R., Garibaldi, B. T., Golden, S. H., Lorigiano, T. J., O'Conor, K. J., Kachalia, A., Kahn, J. (2021). Opinion: The importance of offering vaccine choice in the fight against COVID-19. *Proceedings of the National Academy of Sciences, 118*(43), 1-4.
- Insel, K. (2012). Treating Children Whose Parents Refuse to Have Them Vaccinated. American Medical Association Journal of Ethics – Virtual Mentor, 14(1), 17-22.
- International Monetary Fund. (2020, October). *World Economic Outlook: A Long and Difficult Ascent*. Retrieved March 30, 2023, from <a href="https://www.imf.org/en/Publications/WEO/Issues/2020/09/30/world-economic-outlook-october-2020">https://www.imf.org/en/Publications/WEO/Issues/2020/09/30/world-economic-outlook-october-2020</a>
- Johnson, A. G., Linde, L., Ali, A. R., DeSantis, A., Shi, M., Adam, C., Armstrong, B., Armstrong, B., Asbell, M., Auche, S., Bayoumi, N. S., Bingay, B., Chasse, M., Christofferson, S., Cima, M., Cueto, K., Cunningham, S., Delgadillo, J., Dorabawila, ... Scobie, H. M. (2023). COVID-19 Incidence and Mortality Among Unvaccinated and Vaccinated Persons Aged ≥12 Years by Receipt of Bivalent Booster Doses and Time Since Vaccination — 24 U.S. Jurisdictions, October 3, 2021–December 24, 2022. Centers for Disease Control and Prevention Morbidity and Mortality Weekly Report, 72(6), 145-152. doi: 10.15585/mmwr.mm7206a3
- Johnson, J. (2021, February 24). Why do people have pox parties? *Medical News Today*. Retrieved from <u>https://www.medicalnewstoday.com/articles/239450</u>
- Kant, I. (1998). *Groundwork of the Metaphysics of Morals*, trans. M. Gregor. Cambridge: Cambridge University Press.
- Kiger, M. E., & Varpio, L. (2020). Thematic analysis of qualitative data: AMEE Guide No. 131, *Medical Teacher*, 42(8), 846-854.

- Kinca, A. (December 28, 2020). Pirmie cilvēki Latvijā saņēmuši vakcīnu pret Covid-19. (The first people in Latvia have received the vaccine against COVID-19). *Latvijas Sabiedriskie Mediji*. Retrieved from <a href="https://www.lsm.lv/raksts/zinas/latvija/pirmie-cilveki-latvija-sanemusi-vakcinu-pret-covid-19.a386884/">https://www.lsm.lv/raksts/zinas/latvija/pirmie-cilveki-latvija-sanemusi-vakcinu-pret-covid-19.a386884/</a>
- Kinca, A., Miksons, K., Milovs, I., Ķezberis, U., Amoliņš, G., Ozola-Balode, Z. (2020, December 26). Latvijā un citur Eiropā piegādā pirmās vakcīnas pret Covid-19. (The first vaccines against COVID-19 are delivered in Latvia and elsewhere in Europe). *Latvijas Sabiedriskie Mediji*. Retrieved from <u>https://www.lsm.lv/raksts/zinas/latvija/latvija-un-citur-eiropa-piegada-pirmasvakcinas-pret-covid-19.a386770/</u>
- King, J., & Ferraz, O. L. M. (2021). Legal, Constitutional, and Ethical Principles for Mandatory Vaccination Requirements for Covid-19. In *Lex-Atlas: Covid-19*. Retrieved from <u>https://lexatlas-c19.org/vaccination-principles/</u>
- Kornhauser, L. (2022). The Economic Analysis of Law. In E. N. Zalta (Ed.), *The Stanford Encyclopedia of Philosophy* (Spring 2022 ed.). Retrieved from <a href="https://plato.stanford.edu/entries/legal-econanalysis/">https://plato.stanford.edu/entries/legal-econanalysis/</a>
- Kuzmina, I. (2021, July 21). Kategoriski pret potēšanos un gatavi pamest darbu. Kāpēc skolotāji negrib vakcinēties? (Categorically against vaccination and ready to quit their jobs. Why do teachers not want to get vaccinated?) LA.lv. Retrieved from <u>https://www.la.lv/kapec-skolotaji-negrib-vakcineties</u>
- Labuschagne, A. (2003). Qualitative Research Airy Fairy or Fundamental? *The Qualitative Report*, 8(1), 100-103.
- LaFollette, H. (1997). Ethics in Practice: An Anthology. Cambridge, MA: Blackwell.
- Larson, H., de Figueiredo, A., Karafillakis, E., Rawal, M. (2018). State of Vaccine Confidence in the EU 2018. Luxembourg: Publications Office of the European Union. doi: 10.2875/241099

- Lazarus, J. V., Ratzan, S. C., Palayew, A., Gostin, L. O., Larson, H. J., Rabin, K., Kimball, S., El-Mohandes, A. (2020). A global survey of potential acceptance of a COVID-19 vaccine. *Nature Medicine*, 27(2), 225-228.
- Lee, C., Whetten, K., Omer, S., Pan, W., Salmon, D. (2016). Hurdles to herd immunity: Distrust of government and vaccine refusal in the US, 2002-2003. *Vaccine*, 34(34), 3972-3978.
- Leshem, E., & Wilder-Smith, A. (2021). COVID-19 vaccine impact in Israel and a way out of the pandemic. *The Lancet*, *397*(10287), 1783-1785.
- LETA. (2021a, December 8). Atklāj, cik daudz pedagogu pametuši darbu Covid-19 sertifikāta neesamības dēļ. (It is revealed how many pedagogues have left work due to a lack of a COVID-19 certificate). LA.lv. Retrieved from <u>https://www.la.lv/atklaj-cik-daudz-pedagogu-pametusi-darbu-covid-19-sertifikataneesamibas-del</u>
- LETA. (2021b, May 18). Valsts ir pārmaksājusi par Covid-19 testēšanas pakalpojumiem, secinājusi Valsts kontrole. (The state has overpaid for COVID-19 testing services, State Audit Office of the Republic of Latvia concludes). *LA.lv*. Retrieved from <u>https://www.la.lv/valsts-ir-parmaksajusi-par-covid-19-testesanas-pakalpojumiemsecinajusi-valsts-kontrole</u>
- Lewandowsky, S., Holford, D., Schmid, P. (2022). Public policy and conspiracies: The case of mandates. *Current Opinion in Psychology*, 47(49), 1-6.
- Link-Gelles, R., Ciesla, A. A., Roper, L. E., Scobie, H. M., Ali, A. R., Miller, J. D., Wiegand, R. E., Accorsi, E. K., Verani, J. R., Shang, N., Derado, G., Britton, A., Smith, Z. R., Fleming-Dutra, K. E. (2023). Early Estimates of Bivalent mRNA Booster Dose Vaccine Effectiveness in Preventing Symptomatic SARS-CoV-2 Infection Attributable to Omicron BA.5– and XBB/XBB.1.5–Related Sublineages Among Immunocompetent Adults Increasing Community Access to Testing Program, United States, December 2022–January 2023. *Centers for Disease Control and Prevention Morbidity and Mortality Weekly Report, 72*(5), 119-124. doi: 10.15585/mmwr.mm7205e1

- Link-Gelles, R., Levy, M. E., Gaglani, M., Irving, S. A., Stockwell, M., Dascomb, K., DeSilva, M. B., Reese, S. E., Liao, I. C., Ong, T. C., Grannis, S. J., McEvoy, C., Patel, P., Klein, N. P., Hartmann, E., Stenehjem, E., Natarajan, K., Naleway, A. L., Murthy, K., ... Tenforde, M. W. (2022). Effectiveness of 2, 3, and 4 COVID-19 mRNA Vaccine Doses Among Immunocompetent Adults During Periods when SARS-CoV-2 Omicron BA.1 and BA.2/BA.2.12.1 Sublineages Predominated VISION Network, 10 States, December 2021–June 2022. *Centers for Disease Control and Prevention Morbidity and Mortality Weekly Report, 71*(29), 931-939. doi: 10.15585/mmwr.mm7129e1
- Lu, M. (2020, April 20). These are the occupations with the highest COVID-19 risk. *World Economic Forum*. Retrieved from <u>https://www.weforum.org/agenda/2020/04/occupations-highest-covid19-risk/</u>
- MacDonald, N. E., The SAGE Working Group on Vaccine Hesitancy. (2015). Vaccine hesitancy: Definition, scope and determinants. *Vaccine*, *34*(14), 4161-4164.
- Marien, S., Hooghe, M. (2011). Does political trust matter? An empirical investigation into the relation between political trust and support for law compliance. *European Journal of Political Research*, *50*(2), 267-291.
- Maslow, A. H. (1954). *Motivation and personality*. New York, NY: Harper & Row Publishers.
- Mayo Clinic Staff. (n.d.-a). *Herd immunity and COVID-19: What you need to know*. Retrieved March 20, 2023, from <u>https://www.mayoclinic.org/diseases-conditions/coronavirus/in-depth/herd-immunity-and-coronavirus/art-20486808</u>

Mayo Clinic Staff. (n.d.-b). *How well do face masks protect against COVID-19?* Retrieved March 29, 2023, from <u>https://www.mayoclinic.org/diseases-conditions/coronavirus/in-depth/coronavirus-mask/art-20485449</u>

- McNicholas, C., & Poydock, M. (2020, May 19). *Who are essential workers?* [Blog post]. Retrieved from <u>https://www.epi.org/blog/who-are-essential-workers-a-comprehensive-look-at-their-wages-demographics-and-unionization-rates/</u>
- Mill, J. S. (1859). On Liberty. London: John W. Parker and Son.
- Miller, K. (2023, March 7). The Most Common Long COVID Symptoms Doctors Are Seeing Right Now. Prevention. Retrieved from <u>https://www.prevention.com/health/a43234324/most-common-long-covidsymptoms/#what-are-the-most-common-long-covid-symptoms-0</u>
- Mills, M. C., & Rüttenauer, T. (2022). The effect of mandatory COVID-19 certificates on vaccine uptake: synthetic-control modelling of six countries. *The Lancet*, 7(1), 15-22.
   Retrieved from <a href="https://www.sciencedirect.com/science/article/pii/S2468266721002735">https://www.sciencedirect.com/science/article/pii/S2468266721002735</a>
- Ministru Kabinets. (2021a, April 8). Ministru kabineta sēde. 08.04.2021. (Cabinet of Ministers meeting 08.04.2021.) [Video file]. Retrieved from <u>https://www.youtube.com/watch?v=3MBuPGGiHNk</u>
- Ministru Kabinets. (2021b). Epidemioloģiskās drošības pasākumi Covid-19 infekcijas izplatības ierobežošanai. (Epidemiological Safety Measures for the Containment of the Spread of COVID-19 Infection). Retrieved from <u>https://likumi.lv/ta/id/326513-epidemiologiskas-drosibas-pasakumi-covid-19-</u> infekcijas-izplatibas-ierobezosanai
- Ministru Kabinets. (2021c). Informatīvais ziņojums "Iespējamie vakcinēšanās pret Covid-19 infekciju motivējošie līdzekļi." (Informative report "Potential motivational agents for vaccination against Covid-19 infection.") Retrieved March 29, 2023, from <a href="https://view.officeapps.live.com/op/view.aspx?src=https%3A%2F%2Ftap.mk.gov.lv%2Fdoc%2F2021\_04%2FTM\_InfoZin\_vakcin\_motivac\_260.972.docx&wdOrigin=B\_ROWSELINK">https://view.officeapps.live.com/op/view.aspx?src=https%3A%2F%2Ftap.mk.gov.lv%2Fdoc%2F2021\_04%2FTM\_InfoZin\_vakcin\_motivac\_260.972.docx&wdOrigin=B\_ROWSELINK</a>
- Ministru Kabinets. (2020a). Par ārkārtējās situācijas izsludināšanu. (Regarding Declaration of the Emergency Situation). Retrieved from <u>https://likumi.lv/ta/id/318517</u>

- Ministru Kabinets. (2020b). Noteikumi par dīkstāves pabalstu darbiniekiem, kurus skar Covid-19 izplatība. (Regulations Regarding the Allowance for Idle Time for the Employees Affected by COVID-19). Retrieved from <u>https://likumi.lv/ta/id/313429-noteikumi-par-dikstaves-pabalstu-darbiniekiem-kurusskar-covid-19-izplatiba</u>
- Mohammed, I., Nauman, A., Paul, P., Ganesan, S., Chen, K. H., Jalil, S. M. S., Jaouni, S. H., Kawas, H., Khan, W. A., Vattoth, A. L., Al-Hashimi, Y. A., Fares, A., Zaeghlache, R., Zakaria, D. (2022). The efficacy and effectiveness of the COVID-19 vaccines in reducing infection, severity, hospitalization, and mortality: a systematic review. *Human Vaccines & Immunotherapeutics*, *18*(1), Article e2027160. doi: 10.1080/21645515.2022.2027160
- Morens, D. M., Folkers, G. K., & Fauci, A. S. (2022). The Concept of Classical Herd Immunity May Not Apply to COVID-19. *The Journal of Infectious Diseases*, 226(2), 195-198. Retrieved from <u>https://academic.oup.com/jid/article/226/2/195/6561438</u>
- Mumcuoglu, O., Mackos, D. & Vardon, E. (2021, December 31). Factbox: Countries making COVID-19 vaccines mandatory. *Reuters*. Retrieved from <u>https://www.reuters.com/business/healthcare-pharmaceuticals/countries-making-</u> covid-19-vaccines-mandatory-2021-08-16/
- Nacionālais Veselības Dienests. (2021, July 7). Aicina sekot līdzi digitālo Covid-19 sertifikātu derīguma termiņam; noteiktos gadījumos sertifikāts var būt atsaukts. (Invitation to follow the validity of digital COVID-19 certificates; in certain cases, the certificate may be revoked). Retrieved from <u>https://www.vmnvd.gov.lv/lv/jaunums/aicina-sekot-lidzi-digitalo-covid-19-</u> sertifikatu-deriguma-terminam-noteiktos-gadijumos-sertifikats-var-atsaukts
- Nadeau, B. L. (2021, November 22). These Fools Thought a COVID Party Sounded Fun. Now They Are on Respirators. *Daily Beast*. Retrieved from <u>https://www.thedailybeast.com/these-fools-thought-a-covid-party-sounded-fun-now-they-are-on-respirators</u>

Najera, R. F. (2021, August 20). The Nuremberg Code Does Not Apply to Licensed, Approved, and Authorized Vaccinations, Only to the Research That Led to That Licensure, Approval, and Authorization [Blog post]. Retrieved from <u>https://historyofvaccines.org/blog/the-nuremberg-code-does-not-apply-to-licensed-approved-and-authorized-vaccinations-only-to-the-research-that-led-to-that-licensure-approval-and-authorization</u>

Nodarbinātības Valsts Aģentūra. (2021, December). Pārskats par bezdarba situāciju valstī (2021. gada decembris). (Overview of the unemployment situation in the country (December 2021)). Retrieved March 20, 2023, from <u>https://www.nva.gov.lv/lv/media/13648/download?attachment</u>

Nuremberg Code. (n.d.). Retrieved March 30, 2023, from <u>https://research.unc.edu/human-research-ethics/resources/ccm3\_019064/</u>

- OECD. (n.d.). *Trust in government*. Retrieved February 11, 2023, from https://data.oecd.org/gga/trust-in-government.htm
- OECD. (2013, November). *Government at a Glance 2013*. Retrieved February 23, 2023, from <u>https://www.oecd-ilibrary.org/governance/government-at-a-glance-2013\_gov\_glance-2013\_en</u>
- Office for National Statistics. (2023). Prevalence of ongoing symptoms following coronavirus (COVID-19) infection in the UK [Data file]. Retrieved March 30, 2023, from https://www.ons.gov.uk/peoplepopulationandcommunity/healthandsocialcare/conditio nsanddiseases/datasets/alldatarelatingtoprevalenceofongoingsymptomsfollowingcoron aviruscovid19infectionintheuk

Oficiālās statistikas portāls. (2022). Bezdarbnieku skaits un bezdarba līmenis pēc dzimuma un vecuma grupām 2002Q1-2022Q4 (Number of unemployed persons and level of unemployment by gender and age groups 2002Q1-2022Q4) [Data file]. Retrieved March 30, 2023, from <u>https://data.stat.gov.lv/pxweb/lv/OSP\_PUB/START\_EMP\_NBBA\_NBB1/NBB16</u> 0c/ O'Mahoney, L. L., Routen, A., Gillies, C., Ekezie, W., Welford, A., Zhang, A., Karamchandani, U., Simms-Williams, N., Cassambai, S., Ardavani, A., Wilkinson, T. J., Hawthorne, G., Curtis, F., Kingsnorth, A. P., Almaqhawi, A., Ward, T., Ayoubkhani, D., Banerjee, A., Calvert, M., ... Khunti, K. (2023). The prevalence and long-term health effects of Long Covid among hospitalised and non-hospitalised populations: a systematic review and meta-analysis. *eClinicalMedicine*, 55, Article 101762. Retrieved from <u>https://www.thelancet.com/journals/eclinm/article/PIIS2589-5370(22)00491-</u>

6/fulltext#articleInformation

- Oster, M. E., Shay, D. K., Su, J. R., Gee, J., Creech, C. B., Broder, K. R., Edwards, K., Soslow, J. H., Dendy, J. M., Schlaudecker, E., Lang, S. M., Barnett, E. D., Ruberg, F. L., Smith, M. J., Campbell, M. J., Lopes, R. D., Sperling, L. S., Baumblatt, J. A., Thompson, D. L., ... Shimabukuro, T. T. (2022). Myocarditis Cases Reported After mRNA-Based COVID-19 Vaccination in the US From December 2020 to August 2021. *The Journal of the American Medical Association*, 327(4), 331-340. Retrieved from https://jamanetwork.com/journals/jama/fullarticle/2788346
- O'Sullivan, S. (2022). COVID-19 Mandatory Vaccination Ethical and Human Rights. Retrieved from https://assets.gov.ie/217200/aa9eb054-3804-4ae3-8e03-35935fef2c52.pdf
- Ozola-Balode, Z. (2021, May 5). Vakcinācija pret Covid-19: jaunieši skeptiskāki, birojs plāno uzrunāt neizlēmušos. (Vaccination against COVID-19: young people are more skeptical, the Office plans to address the undecided.) *Latvijas Sabiedriskie Mediji*. Retrieved from <u>https://www.lsm.lv/raksts/zinas/latvija/vakcinacija-pret-covid-19-jauniesi-skeptiskakibirojs-plano-uzrunat-neizlemusos.a403309/</u>
- Pandemic (n.d.). In *Merriam-Webster.com Dictionary*. Retrieved November 22, 2022, from <u>https://www.merriam-webster.com/dictionary/pandemic</u>

- Pētījumu Centrs SKDS. (2021a, December). Pētījums par sabiedrības attieksmi pret COVID-19. Latvijas iedzīvotāju internetaptauja. (Research on society's attitude towards COVID-19. Internet survey of Latvian citizens). Retrieved November 25, 2022, from <u>http://petijumi.mk.gov.lv/sites/default/files/title\_file/Sab\_attieksme\_pret\_Covid\_PETI\_JUMS.pdf</u>
- Pētījumu Centrs SKDS. (2021b, August). Pētījums par sabiedrības attieksmi pret COVID-19. Latvijas iedzīvotāju aptauja. (Research on society's attitude towards COVID-19. Survey of Latvian citizens). Retrieved March 30, 2023, from <u>http://petijumi.mk.gov.lv/sites/default/files/title\_file/SKDS\_aptauja\_Aug2021.pdf</u>
- Petraviča, R. (2021, September 2). Vaccination must be purely voluntary [Meeting transcript]. In the 13th Saeima's fall session's first meeting. Retrieved from <u>https://saeima.lv/lv/transcripts/view/2294</u>
- Philippe, C., Bar-Yam, Y., Bilodeau, S., Gershenson, C., Raina, S. K., Chiou, S. T., Nyborg, G. A., Schneider, M. F. (2023). Mass testing to end the COVID-19 public health threat. *The Lancet Regional Health Europe*, 25, Article 100574. Retrieved from <a href="https://www.thelancet.com/journals/lanepe/article/PIIS2666-7762(22)00270-8/fulltext">https://www.thelancet.com/journals/lanepe/article/PIIS2666-7762(22)00270-8/fulltext</a>
- Porat, T., Burnell, R., Calvo, R. A., Ford, E., Paudyal, P., Baxter, W. L., Parush, A. (2021). "Vaccine Passports" May Backfire: Findings from a Cross-Sectional Study in the UK and Israel on Willingness to Get Vaccinated against COVID-19. Vaccines, 9(8), Article 902. Retrieved from <u>https://www.mdpi.com/2076-393X/9/8/902</u>
- Poteyeva, M. (2018, December 17). Social capital. In *Encyclopedia Britannica*. Retrieved March 30, 2023, from <u>https://www.britannica.com/topic/social-capital</u>

Г

Powell, A. (2021, February 25). Vaccines can get us to herd immunity, despite the variants. *The Harvard Gazette*. Retrieved from <u>https://news.harvard.edu/gazette/story/2021/02/vaccines-should-end-the-pandemic-despite-the-variants-say-experts/</u> Raimonds Upmalis. (2021, September 22). Sakarā ar studiju atsākšanu Latvijas Universitātē steidzami meklēju iespēju saslimt ar CO... vīrusu. Lūdzu, palīdziet ar kontaktiem! Rados neviena slimnieka (Due to studies restarting at the University of Latvia, I am urgently looking for a chance to get infected with the CO... virus. Please, help with contacts! There are no sick people among my relatives) [Image attached] [Status update]. Facebook. Retrieved from https://www.facebook.com/photo?fbid=1267437727027278&set=a.17416273302145

<u>5</u>

- Saeima. (n.d.). *About the Saeima*. Retrieved November 24, 2022, from <u>https://www.saeima.lv/en/about-saeima</u>
- Saeima. (2009). Pacientu tiesību likums. (Law on the Rights of Patients). Retrieved from https://likumi.lv/ta/id/203008-pacientu-tiesibu-likums
- Saeima. (2020). *Law on the Management of the Spread of COVID-19 Infection*. Retrieved from <u>https://likumi.lv/ta/en/en/id/315278</u>
- Saghai, Y. (2014). Radically questioning the principle of the least restrictive alternative: a reply to Nir Eyal. *International Journal of Health Policy and Management*, 3(6), 349-350. Retrieved from <u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4226626/</u>
- Satversmes Sapulce. (1998). *Latvijas Republikas Satversme*. (*Constitution of the Republic of Latvia*). Retrieved from <u>https://likumi.lv/ta/id/57980#p115</u>

Schwartz, M. C. (2020). The Ethics of Pandemics. Peterborough: Broadview Press.

Scobie, H. M., Johnson, A. G., Suthar, A. B., Severson, R., Alden, N. B., Balter, S., Bertolino, D., Blythe, D., Brady, S., Cadwell, B., Cheng, I., Davidson, S., Delgadillo, J., Devinney, K., Duchin, J., Duwell, M., Fisher, R., Fleischauer, A., Grant, A., ... Silk, B. J. (2021). Monitoring Incidence of COVID-19 Cases, Hospitalizations, and Deaths, by Vaccination Status — 13 U.S. Jurisdictions, April 4–July 17, 2021. *Centers for Disease Control and Prevention Morbidity and Mortality Weekly Report*, 70(37), 1284-1290. doi: 10.15585/mmwr.mm7037e1

- Sikora, D., Rzymski, P. (2022). COVID-19 Vaccination and Rates of Infections,
  Hospitalizations, ICU Admissions, and Deaths in the European Economic Area during
  Autumn 2021 Wave of SARS-CoV-2. *Vaccines*, *10*(3), Article 437.
  doi: 10.3390/vaccines10030437
- Simpson, S. D. (2022, April 14). The Cost of Unemployment to the Economy. *Investopedia*. Retrieved from <u>https://www.investopedia.com/financial-edge/0811/the-cost-of-unemployment-to-the-</u> economy.aspx
- Skelton, A., & Forsberg, L. (2020). In M. C. Schwartz (Ed.), *The Ethics of Pandemics* (pp. 198-203). Peterborough: Broadview Press.
- Slimību profilakses un kontroles centrs. (2022). *Covid-19 statistika*. (*COVID-19 statistics*). Retrieved March 27, 2023, from <u>https://www.spkc.gov.lv/lv/covid-19-statistika</u>
- Slimību profilakses un kontroles centrs. (2023). *Profilakses pasākumi un rekomendācijas.* (*Preventive measures and recommendations*). Retrieved March 29, 2023, from <u>https://www.spkc.gov.lv/lv/profilakses-pasakumi-un-rekomendacijas</u>
- Slimību profilakses un kontroles centrs, Nacionālais veselības dienests, Zāļu valsts aģentūra, Veselības ministrija, Imunizācijas valsts padome. (2022, October 31). Covid-19 vakcinācijas rokasgrāmata. (COVID-19 Vaccination Guide). Retrieved from <u>https://www.spkc.gov.lv/lv/media/17672/download</u>
- Smith, P. J., Humiston, S. G., Marcuse, E. K., Zhao, Z., Dorell, C. G., Howes, C., Hibbs, B. (2011). Parental delay or refusal of vaccine doses, childhood vaccination coverage at 24 months of age, and the Health Belief Model. *Public Health Reports*, 126(2), 135-146.
- Solis-Moreira, J. (2021, November 13). How did we develop a COVID-19 vaccine so quickly? *Medical News Today*. Retrieved from <u>https://www.medicalnewstoday.com/articles/how-did-we-develop-a-covid-19-vaccine-so-quickly</u>

- Sondare, M. (2021, August 17). Vakcinācijas birojs izveidots, lai 'visus degunus dabūtu vienā virzienā', saka Pavļuts. (The Vaccination Office was created to 'get everyone in the same direction', says Pavļuts). *Delfi*. Retrieved from <u>https://www.delfi.lv/news/national/politics/vakcinacijas-birojs-izveidots-lai-visusdegunus-dabutu-viena-virziena-saka-pavluts.d?id=53495173</u>
- Spiliopoulos, L. (2022). On the effectiveness of COVID-19 restrictions and lockdowns: Pan metron ariston. *BMC Public Health*, 22, Article 1842. Retrieved from <u>https://bmcpublichealth.biomedcentral.com/articles/10.1186/s12889-022-14177-7</u>
- Sprengholz, P., Felgendreff, L., Böhm, R., Betsch, C. (2022). Vaccination policy reactance: Predictors, consequences, and countermeasures. *Journal of Health Psychology*, 27(6), 1394-1407. Retrieved from <u>https://journals.sagepub.com/doi/full/10.1177/13591053211044535</u>
- Sprengholz, P., Betsch, C., & Böhm, R. (2021). Reactance revisited: Consequences of mandatory and scarce vaccination in the case of COVID- 19. *Applied Psychology: Health and Well-Being*, 13(4), 986-995. Retrieved from <u>https://iaap-journals.onlinelibrary.wiley.com/doi/full/10.1111/aphw.12285</u>
- Stefanoff, P., Mamelund, S. E., Robinson, M., Netterlid, E., Tuells, J., Bergsaker, M. A. R., Heijbel, H., Yarwood, J. (2010). Tracking parental attitudes on vaccination across European countries: The Vaccine Safety, Attitudes, Training and Communication Project (VACSATC). *Vaccine*, 28(35), 5731-5737.
- Stepaņenko, J. (2021, August 4). If we talk about human [Meeting transcript]. In the 13th Saeima's emergency session's remote meeting. Retrieved from <u>https://saeima.lv/lv/transcripts/view/2287</u>

Swatridge, C. (2014). Effective argument andtical thinking. Oxford University Press.

Sween, L., Ekeoduru, R., Mann, D. (2022). Ethics and Pitfalls of Vaccine Mandates. ASA Monitor, 86(2), 24–25. Retrieved from <u>https://pubs.asahq.org/monitor/article/86/2/24/118298/Ethics-and-Pitfalls-of-Vaccine-Mandates</u>

- Talic, S., Shah, S., Wild, H., Gasevic, D., Maharaj, A., Ademi, Z., Li, X., Xu, W., Mesa-Eguiagaray, I., Rostron, J., Theodoratou, E., Zhang, X., Motee, A., Liew, D., Ilic, D. (2021). Effectiveness of public health measures in reducing the incidence of covid-19, SARS-CoV-2 transmission, and covid-19 mortality: systematic review and meta-analysis. *The BMJ*, *375*, Article e068302. Retrieved from <a href="https://www.bmj.com/content/375/bmj-2021-068302">https://www.bmj.com/content/375/bmj-2021-068302</a>
- Trading Economics. (2022). Latvia Coronavirus COVID-19 Vaccination Total [Data file]. Retrieved March 31, 2023, from <u>https://tradingeconomics.com/latvia/coronavirus-vaccination-total</u>
- Trent, M., Seale, H., Chughtai, A. A., Salmon, D., MacIntyre, C. R. (2022). Trust in government, intention to vaccinate and COVID-19 vaccine hesitancy: A comparative survey of five large cities in the United States, United Kingdom, and Australia. *Vaccine*, 40(17), 2498-2505.
- TV3. (2022, October 6). Aptauja: 62% iedzīvotāju neplāno vakcinēties ar uzlabotajām Covid-19 vakcīnām. (Survey: 62% of citizens aren't planning to vaccinate with the improved COVID-19 vaccines). TV3 Ziņas. Retrieved from <u>https://zinas.tv3.lv/900-sekundes/aptauja-62-iedzivotaju-neplano-vakcineties-aruzlabotajam-covid-19-vakcinam/</u>
- United Nations International Children's Emergency Fund. (2021). 43% citizens believe in a conspiracy theory regarding the concealment of information about vaccines.
   Retrieved November 27, 2022, from
   <u>https://www.unicef.org/montenegro/en/stories/43-citizens-believe-conspiracy-theory-regarding-concealment-information-about-vaccines</u>
- United Nations. (n.d.). *Democracy*. Retrieved March 30, 2023, from <u>https://www.un.org/en/global-</u> <u>issues/democracy#:~:text=and%20fundamental%20freedoms-</u> <u>,Freedom%20of%20association,the%20will%20of%20the%20people</u>
- United Nations. (1948). Universal Declaration of Human Rights. Retrieved March 30, 2023, from <u>https://www.un.org/en/about-us/universal-declaration-of-human-rights</u>

- United Nations. (2021, July 20). *Trust in public institutions: Trends and implications for economic security*. Retrieved from https://www.un.org/development/desa/dspd/2021/07/trust-public-institutions/
- Vaccine refusal. (n.d.). In *Medical Dictionary*. Retrieved January 10, 2023, from <u>https://medical-dictionary.thefreedictionary.com/vaccine+refusal</u>
- Valsts Izglītības Informācijas Sistēma. (2019). Pedagogu skaits 2018./2019. m.g. (Number of pedagogues in the study year 2018./2019.) [Data file]. Retrieved March 20, 2023, from <u>https://www.viis.gov.lv/dati/pedagogu-skaits</u>
- van Brugen, I. (2021, November 26). Man Who Went To COVID Party to Build Immunity Dies From the Virus. *Newsweek*. Retrieved from <u>https://www.newsweek.com/covid-party-austria-italy-bolzano-man-dies-virus-green-pass-immunity-1653601</u>
- van den Hoven, M. (2012). Why One Should Do One's Bit: Thinking about Free Riding in the Context of Public Health Ethics. *Public Health Ethics*, 5(2), 154-160. Retrieved from <u>https://academic.oup.com/phe/article/5/2/154/1494175</u>
- Verweij, M. (2022). The (Un)fairness of Vaccination Freeriding. *Public Health Ethics*, 15(3), 233-239. Retrieved from <u>https://academic.oup.com/phe/article/15/3/233/6827162</u>
- Verweij, M. & Dawson, A. (2004). Ethical principles for collective immunisation programmes. *Vaccine*, 22(23-24), 3122-3126. doi: 10.1016/j.vaccine.2004.01.062
- Veselības ministrija. (2022). Par mutes un deguna aizsegu lietošanu. (About the use of face masks). Retrieved from <u>https://www.vm.gov.lv/lv/par-mutes-un-deguna-aizsegu-lietosanu</u>

- Veselības ministrija. (2021a). Veselības ministrija aicina uzdot savus jautājumus par vakcināciju un tiešsaistes sarunā saņemt atbildes no vadošajiem nozares ekspertiem (video). (The Ministry of Health invites you to ask questions about vaccination and receive answers from leading industry experts in a webinar). Retrieved from https://www.vm.gov.lv/lv/jaunums/veselibas-ministrija-aicina-uzdot-savusjautajumus-par-vakcinaciju-un-tiessaistes-saruna-sanemt-atbildes-no-vadosajiemnozares-ekspertiem-video
- Veselības ministrija. (2021b). Bez vakcīnas nav uzvaras! Veselības ministrija uzsāk vakcinācijas aptveri veicinošu reklāmas kampaņu\* (There is no victory without a vaccine! The Ministry of Health has launched a promotional vaccination uptake advertisement campaign\*). Retrieved from https://www.vm.gov.lv/lv/jaunums/bez-vakcinas-nav-uzvaras-veselibas-ministrijauzsak-vakcinacijas-aptveri-veicinosu-reklamas-kampanu
- Veselības ministrija. (2021c, September 21). Covid-19 vakcinācijas gaitas ziņojums (Information on COVID-19 vaccination progress). Retrieved from <u>https://tapportals.mk.gov.lv/meetings/cabinet\_ministers/884a8552-0a22-4712-aebd-4accd7fe29c8</u>
- Wilkenfield, D., & Johnson, C. M. (2022). In Defense of Vaccine Mandates: An Argument from Consent Rights. *Public Health Ethics*, 15(1), 27-40. Retrieved from <u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC9383732/</u>
- The World Bank. (2022). *Net migration Latvia* [Data file]. Retrieved March 30, 2023, from <a href="https://data.worldbank.org/indicator/SM.POP.NETM?locations=LV">https://data.worldbank.org/indicator/SM.POP.NETM?locations=LV</a>
- World Health Organization. (n.d.). Coronavirus disease (COVID-19) pandemic. Retrieved November 27, 2022, from https://www.who.int/europe/emergencies/situations/covid-19
- World Health Organization. (2021a). Coronavirus disease (COVID-19): Vaccine research and development. Retrieved November 25, 2022, from <u>https://www.who.int/news-room/questions-and-answers/item/coronavirus-disease-(covid-19)-vaccine-research-and-development</u>

World Health Organization. (2021b). Vaccines and immunization: What is vaccination? Retrieved November 22, 2022 from <u>https://www.who.int/news-room/questions-and-answers/item/vaccines-and-immunization-what-is-vaccination</u>

- Yanovskiy, M., & Socol, Y. (2022). Are Lockdowns Effective in Managing Pandemics? International Journal of Environmental Research and Public Health, 19(15), Article 9295. Retrieved from <u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC9368251/</u>
- Zvērs, J. (2022, August 31). Latvijā trūkst skolotāju; dažviet šī iemesla dēļ nenotiks atsevišķi mācību priekšmeti. (There is a shortage of teachers in Latvia; in some places, unfortunately, certain classes won't take place because of this). *TV3 Ziņas*. Retrieved from <u>https://zinas.tv3.lv/latvija/latvija-trukst-skolotaju-dazviet-si-iemesla-del-nenotiks-atseviski-macibu-prieksmeti/</u>
- Zvirbulis, Ģ. (2021, March 11). "Oribalt" noliktavā "iesprūdušo" Covid-19 vakcīnu izvadāšanai mēģina piesaistīt palīdzību. (Trying to attract help to distribute the "stuck" COVID-19 vaccines from "Oribalt" warehouse). Latvijas Sabiedriskie Mediji.
   Retrieved from <u>https://www.lsm.lv/raksts/zinas/latvija/oribalt-noliktava-iespruduso-covid-19-vakcinu-izvadasanai-megina-piesaistit-palidzibu.a396306/</u>
- Zvirbulis, Ģ. (2020, November 30). Survey: Latvian trust in official information is dwindling. *Latvijas Sabiedriskie Mediji*. Retrieved from <u>https://eng.lsm.lv/article/society/society/survey-latviantrust-in-official-information-is-</u> <u>dwindling.a383514/</u>
- Žilde-Krēvica, K. (2021, May 1). Aprēķinātas vienas Covid-19 vakcīnas kopējās izmaksas valstij katra pote maksā nedaudz virs 22 eiro. (The total costs of one COVID-19 vaccine have been calculated each vaccine costs the state a little over 22 euros). *TV3 Ziņas*. Retrieved from <a href="https://zinas.tv3.lv/latvija/sabiedriba/aprekinatas-vienas-covid-19-vakcinas-kopejas-izmaksas-valstij-katra-pote-maksa-nedaudz-virs-22-eiro/">https://zinas.tv3.lv/latvija/sabiedriba/aprekinatas-vienas-covid-19-vakcinas-kopejas-izmaksas-valstij-katra-pote-maksa-nedaudz-virs-22-eiro/</a>